CAPITAL UNIVERSITY OF SCIENCE AND TECHNOLOGY, ISLAMABAD



Analytical and DNN Based Prediction of DC Characteristics of Nanoscale FinFETs

by

Qamaruddin

A thesis submitted in partial fulfillment for the degree of Doctor of Philosophy

in the

Faculty of Engineering Department of Electrical Engineering

2022

Analytical and DNN Based Prediction of DC Characteristics of Nanoscale FinFETs

By Qamaruddin (DEE153003)

Dr. Antonio Jesus Garcia Loureiro, Professor University of Santiago de Compostela, Spain (Foreign Evaluator 1)

Dr. Luo Jun, Associate Professor IMCAS, Chaoyang District, Beijing, China (Foreign Evaluator 2)

> Dr. Muhammad Mansoor Ahmed (Thesis Supervisor)

Dr. Noor Muhammad Khan (Head, Department of Electrical Engineering)

> Dr. Imtiaz Ahmad Taj (Dean, Faculty of Engineering)

DEPARTMENT OF ELECTRICAL ENGINEERING CAPITAL UNIVERSITY OF SCIENCE AND TECHNOLOGY ISLAMABAD

 $\mathbf{2022}$

Copyright \bigodot 2022 by Qamaruddin

All rights reserved. No part of this thesis may be reproduced, distributed, or transmitted in any form or by any means, including photocopying, recording, or other electronic or mechanical methods, by any information storage and retrieval system without the prior written permission of the author. Dedicated to my family



CAPITAL UNIVERSITY OF SCIENCE & TECHNOLOGY ISLAMABAD

Expressway, Kahuta Road, Zone-V, Islamabad Phone:+92-51-111-555-666 Fax: +92-51-4486705 Email: info@cust.edu.pk Website: https://www.cust.edu.pk

CERTIFICATE OF APPROVAL

This is to certify that the research work presented in the thesis, entitled "Analytical and DNN Based Prediction of DC Characteristics of Nanoscale FinFETs" was conducted under the supervision of Dr. Muhammad Mansoor Ahmed. No part of this thesis has been submitted anywhere else for any other degree. This thesis is submitted to the Department of Electrical Engineering, Capital University of Science and Technology in partial fulfillment of the requirements for the degree of Doctor in Philosophy in the field of Electrical Engineering. The open defence of the thesis was conducted on December 09, 2022.

Student Name : Qamaruddin (DEE153003)

The Examining Committee unanimously agrees to award PhD degree in the mentioned field.

Examination Committee :

External Examiner 1: Dr. Shabbir Majeed Chaudhry (a) Professor UET, Taxila (b) External Examiner 2: Dr. Adnan Noor Assistant Professor GIK Institute, Topi, Swabi (c) Internal Examiner : Dr. Muhammad Ashraf Professor CUST, Islamabad Supervisor Name : Dr. Muhammad Mansoor Ahmed Professor CUST, Islamabad Name of HoD : Dr. Noor Muhammad Khan Professor CUST, Islamabad Name of Dean : Dr. Imtiaz Ahmed Taj Professor CUST, Islamabad

AUTHOR'S DECLARATION

I, Qamaruddin (Registration No. DEE153003), hereby state that my PhD thesis entitled, 'Analytical and DNN Based Prediction of DC Characteristics of Nanoscale FinFETs' is my own work and has not been submitted previously by me for taking any degree from Capital University of Science and Technology, Islamabad or anywhere else in the country/ world.

At any time, if my statement is found to be incorrect even after my graduation, the University has the right to withdraw my PhD Degree.

(**Qamaruddin**) Registration No : DEE153003

Dated: 01

December, 2022

PLAGIARISM UNDERTAKING

I solemnly declare that research work presented in the thesis titled "Analytical and DNN Based Prediction of DC Characteristics of Nanoscale FinFETs" is solely my research work with no significant contribution from any other person. Small contribution/ help wherever taken has been duly acknowledged and that complete thesis has been written by me.

I understand the zero-tolerance policy of the HEC and Capital University of Science and Technology towards plagiarism. Therefore, I as an author of the above titled thesis declare that no portion of my thesis has been plagiarized and any material used as reference is properly referred/ cited.

I undertake that if I am found guilty of any formal plagiarism in the above titled thesis even after award of PhD Degree, the University reserves the right to withdraw/ revoke my PhD degree and that HEC and the University have the right to publish my name on the HEC/ University Website on which names of students are placed who submitted plagiarized thesis.

(Qamaruddin) Registration No: DEE153003

Dated: Of December, 2022

List of Publications

It is certified that following publication(s) have been made out of the research work that has been carried out for this thesis:-

- Q. U. D. Memon, U. F. Ahmed and M. M. Ahmed. "A Unified Depletion/Inversion Model for Heterojunction Trigate FinFETs DC Characteristics," *IEEE Access* 9(2021), 89768-89777.
- Q. U. D. Memon, U. F. Ahmed and M. M. Ahmed. "A Schrödinger-Poisson Model for Output Characteristics of Trigate Ballistic Si Fin Field Effect Transistors (FinFETs)," *International Journal of Numerical Modelling: Electronic Networks, Devices and Fields* (2021), 35(1), e2927.
- Q. U. D. Memon, U. Munir U. F. Ahmed and M. M. Ahmed. "An Efficient Technique to Predict DC Characteristics of Nano FinFETs Using Deep Neural Network," *Journal of Computational Electronics* (2022), pp. 1-12.

(Qamaruddin)

Registration No: DEE153003

Acknowledgement

All praise to **Almighty Allah** who gave me the abilities to accomplish all the tasks and complete my PhD thesis.

I would like to express my heartiest gratitude to my supervisor **Prof. Dr. Muhammad Mansoor Ahmed**, Vice Chancellor, Capital University of Science and Technology (CUST), for his support, encouragement, and mentorship during my PhD. His kind support and tremendous help along with his insightful knowledge assisted me to overcome all the obstacles for achieving the desired goals during my research. It was a great honor for me to work with him and I am grateful for what I have learned from his gracious personality.

I am thankful to the administration of CUST for providing me with an excellent environment perfect for conducting my research. I would like to thank the Dean, Faculty of Engineering, **Prof. Dr. Imtiaz Ahmad Taj**, and Head of Electrical Engineering Department, **Prof. Dr. Noor M. Khan** for their continuous encouragement and support during the entire span of my studies at CUST.

Furthermore, I would also like to express my sincere gratitude to all the members of **Microelectronics and RF Engineering** research group, especially Mr. Umer Farooq Ahmed and Mr. Usama Munir for their support.

Finally, I would also like to express my feelings of gratitude to my family and friends for their exceptional love and encouragement throughout my life and eventually to achieve my goals that would not have been possible without their sincere co-operation and love.

(Qamaruddin)

Abstract

Si technology has played a vital role in today's semiconductor industry and a number of Si based devices have been invented, but none of them got more attention than field effect transistors (FETs). The scaling of FETs was successfully carried out upto 100 nm, but further downscaling was halted due to short channel effects and high leakage current. In order to overcome these problems, FinFETs were introduced which had a 3D channel that inherently provided a better gate on the channel characteristics. In this research, various modeling techniques for FinFETs DC characteristics have been developed to further its advancement for high-tech industry.

The first part of this research presents the development of a comprehensive model for trigate AlGaN/GaN FinFETs DC characteristics both for depletion and inversion mode of operations. For the depletion mode, 2DEG carrier concentration has been evaluated to assess drain current (I_{2D}) . Increase in gate potential after having fully un-depleted 2DEG, caused a corresponding increase in the drain current which could possibly be associated with the creation of additional channels in the device due to carriers inversion process. This component of the drain current (I_{inv}) has been modeled by solving a 2D Poisson equation. It has been shown that the total drain current $I_{ds} = I_{inv} + I_{2D}$. The validity of the developed technique has been demonstrated using experimental data which exhibited a good degree of accuracy.

Second part of the thesis focuses on a Schrödinger-Poisson based technique to predict DC characteristics of trigate Si FinFETs. The channel characteristics are modeled by a 3D Schrödinger wave equation considering quantum-mechanical ballistic transport. Using a non-equilibrium green function (NEGF) and a 3D Poisson equation, drain current associated with source-drain Fermi energy difference is evaluated. Contact resistance and its impact on the drain current is adjusted using a modeled parameter in Fermi-Dirac distribution function. To validate the proposed model, output and transfer characteristics of nanoscale FinFETs are compared with experimental data and a good degree of accuracy is demonstrated. It has been shown that the proposed model has the ability to predict FinFETs characteristics having $T_{fin} = 3 - 35$ nm.

Third part of thesis presents a deep neural network (DNN) model which has been designed to predict DC characteristics of FinFETs prior to their fabrication. To train the model, a sizable dataset has been generated using COMSOL software by varying device fabrication parameters. The model was then trained to a good degree of accuracy with selected dataset of FinFET characteristics, followed by an assessment of training accuracy using the control data. It has been demonstrated that the developed technique has the ability to predict DC characteristics of nanoscale FinFETs by using those variables which are handled by a design engineer. It is further shown that the technique is highly efficient when compared with standard simulation softwares hence, providing the industry a potential time and cost efficient solution.

Contents

Auth	's Declaration	\mathbf{v}
Plagia	ism Undertaking	vi
List o	Publications	vii
Ackn	vledgement	viii
Abstr	ct	ix
List o	Figures	xiv
List o	Tables	xvi
Abbr	viations	xvii
\mathbf{Symb}	ls	xix
 Int 1.1 1.2 1.3 1.4 1.5 	oductionWide Bandgap MaterialsFinFET-Fin Field Effect TransistorHeterostructure FinFETFinFET Layer Structure1.4.1 Substrate1.4.2 Oxide Layer1.4.3 Nucleation Layer1.4.4 Carrier Layer1.4.5 Barrier Layer1.4.6 Contact LayerAdvantages of FinFET	$ \begin{array}{c} 1 \\ 4 \\ 6 \\ 7 \\ 10 \\ 10 \\ 10 \\ 11 \\ 11 \\ 12 \\ 12 \\ 13 \\ 12 \\ 12 \end{array} $
1.6	 1.5.1 Reduction in Gate Induced Drain Leakage 1.5.2 Reduction in Drain Induced Barrier Lowering 1.5.3 Reduction in Gate Parasitic Capacitances 1.5.4 Improvement in Sub-threshold Characteristics FinFET Characteristics 	13 14 15 15 16

	1.7	DC Model				. 17
	1.8	Quantum Mechanical Effects				. 21
	1.9	Research Motivation				. 22
	1.10	Thesis Outline				. 24
2	Lite	erature Review				27
	2.1	FinFET Technology-An Overview			•	. 28
	2.2	Evolution in Heterostructure FinFETs			•	. 31
	2.3	Carrier Transport Mechanism in FinFETs			•	. 34
	2.4	Analytical Models			•	. 39
	2.5	Estimation of FinFETs Characteristics by AI			· •	. 43
	2.6	Thesis Motivation			· •	. 46
		2.6.1 Research Gaps				. 46
		2.6.2 Problem Statement & Methodology				. 48
	2.7	Summary				. 50
3	AU	Initial Depletion/Inversion Model for Heterojunctic)n '.	Fri	gat	e
	F'm	FETs DC Characteristics				52
	3.1		• •	• •	•	. 52
	3.2	AlGaN/GaN FinFET Model	• •	• •	•	. 54
		3.2.1 2DEG Current	• •		•	. 55
		3.2.1.1 Linear Region Current	• •		•	. 59
		3.2.1.2 Saturation Region Current	• •		•	. 60
		3.2.2 Inversion Current	• •		•	. 62
	3.3	Result and Discussion	• •	• •	•	. 67
	3.4	Summary	• •	• •	••	. 74
1		chrödinger-Poisson Model for Output Characterist	ice	of	Tr	i _
4	gate	P Ballistic Si FinFETs	ics	01	11.	76
	4 1	Introduction				76
	4.2	Model Development	•••	• •	•	. 10 79
	1.2	4.2.1 Schrödinger Equation	•••	• •	•	. 15 79
		4.2.2 Current Equation	• •	• •	•	. 15 82
		4.2.3 Poisson Equation	• •	• •	•	. 02 87
	13	Results and Discussion	• •	• •	•	. 01
	4.0	Summary	• •	• •	•	. 55
	4.4	Summary	• •	• •	•	. 102
5	Pre	dict DC Characteristics of Nano FinFETs Using De	ep 1	Ne	ura	al
	Net	work	•			103
	5.1	Introduction				. 103
	5.2	Modeling Methodology				. 107
		5.2.1 DNN Model				. 108
		5.2.2 DC Characteristics of FinFETs				. 110
		5.2.3 Features and Output Vector of the DNN Model				. 112
	5.3	Results and Discussion				. 114

	5.4 Summary	122
6	Conclusion and Future Work6.1Future Works	123 127
Bi	liography	130

List of Figures

1.1	Crossectional view of a trigate FinFET	3
1.2	Band diagram of narrow and wide bandgap materials used in elec- tronic industry.	5
1.3	(a) Trigate and (b) independent gate FinFETs.	8
1.4	Energy band diagram of n channel FinFETs	9
1.5	3D view of a heterostructure FinFET.	9
1.6	Typical (a) output characteristics (b) transfer characteristics (c) output conductance of a FinFET	18
1.7	Leakage current comparison for conventional FET and FinFET.	19
1.8	Comparison of (a) frequency (b) Transconductance-frequency prod- uct (TFP) (c) Gain transconductance frequency product (GTFP) with L_g of conventional FET (CFET) and FinFET devices [46]	20
3.1	3D view of an AlGaN/GaN FinFET channel	53
3.2	Energy band diagram of Heterostructure FinFETs.	55
3.3	x - z cross-sectional view of an AlGaN/GaN FinFET	56
3.4	x - y cross-sectional view of an AlGaN/GaN FinFET wherein the formation of the inversion layer is shown at the two side gates of the device	62
3.5	Potential profile at various points of an AlGaN/GaN FinFET (Device T_1).	69
3.6	Modeled and experimental drain current of AlGaN/GaN FinFETs of Table 3.1: (a) T_1 (b) T_2 (c) T_3 . Operation of the devices at high V_{gs} indicates the formation of inversion layer and its subsequent contribution in the drain current.	71
3.7	Modeled and experimental output conductance of AlGaN/GaN Fin- EETs described in Table 3.1: (a) T_{c} (b) T_{c} (c) T_{c}	72
3.8	Transconductance and drain current versus gate voltage of AlGaN/- GaN FinFETs: device (a) T_1 (b) T_2 (c) T_3 .	73
4.1	a) Channel of a trigate FinFET, b) $x - y$ crossection of a trigate FinFET.	78
4.2	Allowed energy levels in the conduction band of a FinFET	81
4.3	Inflow and outflow of charges in a FinFET channel	83

4.4	a) Variation in the channel potential predicted by the proposed model. b) Charge distribution inside the channel of a trigate Fin-	
	FET. The simulations are at $V_{ds} = 1$ V and $V_{gs} = 1.5$ V using the	<u> </u>
	physical dimensions of the device T_3	. 94
4.5	Fermi-Dirac distribution as a function of applied gate to source voltage, V_{gs} at different energy levels for device T_3 .	. 95
4.6	a) Channel transmission and b) difference in Fermi-Dirac distribu- tion energies at $V_{gs} = 2$ V. The dimensions of the simulated device	
	T_3 is provided in Table 4.1.	. 96
4.7	$I - V$ characteristics of trigate FinFETs for the devices of Table $(4.1; a)$ $T_{\rm c}$ and b) $T_{\rm c}$	98
18	L V characteristics of triggete FinFFTs: a) T and b) T	. 50
4.0	$I = V$ characteristics of trigate FinFETs. a) I_3 and b) I_4	. 99
4.9	Output conductance of trigate FINFLIS for the devices of Table 4.1: a) T_1 , b) T_2 , c) T_3 and d) T_4	. 100
4.10	Transconductance characteristics of trigate FinFETs for the devices of Table 4.1: a) T_1 b) T_2 c) T_3 and d) T_4 .	. 101
5.1	3D view of a tri-gate FinFET.	. 104
5.2	Processing flow to get output from a single neuron of a multilayer network system.	. 108
5.3	Schematics diagram of the proposed DNN model developed to pre-	110
	dict output characteristics of nano-FinFE1s.	. 113
5.4	DNN based training and prediction mechanism to compute DC characteristics of FinFETs.	. 115
5.5	I-V characteristics of nano-FinFETs predicted using a DNN model.	
	From (a) to (f) solid lines indicate data attained from COMSOL	
	simulation while dots represent the characteristics generated by the	
	DNN technique.	. 118
5.6	Transconductance and drain current variation of nano-FinFETs as	
	function of applied bias. (a) to (f) solid lines indicate COMSOL	
	data while dots represent DNN based predicted characteristics	. 119
5.7	Transconductance and drain current variation of nano-FinFETs as	
	function of applied bias. (a) to (f) solid lines indicate COMSOL	
	data while dots represent DNN based predicted characteristics	. 121

List of Tables

1.1	Fundamental features of various types of FETs	2
2.1	A brief history of FinFET technology.	30 22
2.2	A brief eventiew of AlGan/Gan based neterostructure FinFETs.	00 00
2.3 2.4	A brief overview of quantum transport mechanisms of FIFETS DC model expressions for EinFET $I = V$ characteristics	30 40
2.4 2.5	Parameters used in the development of DC analytical models by different researchers	40
2.6	A summary of AI based FETs techniques developed to predict the device characteristics.	45
0.1		c o
3.1	Physical parameters of different devices used in this study.	68 68
3.2	Physical constants used in this study.	68
4.1	Physical parameters of different devices used in this study	93
4.2	Physical constants used in this study.	93
4.3	Average root mean square error (RMSE) of the devices mentioned in Table 4.1	.01
5.1	MSE attained using various activation functions during DNN train-	
	ing process	.09
5.2	Effects of number of hidden layers and neuron on the proposed DNN technique to predict DC characteristics of FinFETs	.10
5.3	Various parameters used in DNN architecture developed to predict	
	DC characteristics of FinFETs	.11
5.4	FinFET parameters used in COMSOL software to generate dataset	
	by varying the given dimensions randomly between the two limits	
	given in the Table	.13
5.5	Typical time consumed by the COMSOL software to generate DC	
	characteristics for two extreme values of various physical variables	
	of FinFET's. A complete set of variables for these devices are given in T_{2} by f_{4}	00
E C	III TADIE 3.4	.20
0.0	dieted characteristics of six different FinFFT having physical di	
	mensions of that of Fig. 5.5	20
	$\mathbf{H}_{\mathbf{U}}$	-40

Abbreviations

2DEG	2 dimensional electron gas
AC	Alternating current
ADS	Advance software design
Al	Aluminum
AlGaAs	Aluminum gallium arsenide
AlGaN	Aluminum gallium nitride
AlN	Aluminum nitride
ANN	Artifical neural network
CAD	Computer aided design
CFET	Conventional field effect transistor
CV	Capacitance-voltage
DC	Direct current
DELTA	Depleted lean channel transistor
DG	Double gate
DIBL	Drian induced barrier lowering
DNN	Deep neural network
DRAM	Dynamic random access memory
FET	Field effect transistor
FinFET	Fin Field effect transistor
GAA	Gate all around
GAAFET	Gate all around field effect transistor
GaAs	Gallium arsenide
GaN	Gallium nitride

Ge	Germanium
GIDL	Gate induced drain leakage
GTFP	Gain transconductance frequency product
HEMT	High electron mobility transistor
IC	Integrated circuit
I - V	Current–voltage
JFET	Junction field effect transistor
MBE	Molecular beam epitaxy
MESFET	Metal semiconductor field effect transistor
MISO	Multiple input single output
ML	Machine Learning
MOSFET	Metal oxide field effect transistor
MSE	Mean square error
NEGF	Non-equilibrium green function
NW	Nanowire
QME	Quantum mechanical effects
RAM	Random acces memory
RF	Radio frequency
RMSE	Root mean square error
SCE	Short channel effect
Si	Silicon
SiC	Silicon carbide
SiO_2	Silicon dioxide
SNM	Static noise margin
SOI	Silicon on insulator
SRAM	Static random access memory
Т	Transistor cell
TCAD	Technology computer aided design
TFP	Transconductance frequency product
TSMC	Taiwan semiconductor manufacturing company

Symbols

2D, 3D	2 and 3 dimensional
a	Grid step
α	Fitting parameter
b	Bias value
β	Model parameter
C_{gd}	Gate to drain capacitor
C_{gs}	Gate to source capacitor
C_{ox}	Oxide capacitance
d	Distance between top gate and 2DEG
Δ	Laplace operator
ΔE_c	Bandgap discontinuity
ΔL_g	Channel length modulation
\mathbf{E}	Energy of wave function
E	Field inside the channel
ϵ_A	AlGaN permittivity
ϵ_G	GaN permittivity
ϵ_{eff}	Permittivity of 2DEG
ϵ_{si}	Silicon permittivity
ϵ_{ox}	Oxide permittivity
E_c	Lowest conduction band energy
E_F	Fermi energy level
E_g	$E_c - E_v$
E_{sat}	Saturation field

E_v	Valence band energy
E_y	y-directed electric field
f_{max}	Maximum frequeuncy
f_t	Unity gain frequeuncy
$\mathbf{f_{1,2}}$	Fermi-Dirac distribution
$\mathbf{G}^{\mathbf{n}}$	Electron Density Matrix
g_d	Output conductance
g_m	Transconductance
Н	Hamiltonian matrix
h	Planks constant
H_{fin}	Fin height
$I_{2\mathrm{D}}$	2DEG current
$I_{\rm 2D(l)}$	2DEG current at linear region
$I_{\rm 2D(s)}$	2DEG current at saturation region
I_{ds}	Drain to source current
I^A_{ds}	Actual drain to source current
I^P_{ds}	Experimental drain to source current
$I_{ds(sat)}$	Drain to source current at saturation region
I_{inv}	Inversion current
k_B	Boltzmann constant
k_i	Wave number
L_{fin}	Fin length
L_g	Gate length
L_{gd}	Gate to drain length
L_{gs}	Gate to source length
L_u	Underlap channel length
M	Number of $I - V$ curves
m	Mass of electron
m^*	Effective mass
N	Number of bias points
N_d	Bulk concentration

n_i	Intrinsic carrier concentration
n_s	2DEG carrier concentration
$P_{1,2}$	Chemical potential energy
Q	Channel charge
Q_i	Inversion charge
q	Electron charge
q_{id}	Electron charges at drain
q_{is}	Electron charges at source
Q_{inv}	Inversion charge
Q_{bulk}	Bulk charge
r	Structure parameter
R_{gd}	Gate to drain resistance
R_{gs}	Gate to source resistance
s	Inflow of electron
T	Temperature
t	Time
T_{fin}	Fin thickness
T_{ox}	Oxide thickness
t_{si}	Silicon thickness
U	Channel potential energy
U_f	Electrochemical Potential
V	Potential energy
V	Applied voltage
V_{ch}	Channel potential
V_d	Drain voltage
v_d	Drift velocity
V_{ds}	$V_d - V_s$
$V_{\rm d(sat)}$	Drain voltage at saturation point
V_{fb}	Flat band voltage
V_g	Gate voltage
V_{gs}	$V_g - V_s$

V_{GF}	Low drain voltage
V_L	Voltage across length
V_s	Source voltage
V_T	Thermal voltage
V_{th}	Threshold voltage
v_{sat}	Saturation velocity
W	Gate width
w	Weight value
W_{eff}	Effective fin width
W_{fin}	Width of the fin
x,y,z	Co-ordinate system
Σ	Rate of outflow of electron
σ	Activation function
\hbar	Reduced Plank's constant
λ_i	Eigenvalues
$\lambda_1, \lambda_2, \lambda_3, V_{g1}, V_{g2}$	Model constants
ζ_1,ζ_2,ζ_3	Temperature dependent parameter
η	Learning factor
μ	Mobility
μ_0	Low field mobility
ϕ_B	Built in potential
ϕ_b	Schottky barrier height, Metal work function
ψ,ψ_p	Surface potential
ψ_d	Surface potential at drain terminal
ψ_s	Surface potential at source terminal
χ	Electron affinity
ψ_0	Mid channel potential
Г	Hermitian matrix
ψ_{xyz}	xyz dependent wave function

Chapter 1

Introduction

The idea of field effect transistor (FET) was initially presented in the first quarter of 20th century however, the inventors were unable to build a working practical semiconducting device based on the proposed concepts [1]. By 1950, the boom in the electronic industry was primarily based on FETs as the large scale integration of these devices flourished the industry. Introduction of FETs in semiconductor industry had increased the number of transistors exponentially on a chip area enabling complex technologies and systems to be developed.

FETs based on their device structure are classified into: a) junction field effect transistors (JFETs); b) metal oxide semiconductor field effect transistors (MOS-FETs); c) metal semiconductor field effect transistors (MESFETs) and d) high electron mobility transistors (HEMTs) [2]. JFETs and MOSFETs were the first and second generation FET devices, respectively; and later on MESFETs and HEMTs were introduced; considered as the third generation devices. However, to cope with the ever-growing device scaling, Fin Field Effect Transistors (Fin-FETs) were first proposed as an alternative solution to JFET technology in 90s by Hisamoto *et al.* [3] and were designed to enhance further scalability of FET technology.

JFET controls the current flow from source-to-drain through a field caused by the gate voltage. Thus, gate terminal plays a crucial role in defining the response of

the device. On account of growing demands of space efficient integrated circuits (IC), the downscaling of JFET was limited due to its higher gate barrier [4] and high gate leakage caused by the narrow bandgap material used in its fabrication [5]. Furthermore, JFET has a high input and low output impedances and the performance of JFETs deteriorates at higher operating frequencies associated with internal Miller capacitors of the device. Improvements in JFET are achieved by growing a layer of Silicon dioxide (SiO₂) on top of Si and by depositing a layer of metal on the very top, and the device thus formed, is called MOSFET [6]. MOSFET is considered a better substitute in semiconductor industry, as it has relatively higher input impedance and a negligible leakage current [5]. MOSFET usage is restricted because of the breakdown of oxide, as the isolation between the gate and channel can be damaged easily by the accumulation of static charges [7]. In addition, both JFET and MOSFET performance deteriorate at higher temperature and this becomes more serious at large scale integration.

MESFET is a field effect transistor similar to JFET with a Schottky junction instead of a *pn*-junction for the gate. In MESFET, compound semiconductor such as, GaAs is used which offers better mobility compared to Si, and thus better high

Parameters	MOSFET	MESFET	HEMT
Material	Elemental	Compound	Heterojunction
Input Impedance $[8]$	Very High ($\sim 10^6$)	High	High
Off Current $[5, 9]$	$\sim 100 \text{ nA}$	${\sim}10~{\rm nA}$	${\sim}10~{\rm nA}$
Substrate	Semiconductor	Semi insulator	Semi insulator
Types	E & D	D	D
Application [10]	Digital and memory	RF and power	Low Noise and
	devices		mm-wave
Usual gate length [7]	$\sim 100 \text{ nm}$	${\sim}60~\mathrm{nm}$	${\sim}30~\mathrm{nm}$

TABLE 1.1: Fundamental features of various types of FETs

E & D:Enhancement and Depletion

frequency characteristics [11]. To further improve the performance, heterojunction materials such as AlGaN/GaN are used and with the introduction of these materials, focus is shifted towards HEMT technology [12]. HEMT offers higher mobility due to the formation of 2-dimensional electron gas (2DEG) [11], and can operate even below milli kelvin.

However, in all above mentioned devices, scalability remained a major concern. Reduction in the device size generated other associated issues commonly known as short channel effects (SCE) which limited further reduction in the device size and hence, number of components per unit area of the chip. However, by using different compound materials and technology modifications, at the most, 32 nm gate length (L_g) devices are reported [13, 14]. Table 1.1 summarizes the properties and scaling limits of conventional FETs.

Beyond 30 nm, SCEs such as drain induced barrier lowering (DIBL), threshold voltage roll off, compression in transconductance, parasitic capacitance and gate leakage through hot electron tunneling, etc; are observed [15]. To cater these problems while also continuing further device scaling, FinFETs were introduced. FinFET has a 3D channel referred to as fin of the device, which is enveloped by the gate metal from three sides [16]. Due to the gate which is wrapped around the channel from three sides, these devices are also called trigate devices as illustrated



FIGURE 1.1: Crossectional view of a trigate FinFET.

in Fig. 1.1. The 3D gate structure allows better gate control upon the channel current and hence SCEs. The introduction of FinFETs made revolutionary improvements in semiconductor technology and opened a new era of IC scalability [17]. First generation FinFETs were based on double-gate (DG) MOSFET technology with reported L_g down to 14 nm [18]. Currently, trigate FinFET is considered as device of the day having good popularity in semiconductor industry. Now 14 nm / 16 nm trigate FinFET technology has become a mature product whereas researchers are working on nanowire FinFETs with sub 10 nm L_g [19].

1.1 Wide Bandgap Materials

A transistor is made up of a semiconductor material, so the material of the device has a major role in determining its performance. Since 1950, Si has been the most often utilized device material and by 1980, its technology attained highest maturity with score of applications in digital industries. Si based technology had reached to its theoretical limits where further optimization in terms of performance and device miniaturization was hard to achieve. The fundamental constraint on further growth of Si-based technology was its slow switching speed, device performance deterioration at higher temperature and substantial power losses. To improve the devices further, both from device size as well as the performance point of view, scientists and technologists moved towards compound semiconductors such as GaAs, SiC, AlGaAs etc.

GaAs based devices, which are based on a compound semiconductor, gained popularity since these devices have better electron mobility and higher saturation velocity (v_{sat}) when compared to Si-based devices. Although GaAs based devices perform relatively better at higher temperature, yet these suffered with some limitations because, the material itself has been classified as narrow bandgap material, causing constraints on its applications in the industry. Nevertheless, GaAs devices offered greater switching frequencies, allowing them, predominantly, to be employed in low-power microwave industry and satellite communication. Si-based



FIGURE 1.2: Band diagram of narrow and wide bandgap materials used in electronic industry.

bipolar devices, on the other hand, have greater power handling capabilities, but low operating frequency ranges. Furthermore, it was also realized that many important applications such as: military; aerospace; automotive; and motor drives required special devices capable to retain their characteristics in harsh environment. To address these constraints, a new class of material termed as wide bandgap material, was adopted to meet the advanced technological requirements. When compared to its counterparts, wide bandgap materials such as SiC, GaN, AIN, AlGaN etc offer relatively better characteristics, and thus emerged as potential candidates for future generation of electronic devices. Among these wide bandgap materials, GaN and SiC provide high mobility along with high temperature stability; allowing them to deliver higher currents with improved switching frequencies. Such materials can also sustain at high electric field even at higher temperature compared to narrow bandgap materials, and are therefore, promising contender for future generation of devices.

A pictorial representation of narrow and wide bandgap materials is shown in Fig. 1.2. Ge has a indirect bandgap of 0.67 eV, while Si exhibits an indirect bandgap of 1.1 eV. Devices made by these materials fail at higher temperature due to breakdown phenomenon, and hence Si and Ge are not promising candidates for high temperature applications. On the other hand, GaAs has a bandgap of 1.43 eV, making it a relatively better option for semiconductor industry provided other parameters such as mobility and carrier velocity are also supportive. Lastly, as depicted in Fig. 1.2, GaN has a 3.4 eV energy gap and could be called as a wide bandgap material. Additionally it has a bandgap that is nearly triple to Silicon's, owing to that it can handle higher voltages, power, and temperature. Since GaN offers higher carrier velocity, therefore, GaN based devices allow smaller, faster, robust and power efficient devices compared to its narrow bandgap counterparts.

1.2 FinFET-Fin Field Effect Transistor

FinFET is an alternative of planar MOSFET structure. Based on the gate configurations, FinFETs are classified into two categories: a) trigate FinFETs and b) independent gate FinFETs. Fig 1.3 (a) & (b) show a crossectional view of a trigate and independent gate FinFETs, respectively. Independent gate device is a four terminal device, while trigate FinFET is a three terminal device. The gate of a trigate FinFET maintain a constant potential, called as equi-potential acting on three sides of the channel, while independent gate FinFET may have the same or independent potentials applied on gates. The basic physical dimensions of a FinFET are: gate length L_g ; fin thickness T_{fin} ; fin length L_{fin} ; and height of the fin, H_{fin} . The effective width (W_{eff}) of a FinFET plays a major role in current and is determined by $W_{eff} = 2H_{fin} + T_{fin}$. To increase the drive current of a FinFET, parallel fins may be added to increase W_{eff} . On the other hand, H_{fin} to T_{fin} ratio should be kept less than 3, to mitigate the SCEs [20]. The energy band diagram of an *n*-channel FinFETs is shown in Fig 1.4. This figure shows a biased system having an oxide layer sandwich with *n*-type semiconductor and the gate metal. In Fig. 1.4, apart from other usual variables ϕ_b represents work-function of the gate metal, E_F is the Fermi energy, E_c energy of the conduction and E_v represents the valance band energy.

A further down scaling of FinFETs is considered as a challenging task both for the researchers and the industry because, as the channel of a FinFET becomes smaller, a variety of effects such as hot carrier effects, velocity overshoot, shift in threshold voltages and degradation of mobility due to surface scattering are observed [21]. To minimize these effects, heterostructure FinFETs are proposed [22] which also provide solution for SCEs especially those related to mobility degradation and thus, eases further scalability in FET devices.

1.3 Heterostructure FinFET

Over the many years, a lot of improvements have been suggested involving structure optimization of FinFET. As the technology advanced and inventions of new gadgets and other electronic devices flooded the market, one of the key demands pertaining to device operation was its ability to perform in an unfriendly environment, where most of the other devices either fail or exhibit an uncertain behavior. Researchers have been able to fabricate heterojunction FinFETs by combining two mis-matched semiconductor layers such as, AlGaN/GaN. A combination of such



FIGURE 1.3: (a) Trigate and (b) independent gate FinFETs.

two layers, because of mis-matched bandgap, allows the free electrons to trap at the interface of the two materials, defining a charge layer referred to as 2DEG also shown in Fig. 1.5 [23, 24]. Electrons once trapped in 2DEG are available for conduction at below liquid helium temperature. The devices made by such a material, offer stable characteristics for a wide range of temperatures, starting from milli kelvin to a reasonably high temperature observed in automobile and other engine environments. Additionally, charges trapped in the form of 2DEG are not associated with lattice atoms, rather they are trapped in a quantum well, therefore, they offer superior mobility and high v_{sat} when compared with bulk materials.

Wide bandgap heterostructure offers an additional benefit, because of piezoelectric

effect [25] observed in such materials. This effect allows a natural generation of carriers and their trapping in the quantum well without involving the spacer layer to reduce the ionized ions scattering. Thus, the devices made from such a heterojunction, offer better mobility along with the ease of fabrication, and owing to that, heterostructure FinFETs are preferred in the industry [26].



FIGURE 1.4: Energy band diagram of n channel FinFETs.



FIGURE 1.5: 3D view of a heterostructure FinFET.

1.4 FinFET Layer Structure

Based on the composition of device material, FinFETs are classified into homo and heterostructure FinFETs. In a heterostructure FinFET, fin of the device comprises of two materials having different bandgaps, such as AlGaAs/GaAs or AlGaN/GaN etc as explained in the preceding sections. On the other hand, homostructure devices are based on elemental or compound semiconductor materials, such as Si, GaAs or GaN etc. Based on the type of the device, FinFETs can be realized by employing various layers as explained below.

1.4.1 Substrate

Substrate is an important part of a FinFET often referred to as the base of device, upon which the entire structure is built on, as shown in Fig. 1.1. Selection of substrate depends upon the device layer structure. For the devices in which fin is defined by the Si material, substrates of two types are normally used i.e., Si and SOI. Amongst these, Si is the most commonly used material which offer acceptable thermal conductivity [27], while SOI substrate incorporates an insulating layer between the main substrate and the channel to reduce the effect of carrier trapping but on the expense of fabrication complexity [28].

1.4.2 Oxide Layer

In FinFET fabrication, oxide layer is employed for various purposes, such as, gate oxide as shown in Fig. 1.3 (a) & (b), capping oxide as illustrated in Fig. 1.1. Changes in gate oxide thickness (T_{ox}) may effect the threshold voltage (V_{th}) and DC performance of the device. In addition, reducing the thickness of oxide may increase the gate leakage current, whereas increasing it beyond a certain limit, reduces gate control upon the channel [29]. Capping oxide is normally employed to maintain device performance under varying ambient conditions. The thickness of capping oxide layers depends upon the drain, source and gate metal geometry [30]. The prime purpose of this layer is that, these terminals along with the exposed semiconductor material should be covered appropriately to make them immune so that they should be less sensitive to varying ambient conditions [31].

1.4.3 Nucleation Layer

Nucleation layer is grown on the substrate by using an epitaxial growth technique such as molecular beam epitaxy (MBE) [21]. The purpose of this layer is to reduce the stress and lattice mismatch between substrate epitaxial layers which will be grown subsequently [21]. Nucleation layers are usually defined by the super lattices and they are normally of the same material as that of channel. Any lattice defects which are there in the substrate are blocked by these epitaxially grown layers [31].

1.4.4 Carrier Layer

In heterostructure devices, channel essentially comprises of two layers commonly known as carrier and barrier layer. The interface of these two layers accommodates free carriers which define 2DEG. The combination of these two layers makes the device heterostructure, because of the bandgap mismatch between these layers [21]. The carriers for 2DEG are supplied by a relatively wide bandgap layer and such a layer can be either intrinsic or extrinsic in nature depending upon the chosen material. If the chosen layer has natural piezoelectric effect as observed in AlGaN; in such a case, carriers can be generated under the influence of piezoelectric effect. The generated carriers then naturally move towards lower bandgap layer to define 2DEG [32]. The benefit of existence of piezoelectric effect in a relatively wide bandgap layer allows the formation of 2DEG without the insertion of carrier layer (doped layer) and hence, enhances the mobility of 2DEG electrons in the absence of ionized ion scattering [33]. In general, materials of both of these layers are chosen to facilitate the formation of 2DEG at the interface, by keeping the bandgap of carrier layer relatively lower than that of barrier layer [34].

1.4.5 Barrier Layer

Barrier layer is formed on top of the carrier layer for the formation of 2DEG as shown in Fig. 1.5. This figure shows the minimum requirements to form a heterostructure FinFET. In order to improve the device performance further, sometimes the barrier layer is also covered with another layer of a compatible material, to minimize oxide formation, specially if the barrier layer contains highly reactive elements such as Al. Special attention is required to design the thickness of barrier layer to avoid parasitic FET formation. Devices which are principally meant to give 2DEG dependent characteristics, require a fully depleted barrier layer at gate-to-source voltages, $V_{gs} = 0$ V. If there is a finite un-depleted thickness of the barrier layer available at $V_{gs} = 0$ V, the device characteristics will be governed by the two layers; a) barrier layer and b) 2DEG layer, which is an undesirable feature in heterostructure FinFETs [21]. Hence, it is a fundamental requirement that the thickness of the barrier layer should be such that at $V_{gs} = 0$ V, it is fully depleted. As a result, the drain-source conduction, under the applied drain bias, will be governed by the 2DEG characteristics of the device [35].

1.4.6 Contact Layer

There are three types of contacts in a FinFET. Two of them are ohmic known as drain and source contacts, and the third one is the gate contact known as Schottky contact, which could be either with or without gate oxide [21]. If the gate contact is without gate oxide, then the device operation would be fundamentally described by the MESFET operation and if the gate metal is placed on top of an oxide layer then the device operation is identical to that of MOSFET. Contact layers are normally grown on top of barrier layer, in order to reduce the contact resistance of the device. Generally, contact resistance is defined as ohmic resistance caused by the ohmic metal plus the resistance offered by the bulk region of the device between gate-drain/gate-source. Presence of heavily doped contact layer on top of barrier layer, but underneath the ohmic metal, will minimize the ohmic contact resistance and hence, the maximum portion of the applied potential shall appear under the gate region, which controls device characteristics. Therefore, the resistance offered by gate-drain/gate-source region is referred to as parasitic resistances and the presence of the contact layer will minimize these [21, 32].

1.5 Advantages of FinFET

The invention of FinFETs has caused a tremendous impact on semiconductor industry and is still evolving with new research. Despite of the manufacturing complexities, the FinFET has made its way into the industry and prospered with promising superiority over its counterparts. The FinFET proved its suitability for IC fabrication due to high scalability and provided a new pathway for Moore's law beyond 20 nm as they have much better performance with reduced power consumption compared to planar transistors. A 16 nm / 14 nm FinFET device can potentially offer a 40 – 50 % increased performance compared to a 28 nm planar device [31]. Furthermore, fins are vertical in nature, allowing better gate control which leads to high fabrication density. Additionally, due to their unique structure, FinFETs have low leakage current, can operate at lower voltage and offer a higher battery life for mobile systems [31, 36]. Following paragraphs discuss some of the major factors which are offered by FinFET technology and therefore preferred by the industry.

1.5.1 Reduction in Gate Induced Drain Leakage

It is an established fact that, the FinFET is a three terminal device and in such a device, the highest potential is observed between the gate and drain terminals. The carriers in this region become hot and have sufficient energy to climb over the barrier, referred to as thermionic emission [37]. These hot carriers are prime contributors to a current called as gate induced drain leakage (GIDL) current. The presence of significant GIDL causes a rise in the temperature of chip and hence
it is a constraining factor which limits the number of devices per unit area of the chip. Researchers have been trying in the past to reduce GIDL in order to improve battery lifetime and also to increase device density on a chip. A crucial factor to reduce GIDL is, to modify the depletion underneath the gate, especially towards gate-drain region. Owing to this fact, earlier in MESFET design, the planar device geometry was modified into recessed gate FET where the gate is placed below the drain-source plane, which modified the edges of depletion and hence, it improved GIDL. A similar philosophy has been applied in FinFET, where the gate-drain depletion has been modified by having a trigate geometry. This improved the performance of the device by reducing undesirable flow of current caused by the gate-drain potential and hence a reduction in GIDL is observed. The presence of oxide layer, its type and thickness will also play a role in determining the magnitude of GIDL [38]. Since GIDL depends upon the depletion geometry in the gate-drain region of the device which is dependent upon T_{fin} and H_{fin} , hence their optimized values will also play a role in reducing GIDL [37].

1.5.2 Reduction in Drain Induced Barrier Lowering

Drain induced barrier lowering (DIBL) is a negative feature observed in Schottky based FETs and it is more prominent in short channel devices. Since, the depletion under a gate is not uniformly distributed and its thickness is more towards the drain side of the device, increased number of charges in the depletion towards the drain side, cause image charges in the gate metal resulting into a reduction in the gate potential, which eventually reduces the thickness of depletion towards the drain side of the device referred to as DIBL [39]. As the carriers are hot in this region, and presence of DIBL offers a relatively lower barrier which results into an increased flow of carriers from the channel to the gate metal thus, causing an undesired flow of current. Normally, such behavior is dominating in those FETs where gate is placed directly upon the semiconductor without intervening oxide layer. Since, the shape of depletion is modified because of the trigate nature of the FinFET, so this geometry also mitigates DIBL and hence, improves the device performance [39].

1.5.3 Reduction in Gate Parasitic Capacitances

Parasitic capacitances referred to as Miller capacitances are unavoidable and they exist between drain-gate and source-gate regions of the device and determine its high frequency characteristics. The parasitic capacitances depend upon the device dimensions, specially L_g [40]. While the L_g decreases in a conventional FET, its Miller capacitors also decrease, but at the same time it reduces its transconductance (g_m) , which results into reduction in its RF performance [41]. Hence, it is always a desirable feature that, the device used in the system should have Miller capacitors as low as possible. Mathematically, Miller capacitors and unity gain frequency (f_t) of operation are related as

$$f_t = \frac{g_m}{2\pi (C_{gs} + C_{gd})}$$
(1.1)

where, C_{gs} is gate-to-source and C_{gd} is gate-to-drain Miller capacitors. Nanoscale FinFETs meant for millimeter waves applications, because of their specific geometry, offers exceptionally low value of C_{gs} and C_{gd} along with relatively high value of g_m because of 3D operation of the gate. Thus, FinFETs are highly suitable for high frequency applications. The value of capacitances can be tailored according to the need of an application by optimizing the geometry of fin i.e., T_{fin} , H_{fin} , L_g , and L_{fin} . In short, the negative effects of parasitic capacitance can be overcommed in a more effective manner in FinFET applications which makes them more suitable for high frequency.

1.5.4 Improvement in Sub-threshold Characteristics

Threshold voltage (V_{th}) is the minimum gate-to-source voltage that is required to make a conducting path between the source and drain, if it is not already there.

On the other hand, for depletion type FinFET, where the channel is physically available, V_{th} is a voltage which will deplete the channel completely and ideally at this applied voltage there should be zero drain current flowing between drain and source. The magnitude of V_{th} plays an important role in determining the sensitivity of the device and it will also determine the battery life of a system in which FinFET is used as an active device [42]. In an ideal device, $V_{gs} = V_{th}$ should define $I_{ds} = 0$, but practically speaking, at such condition in conventional FET, I_{ds} has got a non-zero value. To improve this, FinFET technology was introduced, wherein the channel is controlled from 3 different directions, bringing the sub-threshold characteristics closer to their ideal values. This technology also improved the off-state conduction of the device, hence allowing a better device thermal management which eventually led to very large scale device integration [43, 44].

1.6 FinFET Characteristics

FinFETs operate like conventional MOSFET devices i.e., they can have both depletion and enhancement mode of operations. Furthermore, depending upon the dominant carriers, they are classified into either p-channel or n-channel devices. Typical DC characteristics of a FinFET are shown in Fig. 1.6 (a), wherein it can be observed that by increasing V_{ds} , I_{ds} also increases in a linear fashion for small values of V_{ds} and then it saturates to a constant value known as $I_{ds(sat)}$. On the other hand, voltages applied on the gate terminal i.e., V_{gs} , controls the amount of current flowing through the channel by changing the depletion height. Moreover, transfer characteristics $g_m(V_{gs})$ also play an important role in determining the quality of the device as shown in Fig. 1.6 (b). A high value of g_m will indicate a good gate response and such devices will offer a good gain to an incoming signal. It can be seen from Fig. 1.6 (b) that, g_m is also dependent upon V_{ds} and usually the g_m of a device is reported after the onset of current saturation i.e., $V_{ds} \geq V_{ds(sat)}$. The device output conductance, g_d is also used to determine the quality of a finished FinFET. Fig. 1.6 (c) shows a typical g_d response of a FinFET which is

dependent both on V_{gs} and V_{ds} . The data of plot indicate that, the device offer high g_d at relatively lower applied bias and it tends to saturate where current of the device exhibits saturation, and hence it offers lowest value of g_d . A contrast of leakage current for conventional FET and FinFET is given in Fig. 1.7. As depicted, both planar and FinFET offer the same leakage current at higher V_{gs} value, but as V_{gs} decreases, FinFET exhibit considerably lower gate leakages. This phenomenon could be associated with the trigate structure of the device, where the channel is controlled from multiple sides by the gate potential applied on it [45]. Capacitors, C_{gs} and C_{gd} play an important role in device AC performance. Due to the trigate geometry, both C_{gs} and C_{gd} of FinFETs are relatively lower compared to conventional FETs thus, they offer high operating frequency as evident by Eq. 1.1. This is also shown in Fig. 1.8 (a), wherein the maximum achievable frequency, f_{max} as a function of L_g for both conventional FETs and FinFETs are shown. By examining the figure, it can be seen that as L_g reduces, f_{max} increases. However, FinFETs offer higher f_{max} across all L_g values compared to conventional FET.

Other AC performance parameters are transconductance-frequency product (TFP) and gain transconductance frequency product (GTFP). TFP represents the trade off between power and bandwidth in high speed device design, while GTFP gives overall performance criterion of a new device. These parameters provide an excellent estimation of a device capability with respect to AC performance. By examining the TFP and GTFP for conventional FETs and FinFETs plotted in Figs. 1.8 (b) & (c), respectively, one can observe that for the given range of L_g , FinFETs performance is better compared to conventional FETs. Hence, making it a superior device in the FETs family.

1.7 DC Model

In conventional FETs, the potential distribution inside the channel can be estimated by solving a 2D Poisson equation given as [47]



FIGURE 1.6: Typical (a) output characteristics (b) transfer characteristics (c) output conductance of a FinFET.



FIGURE 1.7: Leakage current comparison for conventional FET and FinFET.

$$\frac{\partial^2 \psi(x,y)}{\partial x^2} = \frac{qn_i}{\epsilon_{si}} \exp\left(\frac{\psi(x,y) + \phi_B - V_{ch}(y)}{V_T}\right) + \frac{qN_d}{\epsilon_{si}}.$$
 (1.2)

The parameters used in Eq. 1.2 have usual meaning as defined in the list of variables. The same approach can be applied to evaluate field distribution inside a FinFET however, to accommodate 3D geometry, a 3D Poisson equation will be employed:

$$\frac{\partial^2 \psi(x, y, z)}{\partial x^2} + \frac{\partial^2 \psi(x, y, z)}{\partial y^2} + \frac{\partial^2 \psi(x, y, z)}{\partial z^2} = \frac{qn_i}{\epsilon_{si}} \exp\left(\frac{\psi(x, y, z) + \phi_B - V_{ch}(y)}{V_T}\right) + \frac{qN_d}{\epsilon_{si}}.$$
 (1.3)

By solving Eq. 1.3 under appropriate boundary conditions, charges available in the channel (Q_{inv}) of the device will be calculated. By involving the mobility, μ , I_{ds} can then be evaluated by using expression of the form:



FIGURE 1.8: Comparison of (a) frequency (b) Transconductance-frequency product (TFP) (c) Gain transconductance frequency product (GTFP) with L_g of conventional FET (CFET) and FinFET devices [46].

$$I_{ds} = \mu (2H_{fin} + T_{fin})Q_{inv}\frac{dV(y)}{dy}$$

$$\tag{1.4}$$

1.8 Quantum Mechanical Effects

FinFETs have the ability to control channel current while minimizing SCEs by using multi dimensional 3D gates. This 3D gate geometry provides efficient electrostatic control over the channel by increasing the effective channel width and paves the way for further transistor size reduction. However, upon reducing the L_{fin} to a few nanometers, the T_{ox} also has to be decreased [48]. The decrease in dielectric thickness with the L_{fin} , increases the impact of tunneling significantly. This phenomenon occurs when carriers tunnel through the device's gate dielectric, which ideally should act as a perfect barrier. Although the electron energy is less than the gate dielectric barrier energy, but still, there is a finite probability of carriers transmission, referred to as quantum mechanical effects. Furthermore, the concern becomes more critical when the penetrating particles start behaving like a wave. So, the presence of quantum mechanical effects in nano-FinFETs become unavoidable to comprehend electron transport and to optimize the device size.

A conventional device structure requires classical equations to evaluate electron distribution inside the channel, whereas, in nano-structures electrons behave as a wave classified by a quantum phenomenon, thus, the current caused by such quantum particles shall be governed by a quantum transport model. To move carriers from source to drain and to account for quantum effects, under special circumstances, a hybrid technique involving classical and quantum effects may be employed simultaneously [49]. Quantum mechanical transport models can be obtained by solving the Schrödinger wave equation in three dimensions to establish the contribution of quantum particles responsible to determine the characteristics of the device. On the other hand, in hybrid technique, both Poisson and Schrödinger equations can be engaged simultaneously to evaluate the device characteristics. The main purpose of engaging 3D Poisson equation would be to determine the charge distribution in the channel. With this approach, nano-scale FinFET DC characteristics can be modeled accurately compared to a conventional approach wherein, the Poisson equation coupled with a mobility model is engaged, to determine the device characteristics [21].

1.9 Research Motivation

The era in which we are living can truly be called as advanced era, where our lives are totally dependent upon technology and mostly on smart devices, such as cellphones, laptops and other similar gadgets. These technological advancements have made our lives a lot easier while there is always a desire of reducing the physical size of mobile devices without compromising on their performance. Technological advancements have caused further reduction in the size of semiconductor device which is a basic building block of a smart device. A phenomenal growth in semiconductor industry was observed even after five decades from the development of first FET. As the time passed, and the research advanced in various technological fields, the demand of these devices increased sharply as predicted by Moore's law [15]. With the advent of advance processing techniques even with the reduction of their physical size, the performance of these devices increased exponentially. This made the appliances and gadgets used in our daily life smaller, easier and more feature oriented.

Research has also been carried out to add new materials for the fabrication of FETs to make appliances better with increased fabrication density. In an effort to further enhance the performance of FET based systems where the downscaling was halted at 100 nm due to increased SCEs, multi gate FinFETs are introduced. Multi-gate technology is a relatively newer one but highly promising because it offers improved device performance at nano-scale level. Thus, researchers are concentrating to develop nano-scale FinFETs both on Si and compound semiconductor such as GaAs, AlGaN/GaN etc, to cater industrial needs.

Recently, AlGaN/GaN FinFETs are developed which operate both in depletion as well as in inversion mode of operation. For such 3D structure of FinFETs, there is a need to develop analytical models which can predict the characteristics of FinFETs for their entire range of operations i.e., depletion as well as enhancement mode. The previously reported models, which will be discussed in the next chapter, are meant only for FinFET depletion mode of operation and hence they are silent and unable to predict the device performance in enhancement mode simultaneously. Hence, any such effort, wherein, the FinFET characteristics would be predicted mathematically in a comprehensive manner incorporating both depletion and enhancement mode, shall be an extension in FinFET device understanding and to the best of our knowledge, such a study is not reported in the literature.

FinFETs having gate length below 10 nm exhibit quantum mechanical effects and their characteristics cannot be determined using conventional FET modeling approaches. For such devices, it is imperative to incorporate quantum effects wherein the particles are described by a wave function and the energy of such wave function is described by a quantum expression usually achieved by solving Schrödinger wave equation. Coupling the physical dimensions of the device with quantum mechanical effects is a challenging task and this research work is aimed to develop such a technique which will predict the device response and hence, offer a better device understanding at nano-scale level. Such a model can be employed as a tool in device design and simulation softwares to predict the device response even before its fabrication.

Artificial intelligence (AI) is considered an effective tool and is being employed widely in various fields to have a better and efficient technological solutions. Since, nano-scale FinFET involves a high-tech semiconductor industry, to overcome the product developmental cost, it would be wise to have an AI based technique capable to predict the device characteristics to a good degree of accuracy by involving device physical parameters, which are normally engaged by a design engineer to optimize the device performance. To develop such a system, a dataset involving device physical parameters and corresponding characteristics shall be a core requirement. An exhaustive search for such a database was carried out and it was noted that there is no such data available or reported in the literature. It is proposed that to develop an AI based system by involving an established tool known as COMSOL Multiphysics, a data bank of FinFET characteristics shall be developed. Based on the developed data bank, a deep neural network (DNN) should be proposed to predict the device characteristics involving the device physical dimensions. The proposed technique could be compatible to the relevant industry to predict nano-scale FinFET characteristics prior to their fabrication and hence its engagement could potentially reduce the product cost.

1.10 Thesis Outline

Chapter 1 presented fundamental concepts pertaining to FinFETs fabrication, characterization and their technological importance. It has been discussed that the FinFET would be the device of future and it will play an important role in miniaturized systems specially in mobile technology. It is discussed that nano-scale FinFETs are preferred because they have the ability to mitigate SCEs usually observed in FET devices when the channel length is reduced below 100 nm. It is discussed that, a FinFET performs relatively better than its competitors because of its 3D geometry. The dependence of device characteristics on its various layers is also discussed in this chapter. Different materials, and their associated characteristics, which could be involved in FinFET fabrication are also discussed. The chapter also presents a brief related to FinFET structure when fabricated using 2DEG materials. It has also been described that an accurate device DC model could only be realized if quantum mechanical effects presence in such devices are taken into considerations.

Chapter 2 focuses on previously reported studies related to FinFETs and its technological parameters such as device layer structure, device design and fabrication. A detailed overview of conventional modeling techniques, quantum transport models and applicability of such techniques to a 3D homostructure is presented.

Moreover, quantum transport and involvement of AI to predict device characteristics have also been discussed in this chapter. A comparative analysis of various models is presented and based on the reported data research gaps for future study are developed.

Chapter 3 presents a comprehensive I - V model for the evaluation of above threshold DC characteristics of trigate heterostructure FinFETs, for both depletion and inversion mode of operations. According to our conservative estimate, no such model is reported in the literature combining both depletion and inversion mode of operations simultaneously. Current associated with 2DEG referred to as depletion current has been evaluated by solving a 2D Poisson equation with appropriate boundary conditions. On the other hand, the inversion current is assessed by involving a 3D Poisson equation and the developed I - V model incorporates both the solutions to define a comprehensive expression for the generation of DC characteristics of FinFETs as a function of applied bias. It has been shown that the developed model offers a reasonable accuracy when the modelled characteristics are compared with experimental data.

Chapter 4 presents a Schrödinger-Poisson based technique to predict the behavior of trigate FinFETs. The Schrödinger equation is solved for 3D device structure by incorporating quantum-mechanical ballistic transport. A 3D Poisson equation and non-equilibrium green function are used to evaluate charge distribution which is eventually coupled with Schrödinger equation to assess drain current of FinFETs. The contact resistance and its impact on drain current is also assessed using Fermi-Dirac distribution function. It is demonstrated that the drain current, as a function of drain bias, saturates because of the finite supply of carriers from the source electrode. The model is calibrated using TCAD and its accuracy is established using experimental data.

Chapter 5 presents a technique which predicts DC characteristics of FinFETs prior to their fabrication. The developed technique uses a deep learning AI approach, wherein for the training of the system, a dataset of DC characteristics

is initially developed using COMSOL Multiphysics software, which generates device characteristics by involving its physical parameters including gate length, fin length, fin height, fin thickness and doping concentration. The trained AI network has the ability to generate FinFET characteristics of any dimension prior to its fabrication. It is demonstrated using the control data, that the proposed deep learning approach has the ability to predict device response in seconds as opposed to hours required by standard simulators. Thus, the developed approach could be a beneficial tool for industry to make industrial process efficient both from time as well as cost point of views.

Chapter 6 discusses the outcomes achieved from this research and proposes possible extension which can be carried out as a future work.

Chapter 2

Literature Review

Since the advent of semiconductor devices, there has been always a struggle to improve the device performance, and to do so, a number of device architectural advancements have been proposed such as JFET, MOSFET, MESFET etc. The scalability of these FETs proved to be a cumbersome task for design engineers and as the L_g shrinks beyond 50 nm, it begins to lose its control upon the channel and undesired issues are observed. To overcome the challenges in nano-scale technology, FinFETs were introduced in 1989 by Hisamoto *et al.* [3]. A device, named as depleted lean channel transistor (DELTA) at that time, was developed, which was composed of a vertical fin enveloped by two gate electrodes. A substantially reduced leakage current was observed and a reduction in short channel effects was also noticed in the proposed device design [50], thus improving the controllability of gate upon the channel [51, 52].

The commercialization of FinFETs started in the first decade of the 20th century with the invention of Omega FinFETs by Taiwan Semiconductor Manufacturing Company (TSMC) [53]. Later on, the FinFET evolution was followed by the development of 90 nm Bulk FinFETs by Samsung which was used in RAMs [54]. Commercialization of FinFETs on full scale started with the development of Intel's 22 nm [55] device which was used in the fabrication of 500 million ICs in first 3 years [18]. A number of other well established industries also started the production of FinFETs based memory chips, multi-level cell, SRAM, DRAM etc. FinFETs are expected to continue playing their role in the coming years. It is assumed that FinFET structure having the channel wrapped from three sides may reach to its limits beyond 5 nm. To overcome the issues, gate-all-around FET (GAAFET) may be the leading contender for next generation applications. It is understood that GAAFET can bring better scalability, faster switching time, lower leakage and high chip density. Moreover, vertically aligned nanowires and nanosheets are some other options to be explored for next generation devices [56].

2.1 FinFET Technology-An Overview

With the acceptance of FinFET idea in 1989, numerous researchers have proposed different FinFET structures to enhance electrical performance of the device. In 2000, Chang *et al.* [57] studied the relationship between L_g and V_{th} in a DG MOSFET. They concluded that a single gate device has higher V_{th} , thus, L_g scaling becomes a challenging task. Whereas, a DG device can be employed to get better control upon V_{th} , allowing further scalability of the device. To address V_{th} related issues, Choi *et al.* [58] developed a 20 nm DG FinFET, which helped further to mitigate the device scaling issues. In DG FinFETs a higher drive current of 365 μ A/ μ m compared to 270 μ A/ μ m in a typical MOSFET, was observed along with reduction in OFF state leakage current. Kim *et al.* [59] reported that by using trigate FinFETs, SCEs were reduced.

Sharma *et al.* [46] studied the effects of H_{fin} , T_{fin} , L_{fin} and L_g on RF performance of FinFETs. They observed that by decreasing L_g , an increase in gain and f_{max} is achieved. Furthermore, they observed that FinFETs have the potential to operate at relatively higher frequency compared to conventional FETs. Meshra *et al.* [60] developed 22 nm trigate FinFETs and studied the effects of H_{fin} and number of fins on the device characteristics. They concluded that, in order to improve the drive current, it is better to increase H_{fin} rather to increase the number of fins.

Gu *et al.* [61], analyzed 6-T and 4-T FinFETs based Static Random Access Memory (SRAM) cells and observed a 30 % less Static Noise Margin (SNM) in 6-T designs compared to MOSFET SRAM cell. Ensan *et al.* [62] also developed SRAMs using 21 nm FinFET technology and demonstrated that, SRAMs developed using FinFETs occupy less chip area and also consume less battery power. Mutterja *et al.* [63] developed logic gates using independent DG FinFET technology. They concluded that, an increased device density per unit area can be achieved using FinFETs compared to other established FET technologies.

Nirmal *et al.* [64] did a research on heterojunction FinFETs and observed a substantial reduction in SCEs. Liu *et al.* [65] studied self-heating effects in FinFETs and observed a reduction in channel current due to increased resistance observed at higher temperature possibly caused by the increased amplitude of lattice phonon.

Alvardo *et al.* [66] developed a low noise amplifier using FinFET technology. They extracted the device's intrinsic and extrinsic parameters and studied FinFETs potential as a microwave device. Gaillardon *et al.* [67] studied noise power in FETs and found that FinFETs offer relatively low noise power compared to conventional FETs. Thus, FinFETs based circuitry can potentially increase the battery life up to 25%.

Ledderer *et al.* [68] studied AC performance of 15 nm trigate FinFETs by extracting S-parameters upto 110 GHz. The effects of number of fins on intrinsic device parameters were studied and they concluded that by increasing the number of fins, f_{max} of a FinFET increases. Crupi *et al.* [69] showed that a FinFET has the capability of operating above 110 GHz. They also studied the effects of bias and number of fins on f_{max} . Kundu *et al.* [70], observed S-parameters of 65 nm FinFETs up to 50 GHz. They observed that f_{max} of the device is higher compared to conventional FETs because of the reduction in Miller capacitors and increase in the device g_m . Sakhi *et al.* [71] investigated the FinFETs behavior in subthreshold region. The effects of FinFETs dimensions were also studied on its operations.

Qin *et al.* [72] developed a model to simulate AC performance of FinFETs upto 300 GHz. They studied the effects of fin dimensions on the device AC characteristics. They also showed that L_g has a direct relationship with frequency, while

Author	Ref.	Year	Description
Chang <i>et al.</i>	[57]	2000	Effect of L_g on V_{th} .
Choi <i>et al.</i>	[58]	2001	20 nm DG FinFETs.
Kim <i>et al.</i>	[59]	2004	Improved SCEs in 30 nm DG FinFETs.
Lederer <i>et al.</i>	[68]	2005	S-parameters up to 110 GHz for $L_g=50$ nm.
Guo <i>et al.</i>	[61]	2005	SRAM design using FinFETs.
Crupi <i>et al.</i>	[69]	2006	Effects of number of fins on f_{max} .
Muttreja <i>et al.</i>	[63]	2007	Logic gates using independent DG FinFETs.
Kundu <i>et al.</i>	[70]	2010	S-parameters up to 50 GHz for $L_g = 65$ nm.
Sharma <i>et al.</i>	[46]	2012	Effects of FinFET dimensions.
Mishra <i>et al.</i>	[60]	2013	H_{fin} and its importance in FinFETs.
Nirmal <i>et al.</i>	[64]	2013	Reduced SCEs using dielectric material.
Alvarado <i>et al.</i>	[66]	2013	Low noise amplifier using FinFET.
Liu <i>et al.</i>	[65]	2014	Self-heating effects in FinFETs.
Gaillardon <i>et al.</i>	[67]	2016	Noise power in FinFETs.
Sakhi <i>et al.</i>	[71]	2017	Sub-threshold study of FinFETs.
Qin <i>et al.</i>	[72]	2018	Effect of L_g on S-parameters.
Ensan <i>et al.</i>	[62]	2019	SRAM cell using FinFETs.

TABLE 2.1: A brief history of FinFET technology.

there is an inverse relationship between frequency and T_{fin} . Furthermore, they observed that R_{ds} and R_{gd} decrease by increasing T_{fin} , while L_g has almost no effect on their values. A summary of FinFET technological advancements observed so far is presented in Table 2.1.

2.2 Evolution in Heterostructure FinFETs

By using two dissimilar semiconductors such as AlGaN/GaN one can have a structure known as heterostructure. Such a heterostructure generates 2DEG at its interface and can be used as channel of the device. In 2000, Ambacher *et al.* [73] determined 2DEG sheet carrier concentration, induced by polarization charges using a solution of Schrödinger and Poisson equations under appropriate boundary conditions in heterostructure devices. The study also included the dependence of 2DEG formation on piezoelectric and polarization induced effects.

Knowing the superior electrical qualities of 2DEG heterostructure, in 2002, Mishra *et al.* [74] studied market trends of different semiconductor devices, materials and their applications and predicted that heterostructure devices including FinFETs will dominate the market with attractive superiority in high power radar and wireless communication systems.

In 2013, Takashima *et al.* [75] investigated the effects of sidewall contribution on the device channel by fabricating AlGaN/GaN FinFETs. An improvement in SCEs was observed in the fabricated devices. Furthermore, they also investigated V_{th} dependence upon the T_{fin} and concluded that narrower channel is dominated by the side-gates of the device. Also, superior channel characteristics, such as reduced channel resistance and high g_m , are observed in FinFETs when compared with conventional FETs.

In 2014, Im *et al.* [76] fabricated AlGaN/GaN FinFETs. They studied 2DEG and sidewalls contribution in the channel current and observed excellent on-state performance in the developed devices. In 2015, Jo *et al.* [23] fabricated steeped

wall AlGaN/GaN FinFETs and reported extremely broad g_m , along with high on-state performance.

In 2015, Son *et al.* [24] fabricated multiple AlGaN/GaN FinFETs with various W_{fin} values and observed its effects on g_m , on-state and off-state performances of the device. Im *et al.* [77], in 2016, also fabricated multiple AlGaN/GaN Fin-FETs having different W_{fin} values and investigated temperature dependent DC performance. They investigated contribution of sidewall channels in the overall characteristics of wide and narrow FinFETs. Furthermore, a brief observation has also been carried out to assess g_m dependence on 2DEG and sidewall of the device.

In 2016, Ren *et al.* [78] studied heterostructure FinFETs and verified their results with TCAD simulations. The relationship between V_{th} and W_{fin} has been established using Poisson equation. The model also includes the influence of T_{ox} and dielectric permitivity of FinFETs.

Ghattak *et al.* [79] studied a V_{th} model for heterostructure FinFETs and compared the results with the experimental data. They proposed that in order to transform a normally-ON device into a normally-OFF device, W_{fin} can be scaled accordingly.

In 2017, Zhang *et al.* [80] fabricated heterostructure FinFETs featuring T-shaped gates and achieved linear g_m characteristics. They observed 1.45 times higher current density and 1.66 times higher power density at 8 GHz when compared to traditional HEMTs. In the following year, Zhang *et al.* [81] examined fin configurations in heterostructure FinFETs both theoretically and experimentally. A physics based model has been built for qualitative analysis whereas, the devices were fabricated for experimental verification. They studied g_m response in detail and established that L_{fin} has to be reduced for improved cutoff frequency. They also observed that g_m of a device depends upon H_{fin} and for improved performance an optimized value of H_{fin} is of prime importance.

In 2018, Xu *et al.* [32] simulated 2DEG characteristics in TCAD and proposed different optimized parameters for improved performance of heterostructure Fin-FETs. Their proposed solution suggested that the selection of appropriate device

Author	Ref.	Description
Ambacher (2000)	[73]	2DEG carrier concentration evaluaiton.
Mishra (2002)	[74]	Market trends of heterostructure devices.
Takashima (2013)	[75]	Sidewall contribution in AlGaN/GaN FinFETs.
Im (2014)	[76]	DC performance of AlGaN/GaN FinFETs.
Jo (2015)	[23]	To improve flatness in g_m of AlGaN/GaN FinFETs.
Son (2015)	[24]	Effects of W_{fin} on AlGaN/GaN FinFETs.
Im (2016)	[77]	Temperature effects on AlGaN/GaN FinFETs.
Ren (2016)	[78]	Physics based V_{th} model of AlGaN/GaN FinFETs.
Ghattak (2017)	[79]	V_{th} model for power AlGaN/GaN FinFETs.
Zhang (2017)	[80]	DC and RF performance of AlGaN/GaN FinFETs.
Zhang (2018)	[81]	Power performance of AlGaN/GaN FinFETs.
Xu (2018)	[32]	TCAD simulations of AlGaN/GaN FinFETs.
Ahmed (2020)	[82]	Analytical model of AlGaN/GaN FinFETs.
Huang (2020)	[83]	AlGaN/GaN FinFET for power applications.

TABLE 2.2: A brief overview of AlGaN/GaN based heterostructure FinFETs.

parameters ensures simultaneous activation of 2DEG and sidewall MOS channel formation, at positive V_{th} which is a requirement for normally-OFF operation.

Ahmed *et al.* [82] presented an analytical model for rectangular channel heterostructure FinFETs with three-sided Schottky barrier gate, by utilizing a 3D Poisson equation. In this study, the influence of trigate geometry on charge sheet carrier concentration has been assessed and authors reported that heterostructure FinFETs deplete more rapidly than ordinary HEMTs, owing to the additional field generated by the side gates. In 2020, Huang *et al.* [83] fabricated trigate HEMTs and observed improved on-off current, sub-threshold swing, and breakdown voltage compared to planar HEMTs. Because of these superior characteristics they carry higher potential to be used in next generation high power circuitries.

In a nutshell, it can be concluded that heterostructure FinFETs offer superior electrical performance and they are normally fabricated by using two distinct wide bandgap materials such as AlGaN/GaN. Based on the fin geometry, these devices can normally be ON or OFF as per the need of an application. The gate of heterostructure FinFETs can be placed directly on the semiconductor, alternatively they can be fabricated by using an intervening oxide layer. The characteristics of heterostructure FinFETs are primarily determined by the 2DEG, however, in some cases, it is observed that there is a formation of side channels which also contribute towards the performance of these devices. Those heterostructure FinFETs where side channels are also observed, they can accommodate a bigger gate swing of an incoming signal. Apart from the 2DEG, the performance of FinFETs as discussed above, is heavily dependent upon L_g , L_{fin} , H_{fin} and T_{fin} . It is established that these variables should be scaled appropriately for optimum performance of the device. An overview of AlGaN/GaN heterostructure FinFETs is presented in Table-2.2.

2.3 Carrier Transport Mechanism in FinFETs

Carriers transport in a semiconductor device has been the subject of significant interest over the last few decades. However, continuous downscaling of transistors has made this subject a more challenging job to handle. At nanoscale, it is neither conceptually nor mathematically a simple phenomenon. However, it is still fundamentally important to understand carriers transport mechanism in a nanoscale FinFET in order to describe its operation. It is an established fact that the response of a FinFET is primarily defined by the transportation of its carriers from source to drain which is controlled by the gate terminal of the device. This transportation is dependent upon various possible mechanisms such as:

- (a) drift of charge carriers under applied bias;
- (b) diffusion of carriers because of the density gradient;
- (c) thermionic emission of the carriers because of the channel temperature and
- (d) carriers tunneling under quantum effects.

If the dimension of a FinFET is such that it could be called as a long channel device, then the electrical response of the device would be controlled by the drifting carriers and the channel current will then be dependent upon the mobility offered by these carriers and its associated drift velocity, v_d . It is a known fact that, if we keep on increasing V_{ds} , the velocity saturates at a certain higher applied potential, which is a material dependent property. Therefore, based on FET channel material, the v_{sat} will have a limiting value and beyond that it would be independent from the applied bias. This leads to current saturation as a function of applied V_{ds} which is routinely observed in high frequency FETs [46]. Contrary to this, FinFET is a quantum device where apart from the drift current, special transportation phenomena are also taking place such as velocity overshoot and carriers tunneling [84, 85].

In case, the device current is predominantly associated with drifting of carriers, then a conventional approach is applied where the current is evaluated by involving carriers density and v_d . The carriers density as a function of applied bias can be evaluated by solving Poisson equation for a given device geometry. On the other hand, if L_{fin} is considerably small such that the wave function of one side (source) is overlapping with the other side (drain), for such a channel the carriers transportation falls under the definition of ballistic transport. The device electrical response can then be estimated by engaging Schrödinger wave equation wherein, the carriers of a source side are represented by a wave function which is overlapping to the drain side to allow quantum transport. The device behavior can mathematically be represented by engaging Poisson and Schrödinger equations simultaneously. Poisson equation gives the estimation of carrier density whilst Schrödinger wave equation describes its flow as a propagating wave inside the channel.

If a FinFET channel is defined by a 2DEG, then it is an established fact that such a device is a quantum device and its response can only be estimated by taking into consideration energy wave function representing the energy of the carriers available in 2DEG [84]. So in general, a nanoscale FinFET having a channel made by a semiconductor material or a 2DEG layer, in both the cases a quantum mechanics based carriers transportation will be observed as a dominating current contributor, and the device characteristics assessment requires a solution based on Schrödinger wave equation.

Apart from the channel conduction, FinFETs may have a current flowing from channel to the gate referred to as leakage current. Such a current can either be associated with thermionic emission (long channel devices) or with channel-togate tunneling. Such tunneling will only be observed if the channel wave function describing the movement of the carriers has penetration into the gate electrode. In FETs, channel-gate transportation should be as low as possible and ideally it should be zero.

Short channel FinFETs usually require complex mathematical formulation to represent their electrical response as a function of applied bias. A number of studies have been conducted in last two decades, listed in Table-2.3; wherein, researchers tried to investigate carriers transport mechanism for various types of FinFETs.

In 2004, Kim *et al.* [59] analyzed FinFETs performance using classical and quantum mechanical approaches. They investigated device performance by solving 2D Schrödinger-Poisson equations for different values of T_{fin} and validated their results with experimental data. Shao *et al.* [86] presented a quantum model for trigate FinFETs structures using ballistic transport. The model comprises of non-equilibrium Green's function (NEGF) and Schrödinger wave equation to investigate quantum transport and electron density profile. Khan *et al.* [87] developed a fully self-consistent quantum transport simulator using contact block reduction approach. A brief investigation related to H_{fin} variation and its impact on the device characteristics has been carried out by using the developed 2D & 3D simulators. Additionally, a comparative study has also been carried out on electron densities in DG and trigate FinFETs.

Natori *et al.* [88] came up with a compact transport model for nanowire FinFETs for the understanding of device operation. The research also discussed SCEs and temperature based performance of the device. They also studied quantum capacitance and quantum current and concluded that in linear region, the magnitude of current in a nanowire FinFET is primarily controlled by quantum capacitance. Dastjerdy [89] used NEGF approach to explore the features of nanowire FinFETs based on 3D quantum effects. The study concluded that Si nanowire performs better in FinFETs than a GAA channel.

Deyasi *et al.* [91] investigated the performance of GAA and nanowire FinFETs. They studied the effects of different types of gate structures on sub-threshold swing, channel length modulation, quantum capacitance and g_m , using Schrödinger-Poisson equations. In 2011, Sabry *et al.* [90] simulated FinFETs by involving quantum transport techniques and concluded that NEGF enables better design optimization of FinFETs and provides accurate prediction of its characteristics. Hsiao *et al.* [19] developed a 3D quantum transport model, based on the solution of NEGF and Poisson equation, to investigate electron transport in a GaAs GAA FinFETs. The reported solution considered electron-phonon, ionized impurity and surface roughness scatterings to assess the drain current. Zhang. *et al.* [92] studied the transport properties in nano FinFETs and their study concluded that transport properties are controlled by Fermi energies variation from source-drain sides of a FinFET.

The inventions of 3D devices has urged researchers to develop new electronic switches and circuits for improved performance. In 2022, Huang *et al.* [93] developed a flash memory involving 3D NOR gates. They investigated their proposed 3D device structure, integration scheme and circuit architecture to obtain 3D NOR flash devices. Their reported memory devices have large read current i.e.,

Author	Ref	Description
Kim (2004)	[59]	2D quantum transport model of DG FinFETs.
Shao (2005)	[86]	3D quantum transport model of DG FinFETs.
Khan (2008)	[87]	Ballistic transport model of trigate FinFETs.
Natori (2008)	[88]	Quantum capacitance study in ballistic NW FinFETs.
Dastjerdy (2011)	[89]	3D quantum model of NW FinFETs.
Sabry (2011)	[90]	Quantum transport technique in FinFETs.
Deyasi (2018)	[91]	Gate structure study in GAA FinFETs.
Hsiao (2019)	[19]	Elastic and inelastic scattering in GAA FinFETs.
Zhang (2019)	[92]	FET current pattern study at atomic scale.
Huang (2022)	[93]	Development of 3D NOR gate based memory.
Zhang (2022)	[94]	Scallop shape FinFETs for improved SCEs.
Cheng (2022)	[95]	SiGe vertically stacked GAA FinFETs.

TABLE 2.3: A brief overview of quantum transport mechanisms of FinFETs.

110 μ A/ μ m and by conducting writing-erasing test, they observed writing speed of 10 μ s and erasing speed of 100 ms.

Recently, Zhang *et al.* [94] investigated the performance of scallop-shaped Fin-FETs and attained an improved driving current compared to conventional Fin-FETs. Based on the observed characteristics they claimed that, scallop-FinFET is a promising candidate beyond current FinFET technology. Cheng *et al.* [95] explored fabrication and characterization of vertically stacked 4-level nanowire transistors and reported subthreshold slope of 77 mV/dec, drain induced barrier–lowering of 19 mV/V and $g_m = 83.35 \ \mu\text{S}/\mu\text{m}$ with improved SCEs.

2.4 Analytical Models

A model representing a semiconductor device response usually consists upon a mathematical expression based on the device material properties and its geometry. The model will be an acceptable one if it can generate the device characteristics, as a function of applied bias, same as that of actual device data. A device model can either be a mathematical relationship based upon the device physics or an empirical relation dependent on the device geometrical parameters. A physics based model is referred to as an analytical model, which uses a set of equations that best describe the behavior of a given device [25]. Such a model provides a reasonably good understanding about the physical operation of the device. Involving device physics, in a comprehensive way, in model development, usually leads to a complex mathematical formulation, and an analytical solution of such a formulation becomes a tedious job, which is usually not preferred by a design engineer.

On the other hand, an empirical technique, based on the device geometry, can be employed to predict its behavior. However, such a technique provides limited understanding pertaining to the device's electrical response as a function of applied bias. In an empirical technique, the device geometrical parameters are used in a mathematical expression in such a way that, by changing the device applied bias, the response received from the developed empirical expression is the same as that of the device experimental characteristics [2, 102]. Thus, an empirical technique provides a reasonable ease to model the device characteristics and therefore it is a preferred approach in CAD applications.

In general, in an empirical model there are usually fitting variables and involvement of large number of fitting variables is considered a negative feature of an empirical model. It is therefore always preferred to develop an empirical expression to generate a device characteristics by involving: a) device geometrical variables and for

Author	Ref	I - V Model
Taur (2004)	[96]	$I_{ds} = 2\mu \frac{W}{L_g} C_{ox} \left(2V_T\right)^2 \left[\beta tan\beta - \frac{\beta^2}{2} + r\beta^2 tan^2\beta\right]$
Conde (2005)	[97]	$I_{ds} = 2\mu \frac{W}{L_g} C_{ox} \left[V_{GF} \left(\psi_d - \psi_{so} \right) - \frac{1}{2} \left(\psi_d^2 - \psi_{so}^2 \right) \right]$
		$+\mu \frac{W}{L_g} C_{ox} 4 V_T (\psi_d - \psi_{so})$
		$+t_{si}k_bTn_i\left[e^{\beta\left(\psi_{0L}-V_{ds}\right)}-e^{\beta\psi_0}\right]$
Nikolaos (2012)	[98]	$I_{ds} = 2\mu \frac{W}{L_g} \frac{\epsilon_{ox}}{T_{ox}} (2V_T)^2 \left[q_{is} - q_{id} + \frac{1}{2} (q_{is}^2 - q_{id}^2) \right]$
Paydavosi (2013)	[99]	$I_{ds} = 2\mu \frac{W}{L} \bigg[f(\psi_s) - f(\psi_d) \bigg]$
		$f(\psi_s) = \frac{Q_{inv}^2}{2C_{ox}} + 2V_T Q_{inv} - V_T \left(5V_T \frac{\epsilon_{si}}{t_{si}} + Q_{bulk}\right)$
		$\times \ln \left(5V_T \frac{\epsilon_{si}}{t_{si}} + Q_{bulk} + Q_{inv} \right)$
		$f(\psi_d) = \frac{Q_{inv}^2}{2C_{ox}} + 2V_T Q_{inv} - V_T \left(5V_T \frac{\epsilon_{si}}{t_{si}} + Q_{bulk}\right)$
		$\times \ln \left(5V_T \frac{\epsilon_{si}}{t_{si}} + Q_{bulk} + Q_{inv} \right)$
Kumari (2017)	[100]	$I_{ds} = \frac{k_B T \mu n_i \left[1 - exp\left(\frac{-qV_{ds}}{k_B T}\right) \right]}{A + B}$
		$A = \int_{0}^{L_{u}} \frac{1}{\int_{0}^{H_{fin}} \int_{0}^{T_{fin}} exp\left(\frac{q\psi_{tch1}}{kT}\right) dxdz} dy$

$$B = \int_{L_u}^{L_g} \frac{1}{\int_0^{H_{fin}} \int_0^{T_{fin}} exp\left(\frac{q\psi_{tch2}}{k_BT}\right) dxdz}} dy$$

Ahmed (2019) [101]
$$I_{ds} = 2\mu(V) \frac{W}{L_g} (\psi_d - \psi_s)$$
$$\psi_s \approx -\frac{Q_{invs}^2}{2C_{ox}} - 2V_T Q_{invs} - 2V_T Q_{bulk} \ln(Q_{invs} + 2Q_{bulk})$$
$$+ 2V_T Q_{zs} + 2V_T \frac{Q_{bulk}}{Q_{invs} + 2Q_{bulk}} Q_{zs}$$
$$\psi_d \approx -\frac{Q_{invd}^2}{2C_{ox}} - 2V_T Q_{invd} - 2V_T Q_{bulk} \ln(Q_{invd} + 2Q_{bulk})$$
$$+ 2V_T Q_{zd} + 2V_T \frac{Q_{bulk}}{Q_{invd} + 2Q_{bulk}} Q_{zd}$$

the case of FinFET they are: L_{fin} , T_{fin} , H_{fin} , L_g ; b) applied bias V_{gs} , V_{ds} , and c) material electrical properties such as mobility μ , carrier velocity v_d etc.

Taur *et al.* [96], in 2004, presented an analytical model for DG FETs. The model consists of closed form solution of Poisson and current continuity equations and operates for all region of operations. They demonstrated the validity of their model by comparing the modeled output characteristics with 2D numerical solution without involving any fitting variables.

Conde *et al.* [97] presented a DC model for undoped trigate FinFETs. The study proposed a Poisson based single explicit analytical equation valid for all bias conditions and counts for drift and diffusion transports. The model is valid for long channel devices and showed deterioration when applied to short channel FinFETs [97].

In 2012, Nikolaos *et al.* [98] presented an analytical solution to predict DC characteristics of lightly doped FinFETs, taking into account quantum mechanical and

LABLE Z.						ary ucau II			n researc	ners.	1 K
del	Ref.	L_g	T_{fin}	H_{fin}	V_{th}	T_{ox}	\sim	V_T	μ	ΔL_g	N_i
ur (2004)	[96]	>	>	×	×	>	×	>	>	×	>
nde (2005)	[26]	>	>	×	>	>	>	>	>	×	>
ij (2008)	[103]	>	>	×	>	>	×	>	>	×	×
sayan (2011)	$\left[104 ight]$	>	>	>	×	>	×	>	>	>	×
sh (2012)	[105]	>	>	×	>	>	×	>	>	>	>
sarakis (2012)	[98]	>	>	>	>	>	×	>	>	>	×
ydavosi (2013)	[66]	>	>	×	>	>	>	>	>	×	>
arte (2013)	[106]	>	>	×	>	>	>	>	>	×	×
mari (2017)	[100]	>	>	>	×	>	>	>	×	>	>
med (2019)	[101]	>	>	>	×	>	×	>	>	×	>

used in the development of DC analytical models by different researchers 00000 TARLE 2.5. Pars short channel effects. The proposed model used unified expressions for inversion charge and drain current which are valid for all regions of operation.

In 2013, Paydavosi *et al.* [99] presented first industry-standard compact model to simulate double, triple and GAA FinFETs. All important real device effects, such as short channel effects, quantum mechanical effects, mobility degradation and parasitic components effects are included in the model. In 2017, Kumari *et al.* [100] used a quasi 3D analytical model to predict FinFETs characteristics. In 2019, Ahmed *et al.* [101] developed a surface potential model of FinFET using a Poisson equation. A summary of FinFETs model expressions discussed hitherto are presented in Table 2.4. It is observed that most of the reported models, other than Ahmed [101], used a 2D Poisson equation, whereas a FinFET is a 3D device and to predict an accurate device response it is important that the 3D geometry should be taken into consideration.

A comprehensive literature survey is conducted for geometrical parameters that are used in various reported models. A number of researchers tried to present device models by incorporating device geometry, the detail of which is listed in Table 2.5. As evident from the table, almost all the researchers used L_g , T_{fin} and V_{th} , however, H_{fin} is mostly ignored because they simplified the problem by considering 2D geometry of the device. Therefore, the results generated by these models are prone to error when z-directed field increases with the decrease in device dimensions.

2.5 Estimation of FinFETs Characteristics by AI

In this era, computational power has observed phenomenal growth and use of computational based intelligence has attained an important role in every field of life including medicines, engineering, traffic control, town planning, oceanography, forestry etc. Such an intelligence where human decision making and planning is assisted by a machine, which is trained by an expert using appropriate configuration, design and data, is called artificial intelligence (AI). Training a computational technique by a pre-defined scheme, using appropriate data, enables the machine to predict futuristic response with reasonable accuracy and hence the technique save a lot of resources without compromising on the accuracy of the targeted decision. Owing to these, AI based prediction is getting popularity in almost every field of life including semiconductor technology.

Device fabrication requires a high-tech industry and apart from that, to get a finish device, starting from its substrate to the end product, is a time consuming process. Changing a device geometrical variables, results into changed characteristics. Such a change in the device characteristics would be possible to predict, prior to its fabrication, if an appropriate trained AI tool is available. In recent past, AI based solutions have attracted strong attention and have been applied successfully in device modeling. In this process, a high-end computational resource would be required to train artificial neural networks (ANN), wherein a large number of neural units and/or regression weights are set and calibrated by multiple iterations (Epochs). Once the training is completed successfully with a good degree of accuracy, the AI system, afterwards can work efficiently with conventional computational resources. AI based prediction requires less memory, is more efficient and achieves a good level of accuracy, in relatively less time, compared to analytical and numerical techniques. However, AI model training requires big volume of data and quite a few iterations of training to get an acceptable accuracy. Table-2.6 summarizes AI based modeling techniques reported so far in the last two decades.

In 2009, Alam *et al.* [107] developed an ANN technique for small signal analysis of FinFET and based on the proposed model, they also designed a low noise amplifier. For the proof of the concept, they compared ANN results with a standard circuit simulation tool known as advance design software (ADS). In 2011, Marinkovic *et al.* [108], modeled S-parameters of different technologies such as GaAs HEMTs, GaN HEMTs and Si FinFETs for operating frequencies of 2 to 50 GHz using an AI technique. The AI based estimated modeled parameters were obtained as a function of bias, frequency, temperature and the device geometry. To validate their technique, they compared AI results with experimental data which showed a reasonable compliance. Additionally, the reported technique was also applied

Author	Ref.	Description
Alam (2009)	[107]	Trained AC parameter of nano FinFETs.
Marinkovic (2011)	[108]	Trained RF performance of Si FinFETs.
Chen (2014)	[109]	Extrinsic capacitances of nano FinFETs.
He (2017)	[110]	Trained electrical characteristics of nano FinFETs.
Choi (2020)	[111]	Physical parameters optimization of Si MOSFETs.
Mehta (2020)	[112]	DC and AC performance of FinFETs.
Ghoshhajra (2022)	[113]	Improved performance of FinFETs.

 TABLE 2.6: A summary of AI based FETs techniques developed to predict the device characteristics.

successfully both on real and imaginary parts of the S-parameters independently, to same level of accuracy.

In 2014, Chen *et al.* [109] proposed a ML technique for the prediction of extrinsic capacitances of nano FinFETs. The dataset used for the training of the proposed network is taken from a quasi static electromagnetic software [114] by varying the geometrical parameters of FinFETs. In 2017, He *et al.* [110] evaluated DC characteristics of FinFETs using an ANN approach. They compared the results with the experimental data after the completion of training process. However, the validation of the proposed technique, as far as prediction of the device characteristics is concerned, is not reported by the authors. In 2020, Choi *et al.* [111] studied the manufacturing process of FETs and proposed an AI based technique to optimize it. They used a machine learning approach along with an holistic optimization technique in order to reduce the fabrication cost and time.

Mehta *et al.* [112] proposed a machine learning approach to predict the DC and CV characteristics of FinFETs, by taking datasets from simulated devices

using TCAD. They demonstrated the possibility to predict the device DC and CV characteristics without involving the device physics. However, they did not show the validity of their model using experimental data. Ghoshhajra. *et al.* [113] in 2022, presented a similar approach to predict the device DC characteristics by creating a mapping technique between input and targeted output characteristics of FinFETs. They employed an ANN, wherein they opted multiple input and single output approach and managed to predict the device DC characteristics with 96.5% accuracy after training their ANN with TCAD simulated data.

Numerous researchers, as listed in Table 2.6, claimed the prediction of FinFETs characteristics using ANN techniques. However, none of them studied the variation of the device characteristics as a function of its physical parameters, in an holistic way, such as: L_g ; H_{fin} ; T_{fin} ; L_{fin} ; T_{ox} and N_d . Thus, there is a need to develop a comprehensive ANN model, based on these parameters, capable to predict the device characteristics for any chosen device dimensions.

2.6 Thesis Motivation

Since FinFET technology is relatively new, there are numerous challenges associated to it for optimized device performance and understanding. Based on the research review carried out in the preceding sections, following research gaps, to be investigated in this thesis have been identified:

2.6.1 Research Gaps

a) Heterostructure AlGaN/GaN FinFETs are popular both for high frequency and high power applications. The output characteristics of AlGaN/GaN FinFETs are mainly driven by the 2DEG created at the interface. The current associated with the 2DEG could be referred to as I_{2D} , which mainly determines the depletion mode operation of such devices. On the other hand, in enhancement mode, the literature shows that the device is operational at unusually higher gate bias $V_{gs} \geq 2$ V. Maintaining its characteristics at such a high gate bias indicates that, apart from 2DEG, some parallel conducting channels are also appearing, plausibly because of carriers inversion layer, which is potentially contributing into the drain current called I_{inv} . Thus, the total drain current would be $I_{ds} = I_{2D} + I_{inv}$. To evaluate I_{ds} of such devices where inversion and depletion both processes are taking place by varying the gate potential, there is a need to re-visit the device basic understanding. The literature survey showed that there is no such mathematical tool or model available, which can assess I_{ds} involving both I_{2D} and I_{inv} in a single Fin-FET. Thus, any attempt to develop such a model would be a contribution towards the understanding of nanoscale FinFETs capable to operate both in inversion as well as in depletion modes simultaneously.

b) To accommodate high frequency operation and to minimize SCEs, trigate Fin-FETs in nanoscale regime are fabricated. When $L_g \leq 10$ nm, the carrier transport is ballistic in nature and a solution of 3D Schrödinger wave equation as a function of applied bias is required to represent the transportation of carriers. The number density of these carriers can be known by solving a 3D Poisson equation under appropriate boundary conditions. Coupling 3D Schrödinger equation with 3D Poisson equation can lead to the evaluation of $I_{ds}(V_{ds}, V_{gs})$. So such a 3D technique to asses carriers responsible for current contribution should be more accurate in nature because it reflects the true geometry of the device. According to our assessment no such attempt has been reported in the literature where 3D Schrödinger wave equation and 3D Poisson equation have been solved simultaneously for nanoscale FinFETs with its experimental verification. Any such development will enhance the fundamental understanding of FinFET channel operation at nanoscale regime.

c) As discussed in Section 2.5, researchers in these days are actively involved in predicting FinFETs characteristics by developing ANN models. However, according to our survey, no ANN based model has yet been reported which involves all major device physical parameters such as L_g , H_{fin} , L_{fin} etc, to predict its characteristics. The development of any such model by considering all possible physical

parameters, which are required to be tunned by a design engineer for target characteristics, would be beneficial for the industry. It would save industrial time and resources, provided the model has the ability to predict the device characteristics as a function of target physical variables prior to its fabrication. In this respect, it is realized that the developed AI tool should be robust enough to predict the device output and transfer characteristics as a function of its fabrication parameters to a good degree of accuracy. Additionally, keeping in view the basic philosophy of AI, it can comfortably be claimed that, it would be an efficient technique compared to other such tools which are based on numerical solutions e.g., TCAD and COMSOL.

2.6.2 Problem Statement & Methodology

Based on the research gaps presented in the preceding section, following problem statements and associated research methodologies have been finalized to carry out the research work to be presented in this thesis.

a) To develop a comprehensive model to predict DC characteristics of heterojunction FinFETs both for depletion and inversion modes of operation

In this part of research, it is proposed that a model which can predict the DC characteristics of nanoscale heterojunction FinFETs both for depletion as well as for inversion modes of operation simultaneously, will be developed. It is an established fact that for depletion mode devices at $V_{gs} = 0$ V, $I_{ds} \neq 0$ A and at $V_{gs} = V_{th}$, $I_{ds} = 0$ A, and when the device is operated in enhancement mode, under such scenario, V_{gs} will have a polarity opposite to that of depletion mode V_{gs} polarity. However, V_{gs} in enhancement mode of operation, shall allow a limited variation, i.e., $V_{gs} \lesssim \phi_B$ where ϕ_B is the built-in potential. If the device maintains its operation at $V_{gs} > \phi_B$, this is an indication that a new conducting path is created inside the device which is contributing in the flow of current beside depletion mode current. Such a conducting path may be associated to the accumulation of inversion carriers at the interface of the gate, connecting

the drain-source terminal. Under such circumstances, it can be assumed that $I_{ds} = I_{2D} + I_{inv}$. To develop a model by incorporating this concept, one can proceed first to evaluate I_{2D} by considering carrier concentration (n_s) defining 2DEG of the heterojunction. Once n_s is known for the given geometry, the linear as well as the saturation current can be assessed using device physical dimensions and field distribution inside the channel. On the other hand, inversion mode current can be evaluated independently by considering the device as MOS structure where at the interface, inversion layer is created, contributing into the flow of current. At the end of evaluation, both the concepts, 2DEG and MOS, can be coupled together to develop a unified model which could be referred to as depletion/inversion model of FinFETs. For both the cases i.e., depletion and inversion, a standard Poisson equation shall be solved using appropriate boundary conditions and the accuracy of the solution along with the validity of the developed model can eventually be assessed by comparing the modeled and experimental data of devices with known dimensions.

b) To develop a 3D analytical model based on ballistic transport mechanism for output characteristics of nano-FinFETs

In this part of research, it is proposed that by involving ballistic carriers transport, which contributes significantly in nano FinFETs, one can develop a 3D model to predict output characteristics of such devices. In ballistic carriers transport, the carriers are represented by a wave function, so a 3D Schrödinger wave equation is required to be solved involving the device geometry and Fermi energy variations from source to drain. Since, the Schrödinger wave equation depends upon the potential energy of the system, which changes because of the applied potential, it is considered to be a difficult task to arrive at some analytical solution of the 3D Schrödinger wave equation representing the FinFET channel. Hence, an appropriate numerical technique will be applied to achieve a workable solution. On the other hand, the carriers, which are contributing in the channel current, would be evaluated possibly by solving a 3D Poisson equation for the given device geometry and bias conditions. Coupling Schrödinger-Poisson solutions, the output current as a function of applied bias will then be evaluated. The validity of the model
will be ensured by comparing the modeled and experimental characteristics for a range of FinFETs.

c) To develop an efficient and easy to handle AI technique to predict FinFETs characteristics

The semiconductor technology is considered to be an expensive technology and in order to cut the device cost, there is always a need before arriving at the final recipe, to predict the target device characteristics prior to its fabrication. Device characteristics prediction can be realized by designing an intelligent AI based system. To develop an ANN system for any given problem, a dataset of reasonable size is required, to train the network for future prediction. In this respect, it is proposed that a dataset of DC characteristics can be generated using an established software tool (COMSOL, TCAD etc) by varying the device physical dimensions such as L_g , H_{fin} , T_{fin} , L_{fin} , etc. Distinct subsets of the data will be used for training and testing of the network to achieve an acceptable accuracy. After successful training, the network will be able to predict the device characteristic by using an input vector comprising of the same parameters as that of training. The accuracy of ANN model will be checked by using standard tool such as COMSOL. It is a known fact that the simulation tools based on numerical solutions require high computational resources and they are heavy in time. However, an ANN based system, after achieving the required training accuracy, should be able to predict the device characteristics in exceptionally less time. Hence, the technique can potentially be employed to reduce product cost and industrial process time.

2.7 Summary

In this chapter, a comprehensive literature survey of FinFETs, their operations, associated materials and device analytical models are presented. It has been shown that FinFET geometry and fabrication material are the two striking parameters which determine its characteristics. It has also been established, by analyzing numerous research studies of heterostructure FinFETs, that 2DEG created by

two dissimilar semiconductors offers improved FinFETs characteristics. Transport mechanism of carriers has also been discussed exhaustively for nanoscale FinFETs, which plays an important role in determining their operations. Moreover, several mechanisms which influence the transportation of carriers in FinFETs and their ultimate impact on output current have been discussed. Models developed by various researchers involving 2D and 3D device geometry are also presented along with their limitations. The advancements made by AI in semiconductor industry in the recent past has been discussed especially, with respect to FETs technologies. It has been observed that AI models developed so far have limitations. This is because they do not incorporate all crucial geometrical parameters that define the performance of FinFETs. Hence these models are prone to error. Lastly, based on the conducted literature survey, some research gaps have been identified and hypotheses for the proposed research and associated methodologies have been presented.

Chapter 3

A Unified Depletion/Inversion Model for Heterojunction Trigate FinFETs DC Characteristics

3.1 Introduction

In recent years, FinFETs have become the gold standard in the electronic industry due to their superior properties in comparison to conventional FETs [115, 116]. Their trigate nature suppresses the short channel effects [117], which consequently improves the device transconductance, reduces its leakage current and enhances the device scalability [118–120]. AlGaN/GaN FinFETs have the potential to further supplement FET development owing to their improved 2DEG characteristics and temperature stability [21, 77, 80]. Since they are fabricated using wide bandgap materials, these devices have the potential to offer stable characteristics at elevated temperatures, and thus lead the way for high density electronics [121–125]. The trigate nature of the FinFET incorporates all the advantages of the bulk FinFET, while the AlGaN/GaN heterojunction, which forms the 2DEG, gives higher carrier concentration [126], higher saturation velocity [127] and superior temperature-dependent performance [128].



FIGURE 3.1: 3D view of an AlGaN/GaN FinFET channel.

To help the device attain its maximum potential, mathematical models which correlate the effects of the physical dimensions of the device and its characteristic are required. Fig. 3.1 shows a 3D view of an AlGaN/GaN FinFET. To model such a device, the effects of the trigate must be taken into account while evaluating the current flow between the drain and source electrodes. The total current in such a device is the combination of 2DEG and side channel (inversion) currents [32], which must be evaluated separately for accurate computation of drain current.

In this chapter, a comprehensive I - V model is proposed, which can predict the output and transfer characteristics of trigate AlGaN/GaN FinFETs both in the depletion as well as in the inversion mode of operations. The proposed model uses a modified definition of n_s , presented in Ref. [11], to cater for the effects of the side gates on 2DEG depletion. Inversion current is modeled using the potential distribution acquired by the solution of a 2D Poisson equation. The total current is the sum of both the inversion and 2DEG currents. To the best of our knowledge, there is no such model reported in literature which can assess both depletion/inversion characteristics simultaneously. Authors of Ref. [78, 79, 129] have established models based on the physics of the device to assess the effects of FinFET geometry on threshold voltages, but not the entire I - V characteristics.

In Ref. [82], an attempt has been made to present an analytical model for trigate AlGaN/GaN FinFETs which have no oxide layer beneath the gate electrode. Such a device cannot operate at a relatively high forward bias because of high leakage current and, as a result, there is no possibility of inversion layer formation at the oxide-metal interface. Hence, a model developed for such devices will have limitations in predicting I - V characteristics under inversion. Therefore, the model presented in Ref. [82] which involves the 2DEG current alone, can predict I - V characteristics under the depletion mode only and ceases to work when the device operates in the inversion mode.

3.2 AlGaN/GaN FinFET Model

To evaluate the drain to source current of an AlGaN/GaN FinFET, consider Fig. 3.1. The band diagram of the heterostructure FinFET is given in Fig. 3.2, Because of the bandgap discontinuity between AlGaN and GaN layers, 2DEG is created at the interface, which will be available for conduction and its thickness can be controlled by the gate. Hence the structure of the device by and large allows operation in the depletion mode and the device under consideration is an *n*-type channel. In this mode, 2DEG would be the major contributor to the channel current. As V_{gs} increases, 2DEG begins to un-deplete, while activating the inversion process, which will eventually define the side channels. Thus, the total current in such a FinFET would be a combination of both 2DEG and inversion currents represented by

$$I_{ds} = I_{inv} + I_{2D} \tag{3.1}$$

While proceeding with the model, it is assumed that I_{inv} and I_{2D} are independent of each other and can hence be modeled separately. To get total I_{ds} of a trigate AlGaN/GaN FinFET, both currents can be summed together as given by Eq. (3.1). It is further assumed that; a) inversion only occurs on the two side gates of trigate FinFET; b) 2DEG gets depleted independently from the three sides of the gate and c) carriers of 2DEG and those in the inversion layer move with their



FIGURE 3.2: Energy band diagram of Heterostructure FinFETs.

respective mobilities. The proposed model is a steady-state model since it does not predict transient response of the device and its validity is for $V_{ds} \ge 0$ and $V_{gs} \ge V_{th}$.

3.2.1 2DEG Current

Consider the x - z cross-section of the device as illustrated by Fig. 3.3, wherein it is evident that 2DEG is confined in the x - y dimension and the gate of the device depletes the 2DEG carriers from the x - z plane. Hence, the y dimension has no role in determining the current of the device. Consequently, the Poisson equation in 2D can be applied on the device geometry as shown below

$$\frac{\partial^2 V}{\partial x^2} + \frac{\partial^2 V}{\partial z^2} = -\frac{q}{\epsilon_A} N_d \tag{3.2}$$

Since 2DEG acts as a sheet with infinitesimal width in the z direction, and with the assumption that both the top and side gates act on it independently, the applied voltage on the 2DEG can be divided as

$$V = V_1(x) + V_2(z)$$
(3.3)

Since

$$\frac{\partial^2 V_2}{\partial z^2} = -\frac{qN_d}{3\epsilon_A} \tag{3.4}$$



FIGURE 3.3: x - z cross-sectional view of an AlGaN/GaN FinFET.

Integration of above equation results in

$$\frac{\partial V_2}{\partial z} = -\frac{qN_d}{3\epsilon_A}z + c_1 \tag{3.5}$$

and

$$\left. \frac{\partial V_2}{\partial z} \right|_{z=0} = \frac{q}{\epsilon_{eff}} \left(\sigma - n_s \right) \tag{3.6}$$

where, σ is the sheet charge density and ϵ_{eff} is the effective permittivity of 2DEG as defined in Ref. [130]. c_1 can be found from above equation as,

$$c_1 = \frac{q}{\epsilon_{eff}} \left(\sigma - n_s \right) \tag{3.7}$$

so using c_1 , Eq. (3.5) becomes,

$$\frac{\partial V_2}{\partial z} = -\frac{qN_d}{3\epsilon_A}z + \frac{q}{\epsilon_{eff}}(\sigma - n_s)$$
(3.8)

Integrating Eq. (3.8) once again to get V_2 ,

$$V_2 = -\frac{qN_d}{6\epsilon_A}z^2 + \frac{q}{\epsilon_{eff}}(\sigma - n_s)z$$
(3.9)

Now from Eqs. (3.4) and (3.2), one can write

$$\frac{\partial^2 V_1}{\partial x^2} - \frac{qN_d}{3\epsilon_A} = -\frac{qN_d}{\epsilon_A} \tag{3.10}$$

therefore,

$$\frac{\partial^2 V_1}{\partial x^2} = -\frac{2qN_d}{3\epsilon_A} \tag{3.11}$$

Integrating Eq. (3.11),

$$\frac{\partial V_1}{\partial x} = -\frac{2qN_d}{3\epsilon_A}x + c_2 \tag{3.12}$$

Using the boundary conditions

$$\left. \frac{\partial V_1}{\partial x} \right|_{x=T_{fin}/2} = \frac{q}{\epsilon_{eff}} \left(\sigma - n_s \right) \tag{3.13}$$

 c_2 can be found by substituting Eq. (3.13) into Eq. (3.12)

$$c_2 = \frac{q}{\epsilon_{eff}} \left(\sigma - n_s \right) + \frac{2qN_d T_{fin}}{3\epsilon_A \times 2} \tag{3.14}$$

Using c_2 in Eq. (3.12) and integrating it once again to find V_1

$$V_1 = -\frac{qN_d}{3\epsilon_A}x^2 + \frac{q}{\epsilon_{eff}}(\sigma - n_s)x + \frac{qN_dT_{fin}}{3\epsilon_A}x$$
(3.15)

Using V_1 from Eq. (3.15) and V_2 from Eq. (3.9) into Eq. (3.3) to find the applied voltage on 2DEG,

$$V = -\frac{qN_d}{3\epsilon_A}x^2 + \frac{q}{\epsilon_{eff}}(\sigma - n_s)x + \frac{qN_dT_{fin}}{3\epsilon_A}x - \frac{qN_d}{6\epsilon_A}z^2 + \frac{q}{\epsilon_{eff}}(\sigma - n_s)z$$
(3.16)

Now, considering the device geometry, V at z=d and $x=T_{fin}/2$

$$V = \phi_b - V_{gs} + V(y) + \left(\frac{E_F - \Delta E_c}{q}\right)$$
(3.17)

Here, $V(y=0) = V_s$ and $V(y=L_g) = V_d$. Now, comparing Eqs. (3.17) and (3.16)

$$\phi_{b} - V_{gs} + V(y) + \left(\frac{E_{F} - \Delta E_{c}}{q}\right) = -\frac{qN_{d}}{12\epsilon_{A}}T_{fin}^{2} + \frac{q}{\epsilon_{eff}}(\sigma - n_{s})T_{fin} + \frac{qN_{d}T_{fin}}{6\epsilon_{A}}T_{fin} - \frac{qN_{d}}{6\epsilon_{A}}d^{2} + \frac{q}{\epsilon_{eff}}(\sigma - n_{s})d$$
(3.18)

$$n_s = \frac{\epsilon_{eff}}{q(T_{fin}/2 + d)} \left[V_{gs} - V_{th} - \frac{E_F}{q} - V(y) \right]$$
(3.19)

here,

$$V_{th} = \phi_b - \frac{\Delta E_c}{q} + \frac{qN_d}{6\epsilon_A} d^2 - \frac{qN_d}{12\epsilon_A} T_{fin}^2 - \frac{\sigma q \left[T_{fin}/2 + d \right]}{\epsilon_{eff}}$$
(3.20)

and E_F can be expressed as a function of n_s as [11]

$$E_F = \zeta_1 + \zeta_2 n_s^{1/2} + \zeta_3 n_s \tag{3.21}$$

where $\zeta_{1,2,3}$ are temperature-dependent parameters. So Eq. (3.19) can be written as,

$$n_{s} = \frac{\epsilon_{eff}}{q(T_{fin}/2 + d)} \left[V_{gs} - V_{th} - \frac{\zeta_{1}}{q} - \frac{\zeta_{2}n_{s}^{1/2}}{q} - \frac{\zeta_{3}}{q}n_{s} - V(y) \right]$$
(3.22)

Let $\lambda_1 = \zeta_1/q$; $\lambda_2 = \zeta_2/q$; $\lambda_3 = \zeta_3/q$, which modifies Eq. (3.22) as

$$n_s = \frac{\epsilon_{eff}}{q(T_{fin}/2+d)} \left[V_{gs} - V_{th} - \lambda_1 - \lambda_2 n_s^{1/2} - \lambda_3 n_s - V(y) \right]$$
(3.23)

Using quadratic manipulation

$$n_s = \left[\frac{-\lambda_2 \pm \sqrt{\lambda_2^2 + 4\lambda_4 (V_{gs} - V_{th} - V(y) - \lambda_1)}}{2\lambda_4}\right]^2 \tag{3.24}$$

where,

$$\lambda_4 = \lambda_3 + \frac{q(T_{fin}/2 + d)}{\epsilon_{eff}} \tag{3.25}$$

Eq. (3.24) is referred to as a modified definition of n_s for trigate FinFET and can be used to assess 2DEG current as

$$I_{2\mathrm{D}} = qW n_s \upsilon_d(y) \tag{3.26}$$

where,

$$\upsilon_{d}(y) = \begin{cases} \frac{\mu_{0}E_{y}}{1 + E_{y}/2E_{sat}}, & \text{for} \quad E_{y} = dV(y)/dy < 2E_{sat} \\ \upsilon_{sat}, & \text{for} \quad E_{y} = dV(y)/dy > 2E_{sat} \end{cases}$$
(3.27)

3.2.1.1 Linear Region Current

In the linear region of operation, the device behaves as a resistor such that the current is directly proportional to the applied field which falls in the direction of the current. Keeping in view, the geometry shown in Fig. 3.1, the field component E_y is responsible to control the current flowing between drain & source, and the total linear current for this region of operation can then be obtained by integrating the current expression over the gate length, L_g , as shown below

$$\int_{o}^{L_{g}} I_{2\mathrm{D}(l)} dy = \int_{o}^{L_{g}} qW n_{s} \frac{\mu_{0} E_{y}}{1 + E_{y}/2E_{sat}} dy$$
(3.28)

$$I_{2\mathrm{D}(1)}L_{g} = \int_{o}^{L_{g}} qW n_{s} \frac{\mu_{0}E_{y}2E_{sat}}{2E_{sat} + E_{y}}Eydy$$
(3.29)

As $E_y = dV/dy$; $V_L = E_{sat}L_g$; $v_{sat} = \mu_0 E_{sat}$; $V_{ds} = E_y L_g$, changing limits,

$$I_{2D(l)}L_g = qW \int_o^{L_g} n_s \frac{2v_{sat}}{2V_L + V_{ds}} \frac{dV}{dy} dy$$
(3.30)

Let $\alpha = qWv_{sat}/\lambda_4$ so, Eq. (3.30) becomes

$$I_{2D(l)}L_g = \frac{2\lambda_4 \alpha}{2V_L + V_{ds}} \int_0^{V_{ds}} n_s dV$$
(3.31)

Assume, $V_{g1} = V_{gs} - V_{th} - \lambda_1$ and substituting n_s from Eq. (3.24) to Eq. (3.31)

$$I_{2\mathrm{D}(1)}L_{g} = \frac{2\lambda_{4}\alpha}{2V_{L} + V_{ds}} \int_{0}^{V_{ds}} \left[\frac{-\lambda_{2} \pm \sqrt{\lambda_{2}^{2} + 4\lambda_{4} \left[V_{g1} - V(y)\right]}}{2\lambda_{4}}\right]^{2} dV \qquad (3.32)$$

$$I_{2D(l)} = \frac{\alpha}{2\lambda_4 (2V_L + V_{ds})} \int_0^{V_{ds}} \left(\lambda_2^2 + 4\lambda_4 \left[V_{g1} - V(y)\right] + \lambda_2^2 - 2\lambda_2 \sqrt{\lambda_2^2 + 4\lambda_4 \left[V_{g1} - V(y)\right]}\right) dV$$
(3.33)

$$I_{2\mathrm{D}(1)} = \frac{\alpha}{2V_L + V_{ds}} \left(\left[\frac{\lambda_2^2}{\lambda_4} V(y) + 2V_{g1}V(y) - V(y)^2 \right] \Big|_0^{V_{ds}} - \int_0^{V_{ds}} \frac{\lambda_2}{\lambda_4} \sqrt{\lambda_2^2 + 4\lambda_4 V_{g1} - 4\lambda_4 V(y)} dV \right)$$
(3.34)

As it is known that,

$$\int \sqrt{ax+b}dx = \left(\frac{2b}{3a} + \frac{2x}{3}\right)\left(\sqrt{ax+b}\right) \tag{3.35}$$

$$I_{2\mathrm{D}(1)} = \frac{\alpha}{2V_L + V_{ds}} \left(\frac{\lambda_2^2}{\lambda_4} V_{ds} + 2V_{g1} V_{ds} - V_{ds}^2 - \left[\left(\frac{\lambda_2^2 + 4\lambda_4 V_{g1}}{-6\lambda_4} + \frac{2V(y)}{3} \right) \right] \left(\frac{\lambda_2^2}{\lambda_4} \right) \sqrt{1 + \frac{4\lambda_4 V_{g1}}{\lambda_2^2}} \right]_{0}^{V_{ds}}$$
(3.36)

Using the approximation, $[4\lambda_4 V(y) << \lambda_2^2 + 4\lambda_4 V_{g1}]$

$$I_{2\mathrm{D}(1)} = \frac{\alpha}{2V_L + V_{ds}} \left(V_{ds} \left[2V_{g1} + \frac{\lambda_2^2}{\lambda_4} \left(1 - \frac{2}{3}\sqrt{1 + \frac{4\lambda_4 V_{g1}}{\lambda_2^2}} \right) \right] - V_{ds}^2 \right)$$
(3.37)

$$I_{2\mathrm{D}(l)} = \frac{\alpha(2V_{ds}V_{g2} - V_{ds}^2)}{2V_L + V_{ds}}$$
(3.38)

where

$$V_{g2} = V_{g1} + \frac{\lambda_2^2}{2\lambda_4} \left(1 - \frac{2}{3}\sqrt{1 + \frac{4\lambda_4 V_{g1}}{\lambda_2^2}} \right)$$
(3.39)

3.2.1.2 Saturation Region Current

In saturation region of operation, the V_{ds} reaches $V_{ds(sat)}$ and at that point velocity of carrier attain its maximum velocity referred to as v_{sat} . The device in this region of operation becomes independent of V_{ds} and behaves as current source. The current density (J) for this region of operation becomes $J = Qnv_s$, which can be translated into saturation current as given below

$$I_{\rm 2D(S)} = qWv_{sat}n_s \bigg|_{V_{ds}=V_{d(sat)}}$$
(3.40)

Now using n_s from Eq. (3.24) and applying the limits

$$I_{\rm 2D(S)} = \frac{qwv_{sat}\lambda_4}{\lambda_4} \left(\frac{-\lambda_2 + \sqrt{\lambda_2 + 4\lambda_4(V_{g1} - V_{d(sat)})}}{2\lambda_4}\right)^2 \tag{3.41}$$

Using α and quadratic manipulation

$$I_{2D(S)} = \frac{\alpha}{4\lambda_4} \left[2\lambda_2^2 + 4\lambda_4 V_{g1} - 4\lambda_4 V_{d(sat)} - 2\lambda_2 \sqrt{\lambda_2^2 + 4\lambda_4 V_{g1} - 4\lambda_4 V_{d(sat)}} \right]$$
(3.42)

Using approximation, $[4\lambda_4 V_{d(sat)} << \lambda_2^2 + 4\lambda_4 V_{g1}]$, yields

$$I_{2D(S)} = \alpha \left(V_{g1} + \frac{\lambda_2^2}{2\lambda_4} \left[1 - \sqrt{1 + \frac{4\lambda_4 V_{g1}}{\lambda_2^2}} \right] - V_{d(sat)} \right)$$
(3.43)

Using Eq. (3.39), $I_{2D(S)}$ becomes

$$I_{2D(S)} = \alpha (V_{g2} - V_{d(sat)})$$
(3.44)

Comparing Eqs. (3.44) and (3.38) at $V(y) = V_{d(sat)}$

$$\alpha[V_{g2} - V_{d(sat)}] = \frac{\alpha[V_{d(sat)}V_{g2} - V_{d(sat)}^2]}{2V_L + V_{d(sat)}}$$
(3.45)

So,

$$V_{d(sat)} = \frac{2V_{g2}V_L}{(V_{q2} + 2V_L)}$$
(3.46)

which, by using Eq. (3.44), gives

$$I_{\rm 2D(S)} = \frac{\alpha V_{g2}^2}{V_{g2} - 2V_L} \tag{3.47}$$

For $V_{ds} > V_{d(sat)}$, channel length modulation occurs which can be represented as

$$I_{\rm 2D(S)} = \frac{\alpha V_{g2}^2}{V_{g2} + (2V_L - E_{sat}\Delta L_g)}$$
(3.48)

where ΔL_g is the channel length modulation [11].



FIGURE 3.4: x - y cross-sectional view of an AlGaN/GaN FinFET wherein the formation of the inversion layer is shown at the two side gates of the device.

3.2.2 Inversion Current

For a given n_s , the total magnitude of I_{ds} can be controlled by the width, W, of the device; whereas, the ratio H_{fin}/T_{fin} determines the ratio of I_{2D} and I_{inv} in I_{ds} . If H_{fin}/T_{fin} is higher, there would be a considerable ratio of I_{inv} in the total current and vice versa. It is assumed that inversion only occurs in GaN, which creates two side channels that generate a conducting path between the drain and the source. A schematic showing the formation of the two side channels is given in Fig. 3.4. In a MOS structure, there is a depletion layer which appears inside the semiconductor because of the emission of carriers from the semiconductor to the metal-oxide layer. The thickness of the depletion at equilibrium is determined by the potential resulting from the difference of work functions of the MOS structure [8]. In such a structure, if the gate bias is increased towards a direction which causes a reduction in the depletion, then at zero depletion there is a possibility that carriers of opposite polarity from the bulk could be attracted to the semiconductoroxide interface. This will generate a channel between the drain and the source as shown in Fig. 3.4. This process is referred to as channel inversion. Consider the x-y crossection of the device as shown in Fig. 3.4. Since the top gate is ignored, the 3D potential equation is reduced to

$$\frac{\partial^2 \psi}{\partial y^2} = \frac{q}{\epsilon_G} n_i \exp\left(\frac{\psi - V(x)}{V_T}\right) \tag{3.49}$$

where

$$V_T = \frac{k_B T}{q} \tag{3.50}$$

Seeing as the device is symmetric about the y-axis, the domain $0 \le y \le T_{fin}/2$ is considered and multiplied by a factor of 2, so

$$\frac{\partial^2 \psi}{\partial y^2} = 2 \frac{q}{\epsilon_G} n_i \exp\left(\frac{\psi - V(x)}{V_T}\right)$$
(3.51)

Integrating Eq. (3.51),

$$\left(\frac{d\psi}{dy}\right)^2 = \frac{2V_T q n_i}{\epsilon_G} \exp\left(\frac{\psi - V(x)}{V_T}\right) + c \tag{3.52}$$

where c is a constant. Eq. (3.52) is subject to the following boundary conditions

$$\left. \frac{\partial \psi}{\partial y} \right|_{y=0} = 0 \quad \text{and} \quad \psi(y=0) = \psi_0$$
 (3.53)

Now,

$$\frac{d\psi}{dy} = \sqrt{\frac{2V_T q n_i}{\epsilon_G} \left[\exp\left(\frac{\psi - V(x)}{V_T}\right) - \exp\left(\frac{\psi_0 - V(x)}{V_T}\right) \right]}$$
(3.54)

Integrating the above equation

$$\int dy = \frac{\int \partial \psi}{\sqrt{\frac{2V_T q n_i}{\epsilon_G} \left[\exp\left(\frac{\psi - V(x)}{V_T}\right) - \exp\left(\frac{\psi_0 - V(x)}{V_T}\right) \right]}}$$
(3.55)

Simplification yields

$$y\sqrt{2\frac{V_Tqn_i}{\epsilon_G}} = \frac{\int \partial\psi(x)}{\left(\left[\sqrt{\exp\left(\frac{\psi - V(x)}{V_T}\right)}\right]^2 - \left[\sqrt{\exp\left(\frac{\psi_0 - V(x)}{V_T}\right)}\right]^2\right)^{1/2}} \quad (3.56)$$

To simplify Eq. (3.56) further, let

$$\sqrt{\exp\left(\frac{\psi - V(x)}{V_T}\right)} = a \sec\theta \tag{3.57}$$

and

$$\sqrt{\exp\left(\frac{\psi_0 - V(x)}{V_T}\right)} = a \tag{3.58}$$

Partial differential of Eq. (3.57) gives

$$\frac{1}{2} \left[\exp\left(\frac{\psi - V(x)}{V_T}\right) \right]^{-1/2} \exp\left(\frac{\psi - V(x)}{V_T}\right) \times \frac{\partial \psi}{V_T} = a \sec\theta \tan\theta \ \partial\theta \qquad (3.59)$$

Combining Eqs. (3.57), (3.58) and (3.59) reveals

$$\partial \psi = 2V_T \tan \theta \ \partial \theta \tag{3.60}$$

So, Eq. (3.56) becomes

$$y\sqrt{2\frac{V_Tqn_i}{\epsilon_G}} = \int \frac{2V_T\tan\theta}{\sqrt{a^2\sec^2\theta - a^2}}\partial\theta$$
(3.61)

As $a^2 \sec^2 \theta - a^2 = a^2 \tan^2 \theta$, so

$$y\sqrt{2\frac{V_Tqn_i}{\epsilon_G}} = \int \frac{2V_T\tan\theta}{\tan\theta}\partial\theta = \frac{2V_T\theta}{a} + c_4$$
(3.62)

The expression for θ can be found from Eqs. (3.57) and (3.58) as,

$$\sec^{-1}\left[\sqrt{\exp\left(\frac{\psi-\psi_0}{V_T}\right)}\right] = \theta \tag{3.63}$$

In Eq. (3.62), $c_4 = 0$, using boundary conditions given by Eq. (3.53),

$$\sqrt{2\frac{qn_ia^2}{V_T\epsilon_G}} \times \frac{y}{2} = \theta \tag{3.64}$$

Using the substitution method,

$$\sqrt{2\frac{qn_i a^2}{V_T \epsilon_G}} \times \frac{y}{2} = \sec^{-1} \left[\sqrt{\exp\left(\frac{\psi - \psi_0}{V_T}\right)} \right]$$
(3.65)

By taking cosine and natural log of both sides one can write

$$q\frac{\psi - \psi_0}{2k_BT} = -\ln\left[\cos\left(y \times \sqrt{\frac{q^2 n_i}{2\epsilon_G k_BT}} \exp\frac{(\psi_0 - V(x))}{V_T}\right)\right]$$
(3.66)

Now, at the oxide interface

$$Q_i = \epsilon_{ox} \frac{V_{gs} - V_{fb} - \psi(T_{fin}/2)}{T_{ox}} = \epsilon_G \frac{\partial \psi}{\partial y} \bigg|_{y = T_{fin}/2}$$
(3.67)

here, $C_{ox} = \epsilon_{ox}/T_{ox}$.

Combining Eqs. (3.54) and (3.67),

$$C_{ox} \left[V_{gs} - V_{fb} - \psi(T_{fin}/2) \right] = \left[2\epsilon_G k_B T n_i \right]$$

$$\times \left[\exp\left(\frac{\psi(T_{fin}/2) - V(x)}{V_T}\right) - \exp\left(\frac{\psi_0 - V(x)}{V_T}\right) \right]^{\frac{1}{2}}$$
(3.68)

Let

$$\beta = \sqrt{\frac{q^2 n_i}{2\epsilon_G k_B T}} \exp\left(\frac{\psi_0 - V(x)}{V_T}\right) \times \frac{T_{fin}}{2}$$
(3.69)

So, $\psi(T_{fin}/2)$ from Eq. (3.66) become,

$$\psi(T_{fin}/2) = \psi_0 - 2V_T \ln(\cos\beta) \tag{3.70}$$

To find ψ_0 , consider Eq. (3.69)

$$\beta \times \frac{2}{T_{fin}} \times \sqrt{\frac{2\epsilon_G k_B T}{q^2 n_i}} = \sqrt{\exp\left(\frac{\psi_0 - V(x)}{V_T}\right)} \tag{3.71}$$

Taking natural log of Eq. (3.71)

$$\ln(\beta) + \ln\left[\frac{2}{T_{fin}} \times \sqrt{\frac{2\epsilon_G k_B T}{q^2 n_i}}\right] = \frac{1}{2} \left(\frac{\psi_0 - V(x)}{V_T}\right)$$
(3.72)

So,

$$\psi_0 = V(x) + 2V_T \ln\left(\beta\right) + 2V_T \ln\left[\frac{2}{T_{fin}} \times \sqrt{\frac{2\epsilon_G k_B T}{q^2 n_i}}\right]$$
(3.73)

By using Eqs. (3.70) and (3.73), Eq. (3.68) becomes

$$C_{ox}\left[V_{gs} - V_{fb} - V(x) - 2V_T \ln\left(\beta\right) - 2V_T \ln\left(\frac{2}{T_{fin}}\sqrt{\frac{2\epsilon_G k_B T}{q^2 n_i}}\right) + 2V_T \ln\left(\cos\beta\right)\right] = \sqrt{2\epsilon_G K_B T n_i} \left[\frac{\beta^2 \cos^2\beta \left(\frac{2}{T_{fin}}\sqrt{\frac{2\epsilon_G k_B T}{q^2 n_i}}\right)^2}{V_T} - \frac{\beta^2 \left(\frac{2}{T_{fin}}\sqrt{\frac{2\epsilon_G k_B T}{q^2 n_i}}\right)^2}{V_T}\right]$$
(3.74)

After simplification

$$C_{ox}\left[V_{gs} - V_{fb} - V(x) - 2V_T \ln\left(\beta\right) - 2V_T \ln\left(\frac{2}{T_{fin}}\sqrt{\frac{2\epsilon_G k_B T}{q^2 n_i}}\right) + 2V_T \ln\left(\cos\beta\right)\right] = \frac{2}{T_{fin}} \times 2\epsilon_G V_T \times \beta tan\beta$$
(3.75)

Further simplification yields,

$$\frac{(V_{gs} - V_{fb} - V(x))}{2V_T} - \ln\left(\frac{2}{T_{fin}} \times \sqrt{\frac{2\epsilon_G V_T}{qn_i}}\right) = \ln(\beta) - \ln(\cos\beta) + \frac{2\epsilon_G}{T_{fin}C_{ox}} \times \beta tan\beta$$
(3.76)

To evaluate the inversion current, consider

$$I_{inv} = \mu W Q_i \frac{dV}{dx} \tag{3.77}$$

Integrating Eq. (3.77) over the entire channel

$$\int_{0}^{L_{g}} I_{inv} dx = \mu W \int_{0}^{V_{ds}} Q_{i}(V) dV$$
(3.78)

$$I_{inv} = \frac{\mu W}{L_g} \int_{\beta_s}^{\beta_d} Q_i(B) \frac{dV}{d\beta} d\beta$$
(3.79)

Here, β_s and β_d are solutions of Eq. (3.76) for the source and drain regions, respectively. Using Eq. (3.67), one can write

$$Q_i(\beta) = \epsilon_G \frac{\partial \psi}{\partial y} \Big|_{y=T_{fin}/2}$$
(3.80)

Using Eqs. (3.67), (3.74), and (3.75) in Eq. (3.80)

$$Q_i(\beta) = 2\epsilon_G \times \frac{4}{T_{fin}} V_T \beta \tan\beta$$
(3.81)

Differentiating Eq. (3.76)

$$\frac{dV(x)}{d\beta} = \frac{2V_T}{\beta} + 2V_T tan\beta + \frac{4\epsilon_G \times V_T}{T_{fin}C_{ox}} \times \frac{d}{d\beta} (\beta tan\beta)$$
(3.82)

Combining Eqs. (3.79), (3.81) and (3.82)

$$I_{inv} = \frac{\mu W}{L_g} \times \frac{4\epsilon_G}{T_{fin}} \times \left(2V_T\right)^2 \times \left[\beta \tan\beta - \frac{\beta^2}{2} + \frac{\epsilon_G}{T_{fin}C_{ox}} \times \left(\beta \tan\beta\right)^2\right]_{\beta_d}^{\beta_s} \quad (3.83)$$

3.3 Result and Discussion

To validate the proposed model, devices having physical parameters as shown in Table 3.1 are selected. All physical dimensions of device T_1 and T_2 are identical except for H_{fin} . The thickness of the fin is of paramount importance in AlGaN/-GaN FinFETs which is meant to operate both in inversion as well as in depletion modes. If the thickness of the fin is comparable to two times the thickness of the AlGaN layer, the side gates will fully deplete the 2DEG at $V_{gs} = 0$ V and the device will only function as a normally off device (inversion mode only) [24]. Hence to allow the device to operate in both the depletion as well as in the inversion modes, it is important that T_{fin} must be appropriately scaled relative to the AlGaN layers of the device. Other device constants used for the simulation are reported in Table 3.2. Mobility, μ_0 , is evaluated using numerical optimization and the value so achieved is comparable to data reported previously [23, 42, 131].

Fig. 3.5 shows the potential distribution inside the channel of the device T_1 . The figure has been plotted using a MATLAB code developed for the purpose by employing Eq. (3.76). The dimensions of the fins are normalized, each by their respective length for simplicity. The figure shows three cross-sections of the channel. The plane to the left is the x - y cross-section at z = 0, which represents the 2DEG potential. It is plotted for two different V_{ds} and V_{gs} scenarios. Similarly, the plane to the right shows the potential variation along the x - z cross-section. This cross-section is taken at $y = 0.7L_g$ and shows the effect of drain potential on the device. The plane in the middle shows the source side potential at y = 0.3Lgand its variation with the side gates and the source terminal.

To predict I - V characteristics of AlGaN/GaN FinFETs using the proposed model, V_{th} of the device is first evaluated using Eq. (3.20). This gives the minimum V_{gs} after which the device begins to conduct. For the region $V_{th} \leq V_{gs} \leq V_{fb}$,

Parameters	$T_{1}[24]$	$T_{2}[23]$	$T_{3}[83]$
$L_g \ (\mu \mathrm{m})$	5	5	0.1
$L_{gs} \ (\mu \mathrm{m})$	2	2	2.5
$L_{gd} \ (\mu \mathrm{m})$	17	17	2.5
$T_{fin} (\mathrm{nm})$	150	150	250×8
$H_{fin} (\mathrm{nm})$	250	120	200
T_{ox} (nm)	20	20	10
$d (\rm nm)$	15	15	25
GaN (nm)	60	60	175
$W~(\mu { m m})$	15	15	2
V_{fb} (V)	-0.4	-1.4	-1.5
V_{th} (V)	-5	-2	-4
$\mu_0 \; (m^2 V^{-1} s^{-1})$	0.17	0.17	0.14
Oxide	Al_2O_3	$\mathrm{Al}_2\mathrm{O}_3$	HfO_{2}
ϵ_{ox}	9.1ϵ	9.1ϵ	25ϵ

TABLE 3.1: Physical parameters of different devices used in this study.

TABLE 3.2: Physical constants used in this study.

Parameters	Value	
$n_i \ (\times \ 10^{21} \ \mathrm{m}^{-3})$	1	
$k_B ({\rm J.K^{-1}})$	1.38	
T (K)	300	
$q \ (\times \ 10^{-19} \ {\rm C})$	1.6	
$\epsilon~(\times~10^{-12}~{\rm Fm^{-1}})$	8.85	
ϵ_A	8.8ϵ	
ϵ_G	8.9ϵ	



FIGURE 3.5: Potential profile at various points of an AlGaN/GaN FinFET (Device T_1).

the device operates in the depletion mode, where 2DEG is the main contributor towards the current flowing through the channel, and Eqs. (3.38), (3.46) and (3.47) describe the relationship between the 2DEG current and the applied voltages. Eq. (3.38) describes the linear region current; whereas, Eq. (3.47) represents the saturation current. To evaluate the switching point between the two, Eq. (3.46) is used which provides the value of V_{ds} , after which the channel of the device saturates for a given V_{qs} .

At $V_{gs} = V_{fb}$, 2DEG is completely un-depleted and this defines the upper limit of conventional HEMT devices' drain current. This is because a further increase in V_{gs} will exponentially increase gate leakage and the gate will cease to have any control on the device. However, due to the presence of the oxide layer in AlGaN/GaN FinFETs, gate leakage is minimized above $V_{gs} > V_{fb}$. Consequently, inversion occurs in the GaN layer and the device begins to operate as a regular GaN MOSFET.

In the proposed model, inversion current is given by Eq. (3.83), wherein β is a function of the applied potentials (V_{gs}, V_{ds}) of the device. In this manner, using both the depletion and inversion parts of the model simultaneously, the I - V

characteristics of an AlGaN/GaN FinFET can be assessed for a wider range of applied V_{gs} .

The presented model is applied on the devices given in Table 3.1 while the model parameters are optimized using a cost function defined by the following equation [2].

$$RMSE = \sqrt{\sum_{Q=1}^{N} \left\{ \sum_{P=1}^{M} \left(I_{ds(exp)}^{P,Q} - I_{ds(mod)}^{P,Q} \right)^2 / \sum_{P=1}^{M} I_{ds(exp)}^{P,Q} \right\}}$$

$$\leq E_{r(min)}$$

$$(3.84)$$

Here, $I_{ds(exp)}$ is the observed while $I_{ds(mod)}$ is the modeled drain currents. $E_{r(min)}$ is the minimum error value, and the variables P, Q represent variation in the drain and gate voltages, respectively.

Fig. 3.6 (a), (b) and (c) are plotted to show the results of the proposed model in comparison with experimental data. It is pertinent to mention that the devices presented in Table 3.1 are evaluated for both positive and negative values of V_{gs} in order to ensure their operation in both depletion and inversion modes. Examining Fig. 3.6, it is apparent that the proposed model demonstrates reasonable accuracy in predicting I - V characteristics of the chosen devices across the applied V_{gs} and V_{ds} .

The experimental and modeled output conductance of the chosen devices is shown in Fig. 3.7. The modeled characteristics follow the experimental data reasonably well which, once again, speaks that the proposed model couples the depletion and inversion operation of the selected devices with a good degree of accuracy.

To further investigate the effect of inversion on AlGaN/GaN FinFETs, Fig. 3.8 is plotted. Fig. 3.8(a) and (b) exhibit multiple peaks in the transconductance characteristics which is not a usual response of FETs. It is assumed that 2DEG and inversion currents are coming into action independently; which could possibly cause generation of multiple peaks in the device transfer characteristics. A plausible explanation could be that for long channel devices $(T_1\&T_2)$, the first



FIGURE 3.6: Modeled and experimental drain current of AlGaN/GaN FinFETs of Table 3.1: (a) T_1 (b) T_2 (c) T_3 . Operation of the devices at high V_{gs} indicates the formation of inversion layer and its subsequent contribution in the drain current.



FIGURE 3.7: Modeled and experimental output conductance of AlGaN/GaN FinFETs described in Table 3.1: (a) T_1 (b) T_2 (c) T_3 .



FIGURE 3.8: Transconductance and drain current versus gate voltage of Al-GaN/GaN FinFETs: device (a) T_1 (b) T_2 (c) T_3 .

peak is associated with 2DEG current wherein the the transconductance, g_m peak decreases under relatively high negative gate biases because of the usual nonlinear behavior of the depletion under the applied gate bias. On the other hand, at a relatively higher positive gate bias, I_{inv} also starts contributing to I_{ds} and g_m begins to rise once again, causing another peak in the transfer characteristics of the device. In the case of Fig. 3.8(c), the device has $L_g = 100$ nm, which is significantly lower than the other two devices that did not exhibit multiple peaks; possibly due to the dominant fringing gate field which is routinely observed in short channel FETs.

The plots of Fig. 3.8 demonstrate that the proposed model follows the rise and fall in g_m with relative ease; further confirming its position as a comprehensive model to predict DC characteristics of AlGaN/GaN FinFETs for both depletion and inversion mode of operations. It may be noted that the proposed model is a steadystate model and hence does not predict AC response of the device. Furthermore, it deals with the device under its usual operation and does not predict the device response in break down or sub-threshold regimes. The model could further be extended to evaluate the response of the device at elevated temperatures.

3.4 Summary

In this chapter, an I-V model for trigate AlGaN/GaN FinFET is presented which is capable of predicting the DC characteristics of the device in both depletion and inversion mode of operations. The model is validated using experimental data wherein it showed reasonable accuracy. The model is divided into two parts: the depletion and inversion parts. The depletion part of the model assesses current using the modified definition of 2DEG carrier concentration, n_s , by accommodating the effects of the trigate nature of the device. The inversion part of the model uses Poisson distribution to evaluate the channel potential which is then translated into channel current. The total drain to source current, I_{ds} , is the addition of both the currents predicted by the depletion and inversion parts of the model. It is shown that 2DEG current is the main contributor to I_{ds} for the gate bias $V_{th} \leq V_{gs} \leq V_{fb}$, whilst a further increase in V_{gs} potential beyond V_{fb} will generate an additional component of the current due to the inversion of carriers generated by the two side gates of AlGaN/GaN FinFETs.

Chapter 4

A Schrödinger-Poisson Model for Output Characteristics of Trigate Ballistic Si FinFETs

4.1 Introduction

To cope with the ever growing demands of the electronic industry, FET shrinkage was successfully carried out up to 100 nm but further down scaling was halted due to rise of adverse effects, such as high leakage current, high power dissipation, increased sub threshold swing and hot carriers injection [16, 132]. Therefore, in recent era, nanoscale technology has become a topic of great interest for researchers working to improve the performance of FETs with reduced dimensions [133]. One of the most promising candidates for reducing short channel effects are fin field effect transistors (FinFETs) [59]. Due to their very design, FinFETs exhibit more control upon the channel characteristics and thus open doors for further device scaling [62]. FinFETs offer promising characteristics both at low power to allow further device concentration per unit area, and at a high frequency [134–136] to accommodate the ever expanding communication needs. Apart from the dynamic I - V characteristics models [137], there are several DC models to represent charge transport in electronic devices which include Boltazman transport, hydrodynamic transport and drift-diffusion transport models [138]. However, short channel devices having L_g below 10 nm, there is significant tunneling between source and drain and the dual nature of particles becomes prominent [139]. The particles in such devices begin to behave as waves governed by the Schrödinger wave equation [140]. In recent years, several quantum models have been reported for such devices in literature which aim to investigate the effects of quantum mechanical transport on the device characteristics [19, 88, 91]. These models apply techniques such as non-equilibrium green function (NEGF) to assess quantum mechanical effects in nano devices [86, 89, 141–143].

Venugopala *et al.* developed a computational technique [142] to assess drain current of nanoscale FETs using a two dimensional quantum-mechanical ballistic model wherein the main consideration was carriers scattering while traveling from source to drain electrodes. With this technique, authors demonstrated that carriers of devices with ballistic transport experience less scattering thus offering current density of relatively higher magnitude. On the other hand, a 3D ballistic transport model, by considering electron-phonon interaction, was developed by Jin *et al.* using an NEGF [144]. They formulated their model using a 15 nm long nanowire FET and presented simulation results without any experimental validation. Their work demonstrated that electron-phonon scattering did have a tangible impact on the device characteristics. The simulated data exhibited a significant reduction in drain current (from 16 μ A to 10 μ A) when electron-phonon scattering was taken into consideration.

To solve NEGF, a real space method is considered an appropriate approach; however, it is computational extensive and for nanoscale devices, it is reported that coupled mode approach is efficient and accurate enough to predict quantum transport in nano FETs [84, 145]. In most of the NEGF formalism, the nano FETs characteristics were simulated to demonstrate the validity of the developed techniques; however, the attained characteristics were not compared with experimental data to ensure the applicability of these models [84, 144]. In this part of the research, we have developed a Schrödinger-Poisson model capable of predicting the output characteristics of Si trigate FinFETs by involving quantum-mechanical ballistic transport. In the proposed model, a 3D NEGF along with Poisson equation is solved, which predicts the difference of Fermi energies of the drain and source contacts and hence the flow of electrons. The model is compared with a 2D quantum model [146] and also validated against experimental data. The model shows great promise in predicting the device output and transfer characteristics with a high degree of accuracy. According to our conservative



FIGURE 4.1: a) Channel of a trigate FinFET, b) x - y crossection of a trigate FinFET.

estimate, no such comprehensive data is reported in literature where experimental verification of a NEGF for nanoscale FinFETs is performed.

4.2 Model Development

The proposed model is developed using a representative geometry of a FinFET as illustrated in Fig. 4.1, This structure presents a *n*-type depletion mode FinFETs. A 3D view of the channel is shown in Fig. 4.1 (a) while Fig. 4.1 (b) shows the 2D x - y crossection. Metal gate is placed on the channel using oxide layer having thickness T_{ox} . The channel thickness as defined by Si is T_{fin} , whereas its height is H_{fin} as shown in Fig. 4.1 (a). Under biased conditions, the source is connected to the ground and the drain to $V_d > 0$, which generates a potential in the channel represented by $V_{ch}(y)$. Potential at the middle of the channel is represented by $\psi_0(y)$ which is independent of x. It is assumed that the current flow is ballistic i.e., $Lg \ll l$ where l is the mean free path, and only the conduction band electrons contribute towards the current flow. Also, there is no inter-valley mixing of carriers involved in the conduction. It is also assumed that there is no scattering at drain and source contacts and the variation in potential is slow, such that the system attains steady state at each bias point. Poisson and Schrödinger equations are solved to predict the output characteristics of trigate FinFETs and the details of which are explained below.

4.2.1 Schrödinger Equation

The fundamental equation that governs the model is the time-dependent Schrödinger equation which is given as [140]

$$j\hbar\frac{\partial\psi}{\partial t} = \left(-\frac{\hbar^2}{2m}\Delta + \mathbf{V}\right)\psi \tag{4.1}$$

In Eq. (4.1), Δ is the Laplacian operator and its dimensionality depends upon the application of the equation and **V** is the potential energy. Assuming that in FinFET channel, electrons behave as they would in vacuum, but with a different mass referred to as effective mass, m^* . For such a system, Eq. (4.1) can be written as

$$j\hbar\frac{\partial\psi}{\partial t} = \left(-\frac{\hbar^2}{2m^*}\Delta + \mathbf{U}\right)\psi \tag{4.2}$$

Considering that the system in which Eq. (4.2) shall be applied is time independent, and for such a system the equation reduces to

$$\left(-\frac{\hbar^2}{2m^*}\Delta + \mathbf{U}\right)\boldsymbol{\psi} = \mathbf{E}\boldsymbol{\psi}$$
(4.3)

which can be written as

$$\mathbf{H}\boldsymbol{\psi} = \mathbf{E}\boldsymbol{\psi} \tag{4.4}$$

where,

$$\mathbf{H} = \left(-\frac{\hbar^2}{2m^*}\Delta + \mathbf{U}\right) \tag{4.5}$$

The analytical solution of Eq. (4.4) is impractical as the solution changes due to its dependence upon channel potential energy, U. The magnitude of U can be determined by the charge distribution inside a FinFET channel. Such a charge distribution can be statistically assessed by Fermi dirac distribution function. Thus, this distribution shall be dependent upon the external gate and drain applied potential, which shall be determined in the later part of the developed model. One of the possible ways of addressing this issue is to solve the equation numerically. Apply a finite difference method [147], a 3D representation of Eq. (4.4) can be

$$\mathbf{E}\boldsymbol{\psi}_{xyz} = \mathbf{H}\boldsymbol{\psi}_{xyz} = \left[-\frac{\hbar^2}{2m^*}\left(\frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2} + \frac{\partial^2}{\partial z^2}\right) + \mathbf{U}\right]\boldsymbol{\psi}_{xyz}$$
(4.6)

Considering solution

$$\boldsymbol{\psi}_{xyz} = \psi_o e^{jk_x xa + jk_y ya + jk_z za} \tag{4.7}$$

By assuming that there is no variation in the channel material and the wave function does not penetrate into the oxide layer, one can, by considering nearest



FIGURE 4.2: Allowed energy levels in the conduction band of a FinFET.

neighbors, substitute Eq. (4.7) in Eq. (4.6) to have

$$(\mathbf{E} - \mathbf{U})\psi_{xyz} = \epsilon \psi_{xyz} + t_o(-\psi_{x-1yz} - \psi_{x+1yz} - \psi_{xy-1z} - \psi_{xy+1z} - \psi_{xyz+1} - \psi_{xyz-1})$$
(4.8)

where, $\epsilon = E_c + 6t_o$ and

$$t_o = \frac{\hbar^2}{2m^*a^2} \tag{4.9}$$

Combining Eq. (4.7) and Eq. (4.8), we get

$$\mathbf{E} = \epsilon - 2t_o \cos(k_x a) - 2t_o \cos(k_y a) - 2t_o \cos(k_z a) + \mathbf{U}$$
(4.10)

Eq. (4.10) governs the energies of the conduction band. Only those energies are allowed which satisfy the equation. The conduction band of a biased FinFET is modeled by a shifted flipped cosine wave as shown in Fig. 4.2. Considering the channel with equidistant lattice, the Hamilton operator (**H**) can be discretized using finite difference method which allows Eq. (4.6) in matrix form as

$$\mathbf{E}\boldsymbol{\psi} = \begin{bmatrix} \mathbf{A}_{1} & \mathbf{A}_{2} & \mathbf{A}_{2} & 0 & 0 & \dots & 0 \\ \mathbf{A}_{2} & \mathbf{A}_{1} & \mathbf{A}_{2} & \mathbf{A}_{2} & 0 & \dots & 0 \\ \vdots & \ddots & \ddots & \ddots & \ddots & \ddots & 0 \\ 0 & \ddots & 0 & \mathbf{A}_{2} & \mathbf{A}_{2} & \mathbf{A}_{1} & \mathbf{A}_{2} \\ 0 & 0 & \ddots & 0 & \mathbf{A}_{2} & \mathbf{A}_{2} & \mathbf{A}_{1} \end{bmatrix} \boldsymbol{\psi}$$
(4.11)

where A_1 and A_2 are given as,

$$\mathbf{A_1} = \begin{bmatrix} \epsilon & -t_o & 0 & \dots & 0 \\ -t_o & \epsilon & -t_o & \dots & 0 \\ \vdots & \ddots & \ddots & \ddots & 0 \\ 0 & 0 & -t_o & \epsilon & -t_o \\ 0 & 0 & 0 & -t_o & \epsilon \end{bmatrix}$$
(4.12)

$$\mathbf{A_2} = \begin{bmatrix} -t_o & 0 & 0 & \dots & 0 \\ 0 & -t_o & 0 & \dots & 0 \\ \vdots & \ddots & \ddots & \ddots & 0 \\ 0 & 0 & 0 & -t_o & 0 \\ 0 & 0 & 0 & 0 & -t_o \end{bmatrix}$$
(4.13)

The Hamiltonian matrix defined in Eq. 4.11 consists of diagonal and off-diagonal terms. In the grid, each diagonal element corresponds to a specific point and coupling between the points is represented by off-diagonal elements. Combined definition of Hamiltonian with ohmic contacts is an eigenvalue problem in quantum mechanics. Solving such a problem, the device eigenstates and eigenvalues are determined when isolated from the outside world, i.e., when the system is closed. To model electron transport, an effective mass Hamiltonian is used. The effective mass approximation is valid for the cases where potential relative to the atomic scale is changing slowly and the proposed model was developed under same constraints.

4.2.2 Current Equation

To find the current flow through the channel of a FinFET, consider Eq. (4.4) once again which describes the channel under the conditions when there is no in/out



FIGURE 4.3: Inflow and outflow of charges in a FinFET channel.

flow of electrons. Thus, to describe the current flow, the same is modified as

$$\mathbf{E}\boldsymbol{\psi} = \mathbf{H}\boldsymbol{\psi} + \boldsymbol{\Sigma}\boldsymbol{\psi} + \mathbf{s} \tag{4.14}$$

where, $\Sigma \psi$ describes the outflow of electrons from the contacts. The $\Sigma = \Sigma_1 + \Sigma_2$, represents source and drain energies, respectively, which are responsible for the outflow while **s** represents inflow as shown in Fig. 4.3. Σ can be found by finding eigenvalues and eigenvectors of \mathbf{A}_1 . Σ can be written as

where, $\boldsymbol{\beta}$ is the eigenvector associated with $\mathbf{A_1}$ and $\boldsymbol{\beta}^+$ is its conjugate transpose;

$$\mathbf{A_3} = \begin{bmatrix} t_o e^{jk_1a} & 0 & 0 & \dots & 0 \\ 0 & t_o e^{jk_2a} & 0 & \dots & 0 \\ \vdots & \ddots & \ddots & \ddots & 0 \\ 0 & 0 & 0 & t_o e^{jk_{(n-1)}a} & 0 \\ 0 & 0 & 0 & 0 & t_o e^{jk_na} \end{bmatrix}$$
(4.16)

where k_i can be found by solving the following equation;

$$\mathbf{E} = \lambda_i - 2t_o \cos(k_i a) \tag{4.17}$$

here, λ_i are the eignvalues of A₁. Eq. (4.14) can be written as

$$\boldsymbol{\psi} = [\mathbf{E}\mathbf{I} - \mathbf{H} - \boldsymbol{\Sigma}]^{-1}\mathbf{s} \tag{4.18}$$

As \mathbf{E} is a column vector, in order to correct the dimensionality of Eq. (4.18), it is multiplied by \mathbf{I} , where \mathbf{I} is an identity matrix. Let,

$$\mathbf{G}^{\mathbf{R}} = [\mathbf{E}\mathbf{I} - \mathbf{H} - \boldsymbol{\Sigma}]^{-1} \tag{4.19}$$

So,

$$\boldsymbol{\psi} = \mathbf{G}^{\mathbf{R}}\mathbf{s} \tag{4.20}$$

Now, taking the conjugate transpose (+) of Eq. (4.18) gives,

$$\psi^+ = \mathbf{s}^+ \mathbf{G}^\mathbf{A} \tag{4.21}$$

where,

$$\mathbf{G}^{\mathbf{A}} = \left[\left[\mathbf{E}\mathbf{I} - \mathbf{H} - \boldsymbol{\Sigma} \right]^{-1} \right]^{+} = \left[\mathbf{G}^{\mathbf{R}} \right]^{+}$$
(4.22)

The electron density matrix can be defined as

$$\mathbf{G}^{\mathbf{n}} = 2\pi \boldsymbol{\psi} \boldsymbol{\psi}^{+} \tag{4.23}$$

By combining Eqs. (4.20), (4.21) and (4.23)

$$\mathbf{G}^{\mathbf{n}} = 2\pi \mathbf{G}^{\mathbf{R}} \mathbf{s} \mathbf{s}^{+} \mathbf{G}^{\mathbf{A}} \tag{4.24}$$

In Eq. (4.24), $\mathbf{G}^{\mathbf{n}}$, $\mathbf{G}^{\mathbf{R}}$ and $\mathbf{G}^{\mathbf{A}}$ are functions of energy and have units eV^{-1} and \mathbf{ss}^+ defines the inflow of electrons, which can be written as

$$\Sigma^{\rm in} = 2\pi s s^+ = \Gamma \tag{4.25}$$

here, Σ^{in} represents the outflow of electrons. Eq. (4.25) is valid because for steady flow, current entering must be equal to current leaving. If there is one source of incoming electrons, and two outflow sources as shown in Fig. 4.3, one can write

$$\Sigma^{in} = \Gamma_1 \mathbf{f_1}(\mathbf{E}) + \Gamma_2 \mathbf{f_2}(\mathbf{E}) \tag{4.26}$$

here,

$$\Gamma_i = j[\Sigma_i - \Sigma_i^+] \tag{4.27}$$

 Γ_i is the anti-Hermitian part of Σ_i and it describes the ease of communication between the channel and the contacts. $\mathbf{f_{1,2}}$ is expressed as [148],

$$\mathbf{f_{1,2}}(\mathbf{E}) = \frac{1}{1 + \exp\left[\frac{\mathbf{E} - (P_{1,2} + U_f)}{k_B T}\right]}$$
(4.28)

where

$$U_f = \alpha Q(V_g + V_d) \tag{4.29}$$

In Eq. (4.29), α is a fitting factor which incorporates contact resistance, Q is the total charge and is determined by solving a 3D Poisson equation which is discussed in the subsection to follow. To find channel current consider,

$$j\hbar \frac{d}{dt}\boldsymbol{\psi} = [\mathbf{H} + \boldsymbol{\Sigma}]\boldsymbol{\psi} + \mathbf{s}$$
(4.30)

thus,

$$j\hbar\frac{d}{dt}\boldsymbol{\psi}\boldsymbol{\psi}^{+} = \left[(\mathbf{H} + \boldsymbol{\Sigma})\boldsymbol{\psi}\boldsymbol{\psi}^{+} - \boldsymbol{\psi}\boldsymbol{\psi}^{+}(\mathbf{H} + \boldsymbol{\Sigma}^{+})\right] + \left[\mathbf{ss}^{+}\mathbf{G}^{\mathbf{A}} - \mathbf{G}^{\mathbf{R}}\mathbf{ss}^{+}\right]$$
(4.31)

Eq. (4.31) is obtained using the chain rule for derivatives on Eq (4.30) and making use of Eqs. (4.20) and (4.21). The number of electrons can be obtained by $\text{Trace}[\psi\psi^+]$ which represents the sum of all elements on the main diagonal. Substituting Eqs. (4.23) and (4.25) in Eq. (4.31), one gets

$$\frac{d}{dt}\psi\psi^{+} = \frac{[\mathbf{H}\mathbf{G}^{\mathbf{n}} - \mathbf{G}^{\mathbf{n}}\mathbf{H}] + [\boldsymbol{\Sigma}\mathbf{G}^{\mathbf{n}} - \mathbf{G}^{\mathbf{n}}\boldsymbol{\Sigma}^{+}] + [\boldsymbol{\Sigma}^{\mathbf{in}}\mathbf{G}^{\mathbf{A}} - \mathbf{G}^{\mathbf{R}}\boldsymbol{\Sigma}^{\mathbf{in}}]}{j2\pi\hbar}$$
(4.32)
As Trace[AB] = Trace[BA] and $h = 2\pi\hbar$, therefore

$$\operatorname{Trace}\left(\frac{d}{dt}\psi\psi^{+}\right) = \frac{-i}{h}\operatorname{Trace}\left(\left[\Sigma G^{n} - G^{n}\Sigma^{+}\right] + \left[\Sigma^{in}G^{A} - G^{R}\Sigma^{in}\right]\right)$$
(4.33)

By using Eq. (4.27)

Trace
$$\left(\frac{d}{dt}\psi\psi^{+}\right) = \frac{-i}{h}$$
Trace $\left(-i\Gamma G^{n} + \Sigma^{in}[G^{A} - G^{R}]\right)$ (4.34)

To find $G^A - G^R$ using Eqs. (4.19), (4.22) and (4.27),

$$G^A - G^R = iG^R \Gamma G^A \tag{4.35}$$

Now, Eq. (4.34) becomes

Trace
$$\left(\frac{d}{dt}\psi\psi^{+}\right) = \frac{1}{h}$$
Trace $\left(\Sigma^{in}G^{R}\Gamma G^{A} - \Gamma G^{n}\right)$ (4.36)

To find electronic current per unit energy through a contact, multiply Eq. (4.32) with electronic charge, q

$$I_m = \operatorname{Trace}\left(q\frac{d}{dt}\boldsymbol{\psi}\boldsymbol{\psi}^+\right) = \frac{q}{h}\operatorname{Trace}\left(\boldsymbol{\Sigma}_{\mathbf{m}}^{\mathbf{in}}\mathbf{G}^{\mathbf{R}}\boldsymbol{\Gamma}\mathbf{G}^{\mathbf{A}} - \boldsymbol{\Gamma}_{\mathbf{m}}\mathbf{G}^{\mathbf{n}}\right)$$
(4.37)

Current through one of the contacts of the device is given as,

$$I_1 = \frac{q}{h} \operatorname{Trace} \left[\Gamma_1 \left(\mathbf{f_1} \mathbf{G}^{\mathbf{R}} \Gamma \mathbf{G}^{\mathbf{A}} - \mathbf{G}^{\mathbf{n}} \right) \right]$$
(4.38)

where, $\Sigma_1^{in} = \Gamma_1 \mathbf{f_1}$.

Since there are two contacts, drain and source, so $\Gamma = \Gamma_1 + \Gamma_2$. Also using Eq. (4.24) we get

$$I_{1} = \frac{q}{h} \operatorname{Trace} \left[\Gamma_{1} \left(f_{1} G^{R} \{ \Gamma_{1} + \Gamma_{2} \} G^{A} - G^{R} \{ \Gamma_{1} f_{1} + \Gamma_{2} f_{2} \} G^{A} \right) \right]$$
(4.39)

So,

$$I_1 = \frac{q}{h} \operatorname{Trace} \left[\Gamma_1 \mathbf{G}^{\mathbf{R}} \Gamma_2 \mathbf{G}^{\mathbf{A}} \right] \left(\mathbf{f_1} - \mathbf{f_2} \right)$$
(4.40)

Now, the total current is

$$I_{ds} = \int_{-\infty}^{+\infty} \frac{q}{h} \operatorname{Trace} \left[\Gamma_1 \mathbf{G}^{\mathbf{R}} \Gamma_2 \mathbf{G}^{\mathbf{A}} \right] \left(\mathbf{f_1} - \mathbf{f_2} \right) dE$$
(4.41)

The above equation represents $I_{ds}(V_{ds}, V_{gs})$ characteristics of a FinFET wherein, quantum-mechanical ballistic transport of the carriers is considered. It is pertinent to mention that current saturation in short channel FETs is usually attributed to the velocity saturation of carriers. However, no term representing the velocity is involved in the above expression; thus, the only possibility which could be associated with current saturation in quantum-mechanical transport would be the saturation of carrier supply from the source reservoir.

4.2.3 Poisson Equation

To find Q appearing in Eq. (4.29), a 3D Poisson equation is applied on the channel shown in Fig. 4.1b. The equation includes both inversion and bulk carriers and can be written as

$$\frac{\partial^2 \psi_p}{\partial x^2} + \frac{\partial^2 \psi_p}{\partial y^2} + \frac{\partial^2 \psi_p}{\partial z^2} = \frac{qn_i}{\epsilon_{si}} \exp\left(\frac{\psi_p + \phi_B - V_{ch}}{V_T}\right) + \frac{qN_d}{\epsilon_{si}}$$
(4.42)

where,

$$\phi_B = V_T \ln\left(\frac{N_d}{n_i}\right) \tag{4.43}$$

and

$$V_T = \frac{k_B T}{q} \tag{4.44}$$

where V_{ch} varies as a function of L_g and towards the source side $V_{ch}(0) = V_s$ and $V_{ch}(L_g) = V_d$. Dividing the solution into parts i.e., ψ_{p1} and ψ_{p2} , such that $\psi_p = \psi_{p1} + \psi_{p2}$. Considering,

$$\frac{\partial^2 \psi_{p1}}{\partial x^2} = \frac{qn_i}{\epsilon_{si}} \exp\left(\frac{\psi_{p1} + \phi_B - V_{ch}}{V_T}\right) + \frac{qN_d}{\epsilon_{si}}$$
(4.45)

Integrating Eq. (4.45) twice and applying boundary conditions

$$\left. \frac{\partial^2 \psi_{p1}}{\partial x^2} \right|_{x=0} = 0 \qquad \text{and} \qquad \psi_{p1}(x=0,y) = \psi_0(y) \tag{4.46}$$

Poisson equation contain bulk and inversion carriers such that the solution ψ_{p1}

$$\psi_{p1} = \psi_1 + \psi_2 \tag{4.47}$$

Here, ψ_2 represents bulk carriers and ψ_1 is the potential due to inversion of carriers, which can be written as

$$\frac{\partial^2 \psi_1}{\partial x^2} = \frac{qn_i}{\epsilon_{si}} \exp\left(\frac{\psi_1 + \phi_B - V_{ch}}{V_T}\right)$$
(4.48)

Integrating Eq. (4.48),

$$\left(\frac{d\psi_1}{dx}\right)^2 = \frac{qn_i}{\epsilon_{si}} \exp\left(\frac{\psi_1 + \phi_B - V_{ch}}{V_T}\right) + c \tag{4.49}$$

Using boundary condition, the above equation becomes

$$\frac{d\psi_1}{dx} = \sqrt{\frac{V_T q n_i}{\epsilon_{si}} \left[\exp\left(\frac{\psi_1 + \phi_B - V_{ch}}{V_T}\right) - \exp\left(\frac{\psi_0 + \phi_B - V_{ch}}{V_T}\right) \right]}$$
(4.50)

$$\int \frac{\partial \psi_1}{\sqrt{\frac{V_T q n_i}{\epsilon_{si}} \left[\exp\left(\frac{\psi_1 + \phi_B - V_{ch}}{V_T}\right) - \exp\left(\frac{\psi_0 + \phi_B - V_{ch}}{V_T}\right) \right]}} = \int \partial x \quad (4.51)$$

Further simplification yields

$$\int \frac{\partial \psi_1}{\left(\left[\sqrt{\exp\left(\frac{\psi_1 + \phi_B - V_{ch}}{V_T}\right)}\right]^2 - \left[\sqrt{\exp\left(\frac{\psi_0 + \phi_B - V_{ch}}{V_T}\right)}\right]^2\right)^{1/2}} \quad (4.52)$$
$$= x\sqrt{\frac{V_T q n_i}{\epsilon_{si}}}$$

To simplify the above equation further, let

$$\sqrt{\exp\left(\frac{\psi_1 + \phi_B - V_{ch}}{V_T}\right)} = b \sec\theta \tag{4.53}$$

and

$$\sqrt{\exp\left(\frac{\psi_0 + \phi_B - V_{ch}}{V_T}\right)} = b \tag{4.54}$$

Partial differential of Eq. (4.53),

$$\frac{1}{2} \left[\exp\left(\frac{\psi_1 + \phi_B - V_{ch}}{V_T}\right) \right]^{-1/2} \exp\left(\frac{\psi_1 + \phi_B - V_{ch}}{V_T}\right) \times \frac{\partial \psi_1}{V_T}$$

$$= b \sec\theta \tan\theta \ \partial\theta$$
(4.55)

Combining Eqs. (4.53), (4.54) and (4.55) reveals

$$\partial \psi_1 = 2V_T \tan \theta \ \partial \theta \tag{4.56}$$

Substituting Eqs. (4.53), (4.54) and (4.56) in Eq. (4.52)

$$x\sqrt{\frac{V_Tqn_i}{\epsilon_{si}}} = \int \frac{2V_T\tan\theta}{\sqrt{b^2\sec^2\theta - b^2}}\partial\theta$$
(4.57)

as $b^2 \sec^2 \theta - b^2 = b^2 \tan^2 \theta$, so

$$x\sqrt{\frac{V_Tqn_i}{\epsilon_{si}}} = \int \frac{2V_T\tan\theta}{b\tan\theta}\partial\theta = \frac{2V_T\theta}{b} + c_2 \tag{4.58}$$

Eqs. (4.53) and (4.54) can be used to generate the expression for θ

$$\sec^{-1}\left[\sqrt{\exp\left(\frac{\psi_1 - \psi_0}{V_T}\right)}\right] = \theta \tag{4.59}$$

consider $c_2 = 0$ and by rearranging Eq. (4.58), one gets

$$\sqrt{\frac{qn_ib^2}{V_T\epsilon_{si}}} \times \frac{x}{2} = \theta \tag{4.60}$$

Inserting the value of θ in Eq. (4.59)

$$\sqrt{\frac{qn_ib^2}{V_T\epsilon_{si}}} \times \frac{x}{2} = \sec^{-1}\left[\sqrt{\exp\left(\frac{\psi_1 - \psi_0}{V_T}\right)}\right]$$
(4.61)

Using $\cos(\sec^{-1}\theta) = 1/\theta$, above expression yields

$$\cos\left[\sqrt{\frac{qn_ib^2}{V_T\epsilon_{si}}} \times \frac{x}{2}\right]^2 = \exp\left(-\frac{\psi_1 - \psi_0}{V_T}\right)$$
(4.62)

Taking natural log of both the sides of Eq. (4.62) and using Eq. (4.44) reveals

$$\psi_1 = \psi_0 - \frac{2k_BT}{q} \ln\left[\cos\left(\sqrt{\frac{q^2 n_i^2}{k_B T N_d \epsilon_{si}}} \exp\left(\psi_0 - V_{ch}\right) \times \frac{x}{2}\right)^2\right]$$
(4.63)

The potential caused by the inversion carriers is represented by Eq. (4.63). To assess the bulk carries potential contribution, consider

$$\frac{\partial^2 \psi_2}{\partial x^2} = \frac{qN_d}{\epsilon_{si}} \tag{4.64}$$

Integrating Eq. (4.64) twice and using boundary conditions $\psi_2 = 0$ and $\partial \psi_2 / \partial x |_{x=0} = 0$, one gets

$$\psi_2 = \frac{qN_d x^2}{2\epsilon_{si}} \tag{4.65}$$

The solution for ψ_{p1} as given in Eq. (4.47) becomes

$$\psi_{p1} = \psi_0(y) - \frac{2k_B T}{q} \ln \left[\cos \left(\sqrt{\frac{q^2 n_i^2}{k_B T N_d \epsilon_{si}}} \exp \left(\psi_0(y) - V_{ch}(y) \right)} \times \frac{x}{2} \right)^2 \right] + \frac{q N_d x^2}{2\epsilon_{si}}$$
(4.66)

Also,

$$\frac{\partial^2 \psi_{p2}}{\partial y^2} + \frac{\partial^2 \psi_{p2}}{\partial z^2} = 0 \tag{4.67}$$

With boundary conditions $\psi_{p2}(y = 0, z) = V_s$, $\psi_{p2}(y = L_g, z) = V_d$, $\psi_{p2}(y, z = 0) = \psi_{p1}$, $\psi_{p2}(y, z = H_{fin}) = \phi_B$, and with $\eta = \pi/2L_g$, $\alpha_1 = \pi/2H_{fin}$, the solution

becomes

$$\frac{\partial^2 Y(y)Z(z)}{\partial y^2} = -\frac{\partial^2 Y(y)Z(z)}{\partial z^2} = -\alpha^2$$
(4.68)

where, $\eta = \alpha^2 = \pi/(2L_g)$.

$$\frac{\partial^2 Y(y)}{\partial y^2} + \alpha^2 Y(y) = 0 \tag{4.69}$$

$$\frac{\partial^2 Z(z)}{\partial z^2} - \alpha^2 Z(z) = 0 \tag{4.70}$$

Solution of Eqs. (4.69) and (4.70) can be expressed as

$$Y(x, y, z) = c_3 \cos(\eta y) + c_4 \sin(\eta y)$$
(4.71)

$$Z(x, y, z) = c_5 \exp(-\eta z) + c_6 \exp(\eta z)$$
(4.72)

Using boundary conditions in Eq. (4.71) results,

$$Y(x,0,\dot{z}) = c_3 = V_s \tag{4.73}$$

and

$$Y(x, L_g, \acute{z}) = c_4 = V_d \tag{4.74}$$

Combining Eqs. (4.71), (4.73) and (4.74)

$$Y(x, y, z') = V_s \cos(\eta y) + V_d \sin(\eta y) \tag{4.75}$$

Assuming $\dot{z} = (L_g/H_{fin})z$ and using the boundary condition in Eq. (4.72) yields,

$$Z(x, y, 0) = c_5 + c_6 = \psi_{p1} \tag{4.76}$$

and

$$Z(x, y, H_{fin}) = c_5 \exp(-\pi/2) + c_6 \exp(\pi/2) = \phi_B$$
(4.77)

Simplify Eqs. (4.76) and (4.77)

$$c_5 = \left(\psi_{p1} - \frac{\phi_B - \exp(-\pi/2)\psi_{p1}}{\exp(\pi/2) - \exp(-\pi/2)}\right)\exp(-\alpha_1 \hat{z})$$
(4.78)

and

$$c_6 = \frac{\phi_B - \exp(-\pi/2)\psi_{p1}}{\exp(\pi/2) - \exp(-\pi/2)} \exp(\alpha_1 \hat{z})$$
(4.79)

Substituting Eqs. (4.78) and (4.79) into Eq. (4.72) gives

$$Y(x, y, z') = \left[\frac{\phi_B - \exp(-\pi/2)\psi_{p1}}{\exp(\pi/2) - \exp(-\pi/2)}\exp(\alpha_1 \acute{z}) + \left(\psi_{p1} - \frac{\phi_B - \exp(-\pi/2)\psi_{p1}}{\exp(\pi/2) - \exp(-\pi/2)}\right)\exp(-\alpha_1 \acute{z})\right]$$
(4.80)

The potential of Eqs. (4.75) and (4.80) is

$$\psi_{p2} = [V_s \cos(\eta y) + V_d \sin(\eta y)] \times \left[\frac{\phi_B - \exp(-\pi/2)\psi_{p1}}{\exp(\pi/2) - \exp(-\pi/2)} \exp(\alpha_1 \hat{z}) + \left(\psi_{p1} - \frac{\phi_B - \exp(-\pi/2)\psi_{p1}}{\exp(\pi/2) - \exp(-\pi/2)} \right) \exp(-\alpha_1 \hat{z}) \right]$$
(4.81)

here, $\dot{z} = (L_g/H_{fin})z$.

Combining Eqs. (4.81) and (4.66), the total potential becomes

$$\psi_{p} = \psi_{p1} + \left[V_{s}\cos(\eta y) + V_{d}\sin(\eta y)\right] \left[\frac{\phi_{B} - \exp(-\pi/2)\psi_{p1}}{\exp(\pi/2) - \exp(-\pi/2)}\exp(\alpha_{1}\dot{z}) + \left(\psi_{p1} - \frac{\phi_{B} - \exp(-\pi/2)\psi_{p1}}{\exp(\pi/2) - \exp(-\pi/2)}\right)\exp(-\alpha_{1}\dot{z})\right]$$

$$(4.82)$$

 $\psi_0(y)$ is found by using Gauss's law applied on the channel

$$V_g = V_{fb} + \psi_p + \frac{\epsilon_{si}}{C_{ox}} \frac{\partial \psi_p}{\partial x}.$$
(4.83)

where V_{fb} is flat band voltage.

Combining Eqns. (4.66), (4.82) and (4.83), $\psi_0(y)$ can be evaluated iteratively. Finally, Q is obtained by

$$Q = C_{ox}(V_g - V_{fb} - \psi_p)$$
(4.84)

Parameters	T_1 [104]	T_2 [98]	T_3 [149]	$T_4 \ [98]$
$L_g (\mathrm{nm})$	25	40	45	60
$T_{fin} (\mathrm{nm})$	3	20	30	35
H_{fin} (nm)	50	20	100	65
T_{ox} (nm)	1.5	1.0	2.0	1.7
$N_d \; ({\rm m}^{-3})$	5.0×10^{20}	1.14×10^{16}	$1.0{\times}10^{24}$	$1.0{\times}10^{21}$
V_{fb} (V)	-1.1	-0.9	-0.93	-0.9

TABLE 4.1: Physical parameters of different devices used in this study.

TABLE 4.2: Physical constants used in this study.

Parameters	Value		
$n_i \ (\times \ 10^{16} \ {\rm m}^{-3})$	1		
$k_B (\times 10^{-23} \text{J.K}^{-1})$	1.38		
T (K)	300		
$q \ (\times \ 10^{-19} \ {\rm C})$	1.6		
$\epsilon \ (\times \ 10^{-12} \ \mathrm{Fm^{-1}})$	8.85		
ϵ_{si}	11.68ϵ		
ϵ_{ox}	3.9ϵ		

4.3 Results and Discussion

To calibrate and assess the validity of the proposed model, different devices of varying dimensions are selected, the details of which are given in Table 4.1. Transistor T_1 and T_2 are perceived using TCAD; whereas, T_3 and T_4 are fabricated devices and their fabrication details are given in [98, 149]. Devices T_1 and T_2 can also be materialized using a standard fabrication process involving: a) chip cleaning; b) mesa isolation; c) fin definition using lithographic process and etching; d) drain-source contacts formation using ohmic metal evaporation and lift-off; e) gate oxide and f) Schottky metal deposition using lithography and lift-off.

The proposed model also has physical parameters which are used to determine the output characteristics of trigate rectangular Si FinFETs and the same are given in Table 4.2. The devices mentioned in Table 4.1 are a mixture of actual and TCAD simulated devices which will provide a challenge for the proposed model in



FIGURE 4.4: a) Variation in the channel potential predicted by the proposed model. b) Charge distribution inside the channel of a trigate FinFET. The simulations are at $V_{ds} = 1$ V and $V_{gs} = 1.5$ V using the physical dimensions of the device T_3 .

determining its accuracy across devices of various dimensions. TCAD is used to design, examine, and simulate the structure of devices. The simulation of device characteristics consists of two steps: a) structure creation and b) selection of a numerical model. The structure creation includes the definition of mesh, identification of various regions of the device, its electrodes and doping concentration of active layers involved in the definition of the device. While in the selection of a numerical model, one requires the definition of gate work function, choice of a physical model and mathematical method to be engaged by the simulator. Finally, by engaging all above mentioned features, TCAD will generate I - V characteristics of the device under consideration. Eq. (4.41) can be employed to plot the output characteristics of Si trigate FinFETs. To understand the workings of Eq. (4.41), one should keep in mind that the characteristics of the FinFET channel are determined by the Hamiltonian matrix given in Eq. (4.11). Moreover, the developed model is a nearest neighbor tight binding model which assumes that the channel characteristics at any point are only dependent upon its nearest neighbor in both x and y directions. This assumption eliminates the complexity of solving the time-independent Schrödinger wave equation, as the wave function, ψ , is only dependent upon its nearest neighbors as illustrated by Eq. (4.8).

The Hamiltonian matrix describes the channel properties when no bias is applied. To consider the effect of V_{ds} and V_{gs} on the FinFET energy levels, a modified timeindependent Schrödinger wave equation is considered, as illustrated in Eq. (4.14). The terms $\Sigma \psi$ and **s** depend upon the applied bias, as the inflow and outflow of electrons is dependent upon the difference of Fermi energy of the source and drain contacts. Using the channel surface potential given by Eq. (4.82), charge distribution inside the channel is evaluated and is shown by Fig. 4.4. Fig. 4.4 (a) shows the channel potential of the device T_3 at $V_{ds} = 1$ V and $V_{gs} = 1.5$ V. As observed from the figure, potential is maximum at $T_{fin} = 0$ nm, which is the mid point of channel thickness. This is due to the fact that the two side gates have the same voltages as illustrated by Fig. 4.1 (b). Fig. 4.4 (b) shows the charge distribution predicted by Eq. (4.84). Here, it can be observed that at the mid point of T_{fin} , electron concentration is minimum, whereas it increases at the edges



FIGURE 4.5: Fermi-Dirac distribution as a function of applied gate to source voltage, V_{gs} at different energy levels for device T_3 .



FIGURE 4.6: a) Channel transmission and b) difference in Fermi-Dirac distribution energies at $V_{gs} = 2$ V. The dimensions of the simulated device T_3 is provided in Table 4.1.

of T_{fin} . This can be explained by the inversion of carriers due to the applied V_{gs} , which causes a channel of electrons to formulate on the fringes of T_{fin} .

In the proposed model, Fermi-Dirac distribution is calculated using the solution of a 3D Poisson equation as given by Eq. (4.28). Fig. 4.5 shows the variation in Fermi-Dirac distribution as a function of V_{gs} . The Fermi function associated with the proposed model differs to that of the Fermi function evaluated by [8, 146] because the proposed model takes into account the charge distribution inside the channel of FinFETs using the solution of a 3D Poisson equation.

To evaluate the I - V characteristics of a trigate FinFET, Eq. (4.41) is used wherein ballistic transport is assumed thus, all the conduction band electrons from the source will flow towards the drain, provided that there is a difference in Fermi energies between the two respective contacts. Each energy level will provide current equivalent to hq^{-1} and the total current flowing through the channel is the summation of all the individual currents over the Fermi energy differences. The term Trace $[\Gamma_1 \mathbf{G}^{\mathbf{R}} \Gamma_2 \mathbf{G}^{\mathbf{A}}]$ represents the transmission of the electrons at any given energy level. Transmission is controlled by the applied V_{gs} and V_{ds} as shown by Fig. 4.6 (a). Fig. 4.6 (b) shows the Fermi energy difference at any given energy of the device T_3 at $V_{gs} = 2$ V. Examining Fig. 4.6 (a) and 4.6 (b) simultaneously, it can be observed that only those energy levels contribute towards the device current where the product of transmission and Fermi energy difference is greater than zero. The model is calibrated using TCAD simulations and Fig. 4.7 is plotted to show the ability of proposed model to predict I - V characteristics of trigate FinFETs of varying dimensions. Fig. 4.8 shows the proposed model's performance against fabricated devices. In all reported cases, the model's performance is reasonable which is based on the evaluation of charges as a function of V_{ds} and V_{gs} followed by the device I - V characteristics using Eq. (4.41). It is pertinent to mention that devices T_2 , T_3 and T_4 are relatively larger compared to the device T_1 . For device T_1 , quantum effects are inevitable as its thickness is 3 nm and the model predicts its characteristics equally good as that of the other devices which have relatively larger L_q . The ability of the model to predict characteristics of varying



FIGURE 4.7: I - V characteristics of trigate FinFETs for the devices of Table 4.1: a) T_1 and b) T_2 .



FIGURE 4.8: I - V characteristics of trigate FinFETs: a) T_3 and b) T_4 .

dimensions FinFETs with high accuracy, is primarily based on its comprehensive 3D formulation for both Poisson and Schrödinger equations.

To establish the validity of the proposed model further, output and transfer characteristics of all the transistors under discussion are plotted and shown in Figs. 4.9 and 4.10. Examining the plots of these figures, it is evident that the proposed model is capable of predicting both the output and transfer characteristics of Fin-FETs with high accuracy.

The data presented in Table 4.3 establishes the difference between I-V characteristics modeled using a conventional 2D approach and the proposed 3D technique. The model given in [146] represents I-V characteristics evaluated using a NEGF with conventional Fermi-Dirac distribution [8], while the proposed model uses a 3D NEGF with modified Fermi-Dirac distribution to predict difference in Fermi



FIGURE 4.9: Output conductance of trigate FinFETs for the devices of Table 4.1: a) T_1 , b) T_2 , c) T_3 and d) T_4 .



FIGURE 4.10: Transconductance characteristics of trigate FinFETs for the devices of Table 4.1: a) T_1 b) T_2 c) T_3 and d) T_4 .

TABLE 4.3: Average root mean square error (RMSE) of the devices mentionedin Table 4.1.

$\mathbf{A}_{\text{res}} \mathbf{DMCE} \left(\times 10^{-4} \right)$	Device			
Avg. KNISE (\times 10 ⁻⁷)	T_1	T_2	T_3	T_4
2D Model [146]	6.2	27	3.7	43
Proposed 3D Model	3.2	4.8	0.9	4.4
Relative Improvement	48%	82%	78%	92%

energies and then the device current. As seen from the table, at relatively small L_g , the errors of both the models are comparable i.e., for device T_1 . However, when the device dimensions increase, error values for 2D model increase significantly compared to the proposed model, since quantum effects are significantly less at higher L_g and the proposed model caters for it by employing the 3D Poisson equation.

4.4 Summary

In this chapter, a 3D Schrödinger-Poisson model is presented to predict the DC characteristics of rectangular trigate FinFETs. In the proposed model, a 3D Schrödinger equation is solved for FinFET geometry to find the difference of Fermi energies between the source and drain contacts responsible for current flow. To determine the FinFET channel field, it is necessary to have the knowledge of charge distribution inside the channel. The same is evaluated by solving a 3D Poisson equation considering the constraints defined by the geometry of the device. Coupling Schrödinger-Poisson solution, the output current as a function of applied bias is evaluated using a non-equilibrium green function (NEGF). For validation, experimental FinFET I - V characteristics are compared with the modeled data and a good degree of accuracy is observed. Output conductance and transfer characteristics are also compared which further confirm the validity of the proposed model to predict the DC response of nanoscale FinFETs. The model is checked for devices having $T_{fin} = 3 - 35$ nm to establish its reliability for a wider range of Fin-FETs. Based on the demonstrated results, the proposed model could be engaged in CAD related tools meant to predict DC characteristics of trigate FinFETs.

Chapter 5

Predict DC Characteristics of Nano FinFETs Using Deep Neural Network

5.1 Introduction

The semiconductor industry is the backbone of modern digital and analog devices wherein, transistors play a vital role. To improve performance, the device size is reduced for each coming generation [150, 151]. However, reducing the size of the device causes additional complexities in characteristics. Devices with dimensions less than 10 nm exhibit non-ideal behavior referred to as short channel effects. Special techniques are then required to overcome short channel effects and the inception of FinFET is one of those.

In recent years, FinFET have attracted considerable attention from researchers working in the field of nano-devices. FinFETs relative to conventional FETs offer improved performance because of the wrapped gate geometry which provides better control over the device channel [104, 152, 153]. The introduction of nano-FinFETs triggered a revolution in device scaling and opened a new era of scalability. Currently, tri-gate FinFETs are becoming highly competitive in the market



FIGURE 5.1: 3D view of a tri-gate FinFET.

with established 14 nm/16 nm technologies. Their tri-gate nature, in which a thin channel is wrapped by the gate from three sides gives superb control upon the flow of carriers hence, the design mitigates substantially the short channel effects. This also enables the device to exhibit relatively better transconductance with decreased leakage current when compared with conventional FETs [98, 150, 154–158].

Moreover, to improve the performance, different novel structures of FinFETs have been proposed by varying their physical parameters such as: gate length (L_g) ; fin height (H_{fin}) ; fin thickness (T_{fin}) ; oxide thickness (T_{ox}) ; and doping concentration (N_d) [152, 155, 156, 159]. Fig. 5.1 shows a 3D view of a tri-gate FinFET and its associated physical dimensions. The fin height to thickness ratio (H_{fin}/T_{fin}) plays an important role in determining the performance of the device [160]. A high-aspect ratio FinFET offers improved performance at the cost of additional fabrication challenges [152]. Such devices require several iterations and time; resulting in increased developmental cost. Moreover, while scaling down the device, reduction in T_{fin} beyond certain limits poses serious challenges such as quantum confinement, along with gate wrapping limitations [156]. Source-drain punchthrough is another challenge that could cause a large leakage current due to the absence of gate at the bottom of the fin [155]. A punch-through stopping layer can be employed to stop this leakage current, but this can also induce some other critical issues such as: large drain junction tunneling current and large junction leakage current, and also makes the fabrication process more complicated [161].

In ultra-thin devices, where the complete device has been scaled down, the analysis of its performance is considered to be a challenging task. Researchers working on such devices, usually rely upon computational tools referred to as technology computer-aided design (TCAD), to assess the device performance. Normally, TCAD tools employ numerical computation methods based on the device physics to simulate output characteristics. Though, TCAD is believed to be a fairly successful technique, such an approach is time consuming and computationally expensive and requires a high performance simulation system to predict the device characteristics [98, 162].

Numerous techniques have been proposed to reduce the time, including analytical, numerical, and empirical models which are developed, for a given device, to deal with electrical characteristics as a function of device bias and physical variables [45, 92, 101, 163, 164]. The development of an analytical model requires complete knowledge of carrier transport mechanism and its dependence on the device physics and applied bias which is a cumbersome job and usually not preferred by a design engineer. On the other hand, device characteristics can also be realized by developing an empirical model which can generate the device response as a function of the applied bias [26]. Such a modeling technique is relatively easy and is also a time efficient approach. However, it is usually based on a high number of assumptions, and largely has little or no dependence on the device physics.

The industry has been flooded with less artificial and more intelligent problem solving systems in every field. The analytical models can be referred to as unintelligent models that completely succumb to out-of boundary variation in the system instead of learning from it. On the other hand, machine learning approaches, in this respect have outperformed analytical based modeling techniques. A machine learning approach has the capability of teaching itself in a given environment and then predicting an unknown scenario within a framework for which it has been

trained. As a result, machine learning based device modeling has been a focus of research aimed to reducing the computational time, cost, and human effort for device characteristics [110, 111, 152, 165, 166]. Numerous models have been reported using artificial intelligence based techniques such as an artificial neural network (ANN) model developed by Hatami *et al.* [166] to generate I - V characteristics of metal oxide semiconductor field effect transistor (MOSFET). Hatami et al. model requires training to assess I - V characteristics of MOSFETs. However, their model was incompetent when tested on out-of-sample data. This, showed that the developed ANN architecture was not capable of understanding and linking all those variables responsible for generating I - V characteristics of MOSFETs. He *et al.* [110] also proposed a model with a similar approach and presented the approach with less than 1% error compared with measured data. Choi *et al.* [111] reported a complex approach in which finite equations based on the device physics were employed. They used a machine learning technique to find appropriate modeling functions to predict DC characteristics of a Si MOSFET. Recently, Mehta et al. [112] proposed a machine learning approach to predict the drain current and bias dependent capacitance of FETs by taking datasets from simulated devices.

In general, machine learning approaches are more efficient than conventional techniques but face tough challenges in the dataset that is required for training. Already established algorithms for MOSFETs/MESFETs can be trained for Fin-FETs, with less effort by using a transfer learning approach [167]. However, the use of a transfer learning approach in such a case will have limitations because the dataset for training purposes will be altogether a different one [168]. In view of these constraints, a comprehensive machine learning model has been devised that works on FinFET by involving its physical dimensions. The developed approach can be modified to predict MESFET/MOSFET characteristics provided it has been trained appropriately by using the relevant device dataset. Hence it can safely be assumed that the proposed technique would be sufficiently a versatile for predicting the FET characteristics of various natures. In this chapter, a machine learning approach has been developed which uses Fin-FETs' physical parameters, bias voltage, and DC characteristics to train a deep neural network (DNN). The technique optimizes the solution by learning the device physical parameters through hidden nodes and an activation function which eventually generates the DC characteristics of the device. Once the DNN is trained, it is then engaged to predict the DC characteristics of an unknown FinFET by using its physical parameters. The validity of the proposed technique was checked using COMSOL Multiphysics [169] which showed that the predicted DC characteristics of the nano-FinFETs were within a maximum mean square error (MSE) margin of 3.36×10^{-3} . The proposed technique is highly time efficient, and provides fast prediction of FinFET DC characteristics. To the best of our knowledge, DNN based prediction of nano-FinFET characteristics as a function of their physical dimensions is not reported in the literature. Hence, the technique could be a useful tool for the design engineer to obtain the output characteristics of the FinFET using its physical parameters prior to its realization.

5.2 Modeling Methodology

To develop a technique involving DNN to predict the FinFET's DC characteristics, the first and foremost requirement is selecting an appropriate DNN architecture, involving fundamental FinFET design parameters and associated characteristics. Once the architecture is designed, both its validity and accuracy can be assessed using the FinFET's known characteristics. By evaluating the error values of the target and predicted characteristics, the DNN architecture can be fine-tuned to arrive at an optimized solution. The DNN model thus developed will be trained on a known dataset generated for the purpose. The trained DNN model can then be used to assess the device characteristics as a function of its physical parameters before its realization.

5.2.1 DNN Model

Inspired by the human brain, an artificial neural network is used to solve complex problems by transforming them into various interconnecting nodes [170, 171]. Each node is a linear combination of weights and bias values which processes through an activation function interconnected to adjacent layers. Fig. 5.2 shows the functioning of a single neuron in the network which can be written as

$$x_1^{l+1} = \sum_{j=1}^n x_j^l \omega_j^l + b^l$$
(5.1)

here, n is the total number of neurons in previous layer l where l+1 is the current



FIGURE 5.2: Processing flow to get output from a single neuron of a multilayer network system.

layer. Eq. (5.1) is used to represent a single neuron in a multilayer network. However, for more than one neuron and layer, this equation is required to be generalized and can be written in matrix form as

$$\begin{bmatrix} x_{1}^{l+1} \\ x_{2}^{l+1} \\ \vdots \\ x_{m}^{l+1} \end{bmatrix} = \begin{bmatrix} \omega_{11}^{l} & \omega_{12}^{l} & \dots & \omega_{n1}^{l} \\ \omega_{12}^{l} & \omega_{22}^{l} & \dots & \omega_{n2}^{l} \\ \vdots & \vdots & \ddots & \vdots \\ \omega_{1m}^{l} & \omega_{2m}^{l} & \dots & \omega_{nm}^{l} \end{bmatrix} \begin{bmatrix} x_{1}^{l} \\ x_{2}^{l} \\ \vdots \\ x_{n}^{l} \end{bmatrix} + \begin{bmatrix} b_{1}^{l} \\ b_{2}^{l} \\ \vdots \\ b_{n}^{l} \end{bmatrix}$$
(5.2)

here, m is the number of neurons in the current layer.

Equation. (5.2) is used to calculate the output of each neuron of a multilayer network. To incorporate non-linear properties in a neural network, activation function (σ) is used to arrange the data into a specific range. There are different types of activation functions such as tanh, softmax, rectified linear unit (ReLU) and sigmoid [171]. Although, the activation function is chosen according to the complexity of the neural network, number of inputs, and nature of the problem where it has to be applied, there is no specific rule for the selection of an activation function to achieve better accuracy. A couple of activation functions were tried in the developed DNN and based on MSE it was observed that the sigmoid activation function, defined in Eq. 3.3, offers better performance compared to others as shown in Table 5.1.

$$\sigma(x_j^l) = \frac{1}{1 + \exp(-x_j^l)}$$
(5.3)

It is observed from Eq. (5.2) that, the output is a linear combination of input, weights, and bias data. Eq. (5.3) is used to add nonlinearity in the DNN model. The matrices of weight and bias vectors are updated to fit the training datasets in each epoch, where an epoch is one complete training iteration through the entire dataset. Thus, each epoch tries to update the weights to reduce the error between predicted output and actual output. On initialization of DNN training, the model assigns random weights and bias values, which are processed through each layer of neurons to reach the output layer. In addition, the error obtained in the output

TABLE 5.1: MSE attained using various activation functions during DNN training process.

Activation function	MSE (10^{-3})
Softmax	45.6
Linear	24.2
Tanh	5.2
ReLU	3.3
Sigmoid	3.0

Hidden Layer	Epochs	MSE (10^{-3})
512, 256, 128	120	3.7
256, 128, 64	200	3.7
128, 64, 32	350	3.7
64, 32, 16	800	3.7
512, 512, 512	50	3.7
256, 256, 256	70	3.7
128, 128, 128	200	3.7
64, 64, 64	400	3.7
512, 256, 128, 64	500 to 1000	4.6
256, 128, 64	280	3.7
256, 128	130	3.7
256	500 to 1000	3.9

TABLE 5.2: Effects of number of hidden layers and neuron on the proposedDNN technique to predict DC characteristics of FinFETs.

is also propagated in the next epoch to adjust the weights accordingly in order to minimize the error.

The proposed model is built in the Python programming language using Keras [172]. A learning rate of 0.001 is used in the ADAM optimizer [173], which is computationally efficient, requires less memory, and is suitable for a large dataset and or/ a large feature vector. Different combinations of layers along with epochs, listed in Table 5.2, are also tested to achieve the optimum value of training mean square error (MSE). The minimum MSE is achieved using a three layer network with sigmoid as an activation function, having 512, 256, and 128 neurons in each layer. Other training parameters, used in DNN training to achieve the minimum MSE, are summarized in Table 5.3.

5.2.2 DC Characteristics of FinFETs

FinFET has a 3D channel referred to as the fin of the device which is enveloped on three sides by the gate metal is called a tri-gate device [111]. The 3D gate structure allows better control over the channel current which reduces the device

Parameter	Description	
Traning algorithm	Back Propagation	
Network Type	Feed-Forward	
Activation Function	Sigmoid	
Number of Hidden Layers	03	
Number of Neurons in 1^{st} layer	512	
Number of Neurons in 2^{nd} layer	256	
Number of Neurons in $3^{\rm rd}$ layer	128	
Test data	20%	
Loss Function	MSE	
Epoch	2000	
Learning Rate	0.001	
Optimizer	ADAM	

TABLE 5.3: Various parameters used in DNN architecture developed to predict DC characteristics of FinFETs.

leakage and short channel effects. Fig. 5.1 shows a 3D view of a tri-gate FinFET with its physical dimensions marked as L_g , T_{fin} , H_{fin} , and L_{fin} . This is *n*-channel depletion type FinFET and same has been employed for the generation of dataset for COMSOL simulation.

FinFET device structure employs either bulk or silicon on insulator (SOI) substrate. In SOI substrate, the buried oxide thickness, T_{ox} under the gate contact cannot induce an inversion layer at the bottom surface [162]. Further, depending upon the dominant carrier concentration $N_{d,a}$, they are classified into either p or n-channel devices. For extracting the DC characteristics, n-channel FinFETs have been used in this study, and the device output and transfer characteristics are evaluated as a function of drain-to-source (V_{ds}) and gate-to-source (V_{gs}) voltage. It has been observed that, the magnitude and shape of drain-to-source current (I_{ds}) in both the linear and saturation region of operation are dependent upon the bias and chosen physical dimensions of the FinFETs. For the devices under discussion, V_{ds} bias was kept lower than the breakdown value whilst V_{gs} , which controls the amount of I_{ds} trasversing through the channel by changing the depletion inside, was varied from threshold to a value where a tangible I_{ds} was observed. Depending upon the channel parameters, both normally OFF and ON devices were realized for this study. The dataset of output characteristics as a function of applied bias was prepared by varying the physical variables $(L_g, N_d, T_{fin}...)$.

5.2.3 Features and Output Vector of the DNN Model

The proposed DNN model was trained using simulated DC characteristics with associated physical parameters: L_g , H_{fin} , T_{fin} , T_{ox} , L_{fin} and N_d , (Table 5.4). It was noted through series of experimentations that 60 FinFETs provide sufficient data out of which 80 % can be used to train the DNN. The dimensions of the devices were varied randomly and each device was simulated for multiple V_{gs} values to obtain a family of output characteristics to be employed for training purposes. Furthermore, each I - V curve was also attained for a range of V_{ds} values, which in return adds V_{ds} into the input vector for the training of the model. Using COMSOL, a family of $Ids(V_{ds};V_{gs})$ curves are generated for each device and a minimum number of curves for one device was at least five. If the dataset is based on 60 devices then the minimum number of I - V curves the dataset will have is $60 \times 5 = 300$. Additionally the data was generated by using ordinary and extraordinary dimensions of FinFETs to ensure that the model works both for usual and limiting dimensions of the device. The other variables chosen during DNN training are explained in Table 5.3.

A basic architecture of the training model is shown in Fig. 5.3. The model accepts an input vector comprising of device dimensions L_g , H_{fin} , T_{fin} , T_{ox} , L_{fin} , and N_d and bias voltage. All these are interconnected by a set of weights as shown in Eq. (5.1). Initially, using random weights and biases, the information is processed from the input to the output layers passing through each hidden layer in a process called feedforward propagation. During the feed-forward propagation, Eq. (5.1) is used to evaluate the next neuron, based on the previous layer information. At the output layer, an error is generated between the output and target characteristics which is back propagated [174], to be used in the next epoch. The loop continues until the total number of epochs are reached or the error is minimized [112]. For

Parameters	Values		
	10 nm		
L_d	10 nm		
L_g	5 nm to 75 nm		
L_{fin}	20 nm to 100 nm		
T_{fin}	5 nm to 50 nm		
H_{fin}	5 nm to 50 nm		
T_{ox}	2 nm to 20 nm		
N_d	$1 \times 10^{24} \text{ m}^{-3}$ to $10 \times 10^{24} \text{ m}^{-3}$		
Electron Effective Mass	0.197		
Hole Effective Mass	0.8		
Relative Permittivity (Si)	11.68		
Relative Permittivity (SiO_2)	3.9		
Gate Metal Work Function	$4.37 \mathrm{V}$		
Electron Affinity	4.05 V		
Electron Mobility	$1450 \text{ cm}^2(\text{Vs})^{-1}$		
Hole Mobility	$500 \text{ cm}^2(\text{Vs})^{-1}$		

TABLE 5.4: FinFET parameters used in COMSOL software to generate dataset by varying the given dimensions randomly between the two limits given in the Table.



FIGURE 5.3: Schematics diagram of the proposed DNN model developed to predict output characteristics of nano-FinFETs.

updating the weights, gradient descent algorithm is used in the back propagation in an attempt to reduce the error. Error is defined by the average mean square of the difference between target and network output. Mathematically it can be represented as:

$$MSE = \frac{1}{M} \sum_{j=1}^{M} \left[\frac{1}{N} \sum_{i=1}^{N} \left(I_{ds}^{A}(i,j) - I_{ds}^{P}(i,j) \right)^{2} \right]$$
(5.4)

where i, j represent the variation in I_{ds} as a function of V_{ds} and V_{gs} , respectively, such that I_{ds}^A represents the actual value of the drain current whilst I_{ds}^P is the predicted value at the same point by the network output layer. N represents the number of points while sweeping V_{ds} from minimum to maximum and M represents the number of curves.

The whole methodology for developing a system that can predict the characteristics of an unknown device, is summarized in a flow chart given in Fig. 5.4. The data generated by COMSOL Multiphysics are split into training/test data with 80/20 ratio. After initializing all the DNN parameters as listed in Table 5.3, the model is trained using the training data. It is worth mentioning that the network is trained to attain a minimum value of MSE for given epochs. The trained model is then tested and error is evaluated by comparing the predictions based on the feature vector and the test data. The model is said to be well trained, if it shows a prediction error lower than or closer to the average training MSE.

5.3 Results and Discussion

Since the fin of the device is the structure through which the current flows, its height, thickness and length are the most influential parameters. Further, doping concentration also has a direct influence on the current flow from the device therefore, this same was also taken into consideration as a variable. The 3D structure of the simulated device is shown in Fig. 5.1 and the dimensions of variables are explained in Table 5.4. The device geometry was created in COMSOL software



FIGURE 5.4: DNN based training and prediction mechanism to compute DC characteristics of FinFETs.

and each feature of the device is identified uniquely including: ohmic contact; schottky contact; oxide nature (SiO₂), channel layer and doping density. Then the parameters against which the DC characteristics will be plotted are defined as V_{gs} and V_{ds} . Knowing the device dimensions and associated material properties, COMSOL software uses finite element method to solve Schrödinger–Poisson equations governing the channel properties. It divides the entire channel into small pieces using a grid and the size of the grid plays a crucial role in attaining an appropriate solution. It has been noted that a wider grid can solve a problem quickly but at the expense of accuracy. Furthermore, for large grid sizes, the solution has a tendency to diverge. Therefore, for accurate output with high probability of convergence, the grid size should be as fine as possible.

The DNN used herein had five layers such that input and output were represented by one layer coupled with three hidden layers as shown in Fig. 5.3. The number of neurons in first, second, and third hidden layers were 512, 256 and 128, respectively. It was noted that an increase in the hidden layers deteriorated the system performance, and the system also became inefficient with respect to time. During DNN training, the system attained optimized weights for each neuron of the hidden layers and became intelligent enough to predict DC characteristics of a FinFET for which it did not have the data; this is commonly known as a controlled device. After optimization, it was noted that the system attained an overall minimum MSE $\sim 3 \times 10^{-3}$. This value was found from number of training sessions and graphical inspection of training and DNN output data. It is pertinent to mention here that on average the training session took around 330 s and prediction afterward was achieved in a fraction of a second. The trained model afterwards has the ability to predict a device having parameters within the range listed in Table 4.

On the other hand, the simulations performed in COMSOL take much longer compared to DNN prediction. The time consumed by COMSOL for two typical extreme cases in each experiment is listed in Table 5.5. Examining the table it is obvious that varying, H_{fin} from 5 nm to 50 nm changes the required simulation time from 6 min to 3 h and 29 minutes. The highest observed time in Table 5.5 is 41+ h for a FinFET with the following dimensions: $L_g = 10$ nm, $H_{fin} = 50$ nm, $T_{fin} = 30$ nm, $T_{ox} = 2$ nm, $L_{fin} = 10$ nm and $N_d = 1 \times 10^{24}$ m⁻³. This clearly demonstrates that the time cost for the COMSOL model is significantly higher than for the DNN model which takes milliseconds to compute the characteristics once it has been appropriately trained. Thus, the proposed technique can save considerable industrial time; resulting in a possible reduction in the cost of the product.

Figure 5.5 shows the results of the DNN model in comparison to the COMSOL simulation. The devices were chosen such that their output characteristics would have variations in their profiles associated with the device physical dimensions. This enables us to assess the robustness of DNN based prediction. Figure 5.5(a) shows that the device relative to the device of Fig. 5.5(b) offers better output characteristics both in the linear as well as in the saturation region of operations. The only variation in the device physics is the value of L_g which is 70 nm for 5.5(a) and 10 nm for 5.5(b). Increasing the L_g decreases the fringing effects, and thus allows the gate metal to have better control on the channel. Figure 5.6 shows transfer characteristics of the devices under consideration. This figure once again exhibits the validity of the proposed technique to predict FinFET characteristics.

Examining the plots of Figs. 5.5 and 5.6, it is obvious that one can optimize the physical variables of the device to achieve target characteristics. However, this will generally, will be a cumbersome task and require a considerable amount of time and resources. By employing the proposed technique, this can phenomenally reduced without compromising the quality. MSE values of DNN based predicted characteristics relative to COMSOL data are evaluated and given in Table 5.6. In all the predicted devices the average MSE is less than 3×10^{-3} which demonstrates that the achieved MSE of the predicted characteristics of the test devices is within system defined limits.

Prediction accuracy in any DNN model cannot be associated with a specific element as the DNN passes the patterns of information attained from the input layer to the neurons in the next layers which finally arrive at the output units. This accuracy can nevertheless be increased by increasing the data size and/or improving



FIGURE 5.5: I - V characteristics of nano-FinFETs predicted using a DNN model. From (a) to (f) solid lines indicate data attained from COMSOL simulation while dots represent the characteristics generated by the DNN technique.

network layers design [175]. Training data size, however constrains the number of layers such that more training samples are required for shallower models. The variation of accuracy from device to device as observed in Fig. 5.5 and depicted in Table 5.6 cannot be explicitly associated with any chosen physical parameter of the device, DNN architecture or variables.

To further validate the proposed DNN framework, the output characteristics of another set of six devices are shown in Fig. 5.7. In this comparison, the maximum observed average MSE value was 2.29×10^{-3} for the device of Fig. 5.7(f) and the minimum observed MSE was 0.37×10^{-3} for Fig. 5.7(a). Both these values are within the margin defined by the system ($\approx 3 \times 10^{-3}$).



FIGURE 5.6: Transconductance and drain current variation of nano-FinFETs as function of applied bias. (a) to (f) solid lines indicate COMSOL data while dots represent DNN based predicted characteristics.

Variable	Min, Max Values	Time (hours:minute)	
Fin Hoight H.	05 nm	00:06	
I III II III III III III III III III I	50 nm	03:29	
Cata Longth I	05 nm	03:11	
Gate Length, L_g	75 nm	03:29	
Fin Thickness T_{-}	05 nm	00:12	
Thi Thickness, I fin	50 nm	07:56	
Ovido Thielmoss T	02 nm	04:50	
Oxide Thickness, T_{ox}	20 nm	14:57	
Fin Longth L.	10 nm	41:28	
FIII Length, L_{fin}	100 nm	03:19	
Doning Concentration N	$1 \times 10^{24} \text{ m}^{-3}$	07:20	
Doping Concentration, N_d	$10 \times 10^{24} \text{ m}^{-3}$	32:56	

TABLE 5.5: Typical time consumed by the COMSOL software to generate DCcharacteristics for two extreme values of various physical variables of FinFETs.A complete set of variables for these devices are given in Table 5.4.

TABLE 5.6: Mean square error (MSE) values of simulated and DNN based predicted characteristics of six different FinFET having physical dimensions of that of Fig. 5.5

Devices MSE (10^{-3})							
V_{gs} (V)	Fig. 5.5 a	Fig. 5 .5b	Fig. 5.5c	Fig. 5 .5d	Fig. 5.5 e	Fig. 5.5f	
-1.0	-	-	-	0.472	-	-	
-0.8	-	-	-	0.261	-	-	
-0.6	-	-		0.174	-	-	
-0.4	-	3.364	0.0001	0.120	-	1.04	
-0.2	-	1.846	1.317	0.129	-	1.042	
0	0.501	0.100	1.204	1.001	1.40	0.614	
0.2	0.214	0.416	0.560	0.148	2.403	0.398	
0.4	0.177	0.699	0.328	0.274	1.913	0.434	
0.6	0.224	0.794	0.499	0.222	1.242	0.536	
0.8	0.360	0.712	0.653	0.471	0.898	0.754	
1.0	0.382	0.637	0.920	0.806	0.687	0.679	
Avg	0.310	0.12	0.685	0.292	1.40	0.687	



FIGURE 5.7: Transconductance and drain current variation of nano-FinFETs as function of applied bias. (a) to (f) solid lines indicate COMSOL data while dots represent DNN based predicted characteristics.
5.4 Summary

In this chapter, a technique has been proposed using a deep neural network (DNN) to predict DC characteristics of nano-FinFETs. A sizable dataset of DC characteristics was first generated using an established software tool (COMSOL) by varying the device physical dimensions such as gate length (L_g) , fin height (H_{fin}) , fin thickness (T_{fin}) , and fin length (L_{fin}) . The DNN was then trained using the selected data (80%) and the accuracy of the training process was ensured through minimum target error value. The trained DNN model successfully predicted Fin-FETs DC characteristics as a function of applied bias for a given device physical dimensions. The established technique is exceptionally time efficient when compared with standard device simulation software (COMSOL). A comparison of the simulated and predicted data suggests that the developed technique provides on average MSE lower than 2.29×10^{-3} , and can predict the device response with acceptable accuracy by varying the device physical dimensions. The technique, therefore, provides a useful tool for predicting the DC characteristics of a FinFET prior to its fabrication and can save valuable industrial process time.

Chapter 6

Conclusion and Future Work

The electronics being used nowadays are compact in size, consume less power, have high data rate, and can be transported into space quite easily. This has been made possible due to high performing semiconductor FETs that are improved a lot over the decades. The use of FETs increased exponentially after their invention but the end products involving FETs faced certain performance challenges discussed in detail in this research work. These performance factors are affected increasingly because of the fast reduction in device size along with a pressing demand of better or at least similar level of performance. FET based circuitry, both in analogue and digital domains, have a crucial role in electronic industry, however, FET performance is clamped down due to increased degradation in its characteristics caused by the downscaling of the device.

To overcome the challenges faced by the planar FETs; FinFETs having 3D geometry with multi-dimensional gate to control the channel current are introduced. They offer a viable option for microwave high-tech applications, because of their 3D structure, which permits effective gate control over the channel. Owing to 3D gate geometry, FinFETs have effectively mitigated short-channel effects and offered considerably lower switching time, lower dynamic power loss, and higher current density compared to planar FETs. Because of the improved sub-threshold characteristics and exceptionally low leakage current, FinFETs provided a greater level of integration compared to other FET technologies hence, paved the way for future scaling of integrated circuits.

FinFET performance is heavily reliant on the wafer properties and device geometry. The wafer properties include mobility, doping density, saturation velocity, quality of 2DEG etc. Whereas, device geometry includes; fin height, fin thickness, gate length, fin length and oxide thickness. The electrical performance of a finished FinFET depends upon the carriers mobility, their transportation under applied bias and channel response to the signal received at the gate electrode. In order to get an optimum device both from material and design perspective, it is important to have an appropriate selection of device wafer and also a suitable device geometry as per need of the circuitry. It is pertinent to mention here that in FinFETs, the dimensions pertained to the fin such as fin height, fin length and fin thickness coupled with the gate length are of prime importance. At nanoscale, such devices exhibit non-traditional characteristics known as ballistic transportation of carriers. In ballistic transport, carriers are represented by a quantum wave function and the device, therefore, can be referred to as a quantum device. This necessitate the need of involvement of quantum mechanics to explain its characteristics.

FinFETs channel can be made using a bulk semiconductor material such as SiC, GaN or GaAs. If the bandgap of the material is relatively low, i.e. ≤ 1.5 eV, then the device performance deteriorates at higher temperatures. In order to make the device compatible for harsh environments including higher temperatures, the channel could be defined by using a wide bandgap material, i.e. ≥ 3 eV. A FinFET with bulk semiconductor channel, its characteristics are determined by the properties of semiconductor along with the applied bias. The applied potential creates a field inside the channel and as a result of that, the channel carriers drift from drain to source defining the device output characteristics. The magnitude of the current can be altered by varying the potential applied at the gate electrode. Fin-FETs can also be made by using a combination of high and low bandgap materials to generate 2DEG. The 2DEG is controlled by a 3 dimensional gate simultaneously and thus determines the response of the device. Such FinFETs exhibit their

output and transfer characteristics much better than the bulk FinFETs or planar FETs.

If FinFET has a 2DEG channel, then such FinFETs can offer characteristics both in depletion and inversion modes. To describe the response of such devices, a unified model, which can cater both depletion and inversion characteristics simultaneously, would be required. In the first part of the research, a comprehensive I - V model has been developed and it has been shown that the model works reasonably well both in inversion as well as in depletion modes of operation. In the depletion mode, the drain current was assessed using a modified definition of 2DEG carrier concentration caused by the trigate geometry of the device. On the other hand, the inversion part of the model employed Poisson distribution to evaluate the channel potential to be used for the assessment of channel current. The total drain current offered by such devices would then be $I_{ds} = I_{2D} + I_{inv}$. It has been demonstrated that main contributor to I_{ds} for the gate bias $V_{th} \leq V_{gs} \leq V_{fb}$ is 2DEG of the device, whilst a further increase in V_{gs} potential beyond V_{fb} generates an additional component of the current. This part of the current could be associated with inversion of carriers originated in two side gates of an AlGaN/GaN FinFET.

In the second part of the research, a model has been developed using Schrödinger-Poisson equations to predict DC characteristics of trigate Si FinFETs. In this model, quantum-mechanical ballistic transport of carriers has been considered and channel characteristics have been modeled using a 3D Schrödinger wave equation. Source-drain Fermi energy difference, which is responsible for the flow of carriers, has been evaluated using non-equilibrium green function (NEGF) and 3D Poisson equation. Ohmic contact resistance is an important feature that influences upon the magnitude of drain current and cannot be overlooked in models where accuracy is of prime importance. This feature, in the developed technique, is adjusted using modeled parameters in Fermi-Dirac distribution function. Furthermore, in FinFET at $V_{ds}>V_{ds(sat)}$, the drain current saturates which could be associated to the finite supply of carriers from the source electrode of the device. The developed model is calibrated using a standard tool commonly known as technology computer aided design (TCAD). On the other hand, the validity of the developed technique is demonstrated using the experimental data of nanoscale FinFETs both for their output and transfer characteristics and a good degree of accuracy is observed. It has been established that the developed technique has the ability to predict FinFETs characteristics having $T_{fin} = 3 - 35$ nm. Hence, the technique could be engaged in CAD related tools meant to predict DC characteristics of trigate FinFETs.

In the last part of research, an attempt has been made to develop an efficient and reliable DNN model to predict DC characteristics of nano-FinFETs. DNN model has been developed by realizing an appropriate set of input and output variables with logical links between them. Keeping in view the required accuracy and efficiency, an appropriate number of layers and associated neurons have been selected. To train DNN, there was a need to have an accurate dataset of ample size comprising of FinFETs DC characteristics. For this purpose, by using an established software tool (COMSOL), a sizable dataset was generated by varying the device geometrical parameters such as, gate length (L_g) , fin height (H_{fin}) , fin thickness (T_{fin}) , fin length (L_{fin}) , oxide thickness (T_{ox}) and doping density (N_d) . The DNN was trained using the selected data (80%) and the accuracy of the training process was ensured by monitoring minimum target error value. The trained DNN model was then used to predict FinFETs DC characteristics as a function of applied bias for a given device physical dimensions. Comparison of the simulated and predicted data demonstrated that the developed technique is fairly accurate, and can predict the device response by varying the device physical dimensions to an acceptable accuracy. It has been shown that the established technique is highly efficient in time when compared with standard device simulation softwares such as COMSOL. The technique can, therefore, be a useful tool to predict FinFETs DC characteristics prior to their fabrication and can save valuable industrial process time and cost.

This research primarily deals with the prediction of DC characteristics of nanoscale FinFETs irrespective of their materials. Normally, FinFETs are either Si or heterostructure based devices therefore, two famous structures were chosen to develop techniques for FinFETs DC characteristics which should cover both types of Fin-FETs. In the first part of research, AlGaN/GaN FinFETs were studied and based on their characteristics, a model has been developed and validated against the experimental data. On the other hand, in the second part of the research, another model for DC characteristics prediction of Si based FinFETs has been developed to provide two independent analytical models to cover a complete range of Fin-FET devices. In the third part of research, an AI based technique has been worked out and its validity has been demonstrated using Si FinFET data. However, the developed technique is of generic nature and it has the potential to predict Al-GaN/GaN FinFET characteristics if the training data of such devices are provided as an input data. Thus, all the three parts of research carried out in this thesis are fairly linked with each other.

6.1 Future Works

- 1. Since FinFETs are being used in next generation high density SRAMs and other such circuitries, where the temperature of the chip rises with increasing density of transistors. Therefore, temperature dependent FinFETs performance evaluation shall provide a crucial knowledge for determining future growth of high-tech circuitry involving nano-scale FinFETs. At high temperatures, which is inevitable in nanoscale high density circuitries, the conventional device models may exhibit their limitations and hence bound to deviate from the observed data. This aspect needs to be investigated by developing temperature dependent FinFETs models and their validity at the end of day with experimental data, to establish a complete understanding of the device performance.
- 2. In Chapter-3, a unified depletion-inversion model for heterojunction trigate FinFETs DC characteristics was developed and the same was tested for FinFETs having $L_g = 5 \ \mu m$ to 100 nm. Its validity for devices having $L_g < 100 \ nm$ would be an exciting challenge, which may require certain

modifications in the developed model and the same could be carried out as one of the future extension of this research.

- 3. In FinFETs, the gate length (L_g) of the device plays a crucial role in defining the response of the device. As a rule of thumb, reduction in gate length enhances the high frequency operation of the device. However, this reduction requires an appropriate scaling of other associated physical parameters of the device such as fin height (H_{fin}) , fin thickness (T_{fin}) , fin width (W_{fin}) , fin length (L_{fin}) , oxide thickness (T_{ox}) etc. A comprehensive parametric studies to assess the effect of each of these upon other variables, will help in determining an optimum device having high transconductance, high unity gain frequency with minimum short channel effects.
- 4. The mathematical formulation carried out in this research is restricted to the DC performance of the device. A next logical step to extend this work would be to establish AC characteristics of these FinFETs and then to extract scattering parameters of the device for comparison with experimental AC data to have improved understanding of the device AC response.
- 5. A machine learning technique developed in this study has successfully predicted the device characteristics. However, it was noted that in some cases (less than 5%) the prediction accuracy was not within acceptable range. Hence there is a room to improve the developed technique by incorporating more layers or/and by enhancing the number of neurons of a layer along with other design variables. Additionally, the prediction carried out in this thesis is only for the device DC characteristics, it is therefore, recommended that an appropriate AC dataset can also be generated and a modified DNN may be trained using both AC and DC characteristics to generate full range of characteristics either two independent systems or a single system incorporated with both types of characteristics.
- 6. The machine learning demonstration has been formulated that works on FinFETs DC characteristics by engaging its physical parameters. The proposed DNN model can be altered to anticipate heterostructure, nanowires,

nanosheets, vertically stacked 3D devices etc. Consequently it can safely be said that, the proposed technique is a flexible one to be adopted for a variety of electronic devices provided the model is trained accordingly.

7. In mobile applications, battery life is treated to be an important feature in determining the usefulness of a product. The battery life heavily depends upon the magnitude of off-state current drawn by a circuitry. In case of FinFETs based circuitry, this off-state current defines sub-threshold characteristics of a FinFET. Ideally, in sub-threshold region the current drawn by the device should be zero. To bring the device off-state current closer to ideal value, the sub-threshold performance requires to be enhanced by improving the wafer as well as by optimizing the device geometry. So the device geometry dependent sub-threshold characteristics model will then be a useful tool to enhance the device performance and the same will provide a beneficial knowledge for industrial applications and fundamental understanding of the device.

Bibliography

- L. J. Edgar, "Method and Apparatus for Controlling Electric Currents," Jan. 28 1930, US Patent 1,745,175.
- [2] M. Riaz, M. M. Ahmed, and U. Munir, "An Improved Model for Current Voltage Characteristics of Submicron SiC MESFETs," *Solid-State Electronics*, vol. 121, pp. 54–61, 2016.
- [3] D. Hisamoto, T. Kaga, Y. Kawamoto, and E. Takeda, "A Fully Depleted Lean-Channel Transistor (DELTA), a Novel Vertical Ultrathin SOI MOS-FET," *IEEE Electron Device Letters*, vol. 11, no. 1, pp. 36–38, 1990.
- [4] D. A. Neamen *et al.*, Semiconductor Physics and Devices. McGraw-Hill New York, 1997, vol. 3.
- [5] Roll and Guntrade, Leakage Current and Defect Characterization of Short Channel MOSFETs. Logos Verlag Berlin GmbH, 2012, vol. 2.
- [6] J. Y. Wu, H. H. Wang, Y. H. Wang, and M. P. Houng, "Fabrication of Depletion-mode GaAs MOSFET with a Selective Oxidation Process by Using Metal as the Mask," *IEEE Electron Device Letters*, vol. 22, no. 1, pp. 2–4, 2001.
- [7] W. Haensch, E. J. Nowak, R. H. Dennard, P. M. Solomon, A. Bryant, O. H. Dokumaci, A. Kumar, X. Wang, J. B. Johnson, and M. V. Fischetti, "Silicon CMOS Devices Beyond Scaling," *IBM Journal of Research and Development*, vol. 50, no. 4.5, pp. 339–361, 2006.

- [8] D. A. Neamen, Semiconductor Physics and Devices: Basic Principles. New York, NY: McGraw-Hill, 2012.
- [9] W. Choi, O. Seok, H. Ryu, H.-Y. Cha, and K.-S. Seo, "High Voltage and Low-Leakage-Current Gate Recessed Normally-Off GaN MIS-HEMTs With Dual Gate Insulator Employing PEALD-SiN_x /RF-Sputtered-HfO₂," *IEEE Electron Device Letters*, vol. 35, no. 2, pp. 175–177, 2014.
- [10] H. Hiremath, N. Mallesh, and K. Sunil, "A Comparative Study on Different Semiconductor Materials used for Power Devices and Its Applications," *International Journal of Latest Technology in Engineering, Management and Applied Science*, vol. 07, no. 1, pp. 2278–2540, 2018.
- [11] M. N. Khan, U. F. Ahmed, M. M. Ahmed, and S. Rehman, "An Improved Model to Assess Temperature-Dependent DC Characteristics of Submicron GaN HEMTs," *Journal of Computational Electronics*, pp. 1–10, 2018.
- [12] S. A. Eliza, "Modeling of AlGaN/GaN High Electron Mobility Transistor for Sensors and High-Temperature Circuit Applications," 2008.
- [13] A. Tessmann, A. Leuther, H. Massler, and M. Seelmann-Eggebert, "A High Gain 600 GHz Amplifier Using 35 nm Metamorphic HEMT Technology," in 2012 IEEE Compound Semiconductor Integrated Circuit Symposium (CSICS). IEEE, 2012, pp. 1–4.
- [14] H. Xie, Z. Liu, W. Hu, Z. Zhong, K. Lee, Y. X. Guo, and G. I. Ng, "GaN-on-Si HEMTs Fabricated With Si CMOS-Compatible Metallization for Power Amplifiers in Low-Power Mobile SoCs," *IEEE Microwave and Wireless Components Letters*, vol. 31, no. 2, pp. 141–144, 2020.
- [15] S. E. Thompson and S. Parthasarathy, "Moore's Law: The Future of Si Microelectronics," *Materials today*, vol. 9, no. 6, pp. 20–25, 2006.
- [16] R. Shankar, S. Sharma, and S. Dasgupta, "Recent Trend of FinFET Devices and its Challenges: A Review," in *Proceedings of the IEEE Conference on Emerging Devices and Smart Systems*, 2017.

- [17] T. Chiarella, L. Witters, A. Mercha, C. Kerner, M. Rakowski, C. Ortolland, L.-Å. Ragnarsson, B. Parvais, A. De Keersgieter, S. Kubicek *et al.*, "Benchmarking SOI and Bulk FinFET Alternatives for Planar CMOS Scaling Succession," *Solid-State Electronics*, vol. 54, no. 9, pp. 855–860, 2010.
- [18] M. Bohr, "14 nm Process Technology: Opening New Horizons," in Intel Development Forum, 2014.
- [19] H.-W. Hsiao and Y. R. Wu, "3D Self-Consistent Quantum Transport Simulation for GaAs Gate-All-Around Nanowire Field-Effect Transistor with Elastic and Inelastic Scattering Effects," *physica status solidi (a)*, vol. 216, no. 1, p. 1800524, 2019.
- [20] Y. K. Choi, N. Lindert, P. Xuan, and S. Tang, "Sub 20 nm CMOS FinFET Technologies," *IEEE International Electron Device Meeting*, pp. 1911–1914, 2001.
- [21] U. Ahmed, S. Rehman, U. Rafique, and M. Ahmed, "AlGaN/GaN FinFET: A Comparative Study," in 2018 14th International Conference on Emerging Technologies (ICET). IEEE, 2018, pp. 1–6.
- [22] V. Moroz, S. L. Smith, and Q. Lu, "FinFET with Heterojunction and Improved Channel Control," Mar. 24 2016, US Patent App. 14/925,777.
- [23] Y. W. Jo, D. H. Son, C. H. Won, K. S. Im, J. H. Seo, I. M. Kang, and J. H. Lee, "AlGaN/GaN FinFET With Extremely Broad Transconductance by Side-Wall Wet Etch," *IEEE Electron Device Letters*, vol. 36, no. 10, pp. 1008–1010, 2015.
- [24] D. H. Son, Y. W. Jo, V. Sindhuri, K. S. Im, J. H. Seo, Y. T. Kim, I. M. Kang, S. Cristoloveanu, M. Bawedin, and J. H. Lee, "Effects of Sidewall MOS Channel on Performance of AlGaN/GaN FinFET," *Microelectronic Engineering*, vol. 147, pp. 155–158, 2015.

- [25] S.-U. Rehman, U. F. Ahmed, M. M. Ahmed, and M. N. Khan, "Temperature Dependent Analytical DC Model for Wide Bandgap MESFETs," *IEEE Access*, vol. 7, pp. 49702–49711, 2019.
- [26] Q.-U. D. Memon, U. F. Ahmed, and M. M. Ahmed, "A Unified Depletion/Inversion Model for Heterojunction Trigate FinFETs DC Characteristics," *IEEE Access*, vol. 9, pp. 89768–89777, 2021.
- [27] M. Asheghi, K. Kurabayashi, R. Kasnavi, and K. Goodson, "Thermal Conduction in Doped Single-Crystal Silicon Films," *Journal of Applied Physics*, vol. 91, no. 8, pp. 5079–5088, 2002.
- [28] M. Poljak, V. Jovanović, and T. Suligoj, "Improving bulk FinFET DC Performance in Comparison to SOI FinFET," *Microelectronic Engineering*, vol. 86, no. 10, pp. 2078–2085, 2009.
- [29] N. Kaur, M. Rattan, and S. S. Gill, "Impact of Gate Oxide Thickness and Aspect Ratio of Fin Height and Fin width on Nanoscale Tapered FinFETs," in *IEEE International Conference on Recent Trends in Electronics, Information & Communication Technology (RTEICT).* IEEE, 2016, pp. 1787–1791.
- [30] Z. Zhao, Y. Li, Y. Zan, Y. Li, J. Li, X. Cheng, G. Wang, H. Liu, H. Wang, Q. Zhang *et al.*, "Fabrication Technique of the Si_{0.5} Ge_{0.5} Fin for the High Mobility Channel FinFET Device," *Semiconductor Science and Technology*, vol. 35, no. 4, p. 045015, 2020.
- [31] M. Bhole, A. Kurude, and S. Pawar, "FinFET Benefits, Drawbacks and Challenges," Int. J. of Engineering, Sciences and Research Technology, vol. 2, pp. 3219–3222, 2013.
- [32] Y. Xu, S. Cristoloveanu, M. Bawedin, K. S. Im, and J. H. Lee, "Performance Improvement and Sub-60 mV/decade Swing in AlGaN/GaN FinFETs by Simultaneous Activation of 2DEG and Sidewall MOS Channels," *IEEE Transactions on Electron Devices*, vol. 65, no. 3, pp. 915–920, 2018.

- [33] Y. Long, H. Wang, J. Z. Huang, J. Gong, and X. Jiang, "Performance Enhancement by Gate Tunable Strain in p-type Piezoelectric FinFETs," in 2018 14th IEEE International Conference on Solid-State and Integrated Circuit Technology (ICSICT). IEEE, pp. 1–3.
- [34] S. Groetsch, S. Illek, A. Ploessl, and B. Hahn, "Carrier Layer for a Semiconductor Layer Sequence and Method for Producing Semiconductor Chips," May 25 2010, US Patent 7,723,730.
- [35] F. Cozette, B. Hassan, C. Rodriguez, E. Frayssinet, R. Comyn, F. Lecourt, N. Defrance, N. Labat, F. Boone, A. Soltani *et al.*, "New Barrier Layer Design for the Fabrication of Gallium Nitride-Metal-Insulator-Semiconductor-High Electron Mobility Transistor Normally-Off Transistor," *Semiconductor Science and Technology*, vol. 36, no. 3, p. 034002, 2021.
- [36] T. A. Bhat, M. Mustafa, and M. Beigh, "Study of Short Channel Effects in n-FinFET Structure for Si, GaAs, GaSb and GaN Channel Materials," 2015.
- [37] F. Ana and Najeebuddin, "Suppression of Gate Induced Drain Leakage Current (GIDL) by Gate Work function Engineering: Analysis and Model," *Journal of Electron Devices*, vol. 13, pp. 984–996, 2012.
- [38] J. Min, G. Choe, and C. Shin, "Gate Induced Drain Leakage (GIDL) in MFMIS and MFIS Negative Capacitance FinFETs," *Current Applied Physics*, vol. 20, no. 11, pp. 1222–1225, 2020.
- [39] A. Karsenty and A. Chelly, "Anomalous DIBL Effect in Fully Depleted SOI MOSFETs Using Nanoscale Gate-Recessed Channel Process," Active and Passive Electronic Components, vol. 2015, 2015.
- [40] T. Sakurai and K. Tamaru, "Simple Formulas for Two-and Three-Dimensional Capacitances," *IEEE transactions on electron devices*, vol. 30, no. 2, pp. 183–185, 1983.

- [41] M. M. Ahmed, "Compression in Transconductance at Low Gate Voltages in Submicron GaAs Metal Semiconductor Field-Effect Transistors," Journal of Vacuum Science & Technology B: Microelectronics and Nanometer Structures Processing, Measurement, and Phenomena, vol. 15, no. 6, pp. 2052–2056, 1997. [Online]. Available: https://avs.scitation.org/doi/abs/10. 1116/1.589222
- [42] Q. Dai and J. H. Lee, "Subthreshold Characteristics of AlGaN/GaN MIS-FinFETs with Controlling Threshold Voltages," *Electronics*, vol. 9, no. 11, p. 1967, 2020.
- [43] M. Veshala, R. Jatooth, K. R. Reddy et al., "Reduction of Short-Channel Effects in FinFET," International Journal of Engineering and Innovative Technology (IJEIT) Volume, vol. 2, 2013.
- [44] R. Saha, S. Baishya, and B. Bhowmick, "3D Analytical Modeling of Surface Potential, Threshold Voltage, and Subthreshold swing in Dual-Material-gate (DMG) SOI FinFETs," *Journal of Computational Electronics*, vol. 17, no. 1, pp. 153–162, 2018.
- [45] U. F. Ahmed, M. M. Ahmed, and Q. D. Memon, "Non-Linear Compact Model for FinFETs Output Characteristics," *IET Circuits, Devices & Sys*tems, vol. 13, no. 8, pp. 1249–1254, 2019.
- [46] R. K. Sharma, M. Bucher, and C. A. Dimitriadis, "Analog/RF Performance: Are SOI FinFETs Better than DG SOI MOSFETs?" *IEEE Transactions on Nanotechnology*, vol. 11, no. 5, 2012.
- [47] M. V. Dunga, C.-H. Lin, X. Xi, D. D. Lu, A. M. Niknejad, and C. Hu, "Modeling Advanced FET Technology in a Compact Model," *IEEE Transactions* on *Electron Devices*, vol. 53, no. 9, pp. 1971–1978, 2006.
- [48] H. Sarvari, A. H. Ghayour, Z. Chen, and R. Ghayour, "Analyses of Short Channel Effects of Single-Gate and Double-Gate Graphene Nanoribbon Field Effect Transistors," *Journal of Materials*, vol. 2016, p. 8, 2016.

- [49] G. Jayadeva and A. DasGupta, "Analytical Approximation for the Surface Potential in N-Channel MOSFETs Considering Quantum–Mechanical Effects," *IEEE transactions on electron devices*, vol. 57, no. 8, pp. 1820–1828, 2010.
- [50] S. Panigrahy, "Studies of Short Channel Effects and Performance Enhancement of Nano-MOSFET Based on Multi-Objective Genetic Algorithm Approach," Master's thesis, National Institute of Technology, Rourkela, 2013.
- [51] J. Chen, T. Chan, I. Chen, P. Ko, and C. Hu, "Subbreakdown Drain Leakage Current in MOSFET," *IEEE Electron Device Letters*, vol. 8, no. 11, pp. 515– 517, 1987.
- [52] C. Hu, "Gate Oxide Scaling Limits and Projection," in International Electron Devices Meeting, 1996, pp. 319–322.
- [53] T. S. Park, E. Yoon, and J. H. Lee, "A 40 nm Body-Tied FinFET (OMEGA MOSFET) Using Bulk Si Wafer," *Physics E: Low-dimensional Systems and Nanostructures*, vol. 19, no. 1-2, pp. 6–12, 2003.
- [54] T. K. Liu, "FinFET History, Fundamentals and Future," in Proceedings of the Symposium on VLSI Technology Short Course. University of California, 2012.
- [55] M. Bohr and K. Mistry, "Intel's Revolutionary 22 nm Transistor Technology," *Intel website*, 2011.
- [56] M. Hossain, B. Salhi, A. Mukhaimer, and F. Al-Sulaiman, "Fabrication and Optical Simulation of Vertically Aligned Silicon Nanowires," *Applied Nanoscience*, vol. 6, no. 7, pp. 1031–1036, 2016.
- [57] L. Chang, S. Tang, T.-J. King, J. Bokor, and C. Hu, "Gate Length Scaling and Threshold Voltage Control of Double-Gate MOSFETs," in *International Electron Devices Meeting Technical Digest (Cat. No. 00CH37138)*. IEEE, 2000, pp. 719–722.

- [58] Y.-K. Choi, N. Lindert, P. Xuan, S. Tang, D. Ha, E. Anderson, T.-J. King, J. Bokor, and C. Hu, "Sub-20 nm CMOS FinFET Technologies," in *International Electron Devices Meeting Technical Digest (Cat. No. 01CH37224)*. IEEE, 2001, pp. 19–1.
- [59] K. Kim, O. Kwon, J. Seo, and T. Won, "Nanoscale Device modeling and Simulation: Fin Field-Effect Transistor (FinFET)," *Japanese Journal of Applied Physics*, vol. 43, no. 6S, p. 3784, 2004.
- [60] V. K. Mishra and R. Chauhan, "Comparison of Multiple Fin Heights for Increasing Drain Current in n-FinFET," in 2013 International Conference on Circuits, Controls and Communications (CCUBE). IEEE, 2013, pp. 1–5.
- [61] Z. Guo, S. Balasubramanian, R. Zlatanovici, T.-J. King, and B. Nikolić, "FinFET-based SRAM design," in *Proceedings of the 2005 international symposium on Low Power Electronics and Design*. ACM, 2005, pp. 2–7.
- [62] S. S. Ensan, M. H. Moaiyeri, M. Moghaddam, and S. Hessabi, "A Low-Power Single-Ended SRAM in FinFET Technology," *AEU-International Journal of Electronics and Communications*, vol. 99, pp. 361–368, 2019.
- [63] A. Muttreja, N. Agarwal, and N. K. Jha, "CMOS Logic Design with Independent-Gate FinFETs," in 2007 25th International Conference on Computer Design. IEEE, 2007, pp. 560–567.
- [64] D. Nirmal, P. Vijayakumar, D. M. Thomas, B. K. Jebalin, and N. Mohankumar, "Subthreshold Performance of Gate Engineered FinFET Devices and Circuit with High-k Dielectrics," *Microelectronics Reliability*, vol. 53, no. 3, pp. 499–504, 2013.
- [65] S. Liu, J. Wang, Y. Lu, D. Huang, C. Huang, W. Hsieh, J. Lee, Y. Tsai, J. Shih, Y. H. Lee *et al.*, "Self-Heating Effect in FinFETs and its Impact on Devices Reliability Characterization," in 2014 IEEE International Reliability Physics Symposium. IEEE, 2014, pp. 4A–4.

- [66] J. Alvarado, J. Tinoco, S. Salas, A. Martinez Lopez, B. Soto Cruz, A. Cerdeira, and J. P. Raskin, "SOI FinFET Compact Model for RF Circuits Simulation," in 2013 IEEE 13th Topical Meeting on Silicon Monolithic Integrated Circuits in RF Systems. IEEE, 2013, pp. 87–89.
- [67] P.-E. Gaillardon, R. Walker, and B. S. Rodriguez, "Breakthroughs in Analog and RF Circuit Performance through Steep Slope FinFETs," University of Utah Salt Lake City United States, Tech. Rep., 2016.
- [68] D. Lederer, V. Kilchytska, T. Rudenko, N. Collaert, D. Flandre, A. Dixit, K. De Meyer, and J.-P. Raskin, "FinFET Analogue Characterization from DC to 110 GHz," *Solid-State Electronics*, vol. 49, no. 9, pp. 1488–1496, 2005.
- [69] G. Crupi, D. Schreurs, B. Parvais, A. Caddemi, A. Mercha, and S. Decoutere, "Scalable and Multibias High Frequency Modeling of Multi-Fin FETs," *Solid-State Electronics*, vol. 50, no. 11-12, pp. 1780–1786, 2006.
- [70] A. Kundu, B. Syamal, K. Koley, C. Sarkar, and N. Mohankumar, "RF Parameter Extraction of Bulk FinFET: A non Quasi Static Approach," in 2010 IEEE International Conference of Electron Devices and Solid-State Circuits (EDSSC). IEEE, 2010, pp. 1–4.
- [71] A. Es Sakhi and M. Chowdhury, "Analysis of Device Capacitance and Subthreshold Behavior of Tri-Gate SOI FinFET," *Microelectronics Journal*, vol. 62, pp. 30–37, 2017.
- [72] M. Qin, Y. Sun, X. Li, and Y. Shi, "Analytical Parameter Extraction for Small-Signal Equivalent Circuit of 3D FinFET into Sub-THz Range," *IEEE Access*, vol. 6, pp. 19752–19761, 2018.
- [73] O. Ambacher, B. Foutz, J. Smart, J. Shealy, N. Weimann, K. Chu, M. Murphy, A. Sierakowski, W. Schaff, L. Eastman *et al.*, "Two Dimensional Electron Gases Induced by Spontaneous and Piezoelectric Polarization in Undoped and Doped AlGaN/GaN Heterostructures," *Journal of applied physics*, vol. 87, no. 1, pp. 334–344, 2000.

- [74] U. K. Mishra, P. Parikh, and Y. F. Wu, "AlGaN/GaN HEMTs an Overview of Device Operation and Applications," *Proceedings of the IEEE*, vol. 90, no. 6, pp. 1022–1031, 2002.
- [75] S. Takashima, Z. Li, and T. P. Chow, "Sidewall Dominated Characteristics on Fin-Gate AlGaN/GaN MOS-Channel-HEMTs," *IEEE transactions on electron devices*, vol. 60, no. 10, pp. 3025–3031, 2013.
- [76] K. S. Im, H. S. Kang, J. H. Lee, S. J. Chang, S. Cristoloveanu, M. Bawedin, and J. H. Lee, "Characteristics of GaN and AlGaN/GaN FinFETs," *Solid-State Electronics*, vol. 97, pp. 66–75, 2014.
- [77] K. S. Im, H. S. Kang, D. K. Kim, S. Vodapally, Y. Park, J. H. Lee, Y. T. Kim, S. Cristoloveanu, and J. H. Lee, "Temperature-Dependent Characteristics of AlGaN/GaN FinFETs With Sidewall MOS Channel," *Solid-State Electronics*, vol. 120, pp. 47–51, 2016.
- [78] K. Ren, Y. C. Liang, and C. F. Huang, "Physical Mechanism of Fin-Gate Al-GaN/GaN MIS-HEMT: Vth Model," in 2016 IEEE 4th Workshop on Wide Bandgap Power Devices and Applications (WiPDA). IEEE, 2016, pp. 319– 323.
- [79] P. Ghatak, D. Dutta, and N. Bhat, "A Physics Based Analytical Model for the Threshold Voltage of a Normally-off AlGaN/GaN FinFET," in *International Workshop on the Physics of Semiconductor and Devices*. Springer, 2017, pp. 205–218.
- [80] K. Zhang, Y. Kong, G. Zhu, J. Zhou, X. Yu, C. Kong, Z. Li, and T. Chen, "High-Linearity AlGaN/GaN FinFETs for Microwave Power Applications," *IEEE Electron Device Letters*, vol. 38, no. 5, pp. 615–618, 2017.
- [81] M. Zhang, X. H. Ma, L. Yang, M. Mi, B. Hou, Y. He, S. Wu, Y. Lu, H. S. Zhang, Q. Zhu *et al.*, "Influence of Fin Configuration on the Characteristics of AlGaN/GaN Fin-HEMTs," *IEEE Transactions on Electron Devices*, vol. 65, no. 5, pp. 1745–1752, 2018.

- [82] U. F. Ahmed, M. M. Ahmed, and Q. D. Memon, "A Modified Analytical Model for AlGaN/GaN FinFETs I–V Characteristics," *Semiconductor Sci*ence and Technology, vol. 35, no. 3, p. 035002, 2020.
- [83] K. N. Huang, Y. C. Lin, J. H. Lee, C. C. Hsu, J. N. Yao, C. Y. Wu, C. H. Chien, and E. Y. Chang, "Study of Tri-Gate AlGaN/GaN MOS-HEMTs for Power Application," *Micro and Nano Engineering*, p. 100073, 2020.
- [84] M. Luisier, A. Schenk, and W. Fichtner, "Quantum Transport in Twoand Three-Dimensional Nanoscale Transistors: Coupled Mode Effects in the Nonequilibrium Green's Function Formalism," *Journal of Applied physics*, vol. 100, no. 4, p. 043713, 2006.
- [85] F. Gamiz, "A Comprehensive Study of Velocity Overshoot Effects in Double Gate Silicon on Insulator Transistors," *Semiconductor Science and technol*ogy, vol. 19, no. 3, p. 393, 2003.
- [86] X. Shao and Z. Yu, "Nanoscale FinFET Simulation: A Quasi-3D Quantum Mechanical Model Using NEGF," *Solid-State Electronics*, vol. 49, no. 8, pp. 1435–1445, 2005.
- [87] H. R. Khan, D. Mamaluy, and D. Vasileska, "3D NEGF Quantum Transport Simulator for Modeling Ballistic Transport in Nano FinFETs," in *Journal* of *Physics: Conference Series*, vol. 107, no. 1. IOP Publishing, 2008, p. 012007.
- [88] K. Natori, "Compact Modeling of Ballistic Nanowire MOSFETs," IEEE transactions on electron devices, vol. 55, no. 11, pp. 2877–2885, 2008.
- [89] E. Dastjerdy, R. Ghayour, and H. Sarvari, "3D Quantum Mechanical Simulation of Square Nanowire MOSFETs by Using NEGF Method," *Central European Journal of Physics*, vol. 9, no. 2, pp. 472–481, 2011.
- [90] Y. M. Sabry, T. M. Abdolkader, and W. F. Farouk, "Simulation of Quantum Transport in Double-Gate MOSFETs Using the Non-Equilibrium Green's

Function Formalism in Real-Space: A Comparison of Four Methods," International Journal of Numerical Modelling: Electronic Networks, Devices and Fields, vol. 24, no. 4, pp. 322–334, 2011.

- [91] A. Deyasi and A. Sarkar, "Analytical Computation of Electrical Parameters in GAAQWT and CNTFET With Identical Configuration Using NEGF Method," *International Journal of Electronics*, vol. 105, no. 12, pp. 2144– 2159, 2018.
- [92] X. Zhang and Y. Liu, "Electronic Transport and Spatial Current Patterns of 2D Electronic System: A Recursive Greenâ€[™]s Function Method Study," *AIP Advances*, vol. 9, no. 11, p. 115209, 2019.
- [93] W. Huang, H. Zhu, J. Li, Z. Yan, Y. Zhang, Q. Wang, X. Ai, Z. Xiao, Z. Kong, J. Xiang *et al.*, "A novel 3d nor flash with single-crystal silicon channel: Devices, integration, and architecture," *IEEE Electron Device Letters*, 2022.
- [94] Z. Zhang, W. Gan, J. Li, Z. Kong, Y. Han, Y. Liu, G. Wang, Z. Wu, J. Yu, Q. Zhang *et al.*, "Scallop-shaped p-type finfets with improved short-channel effects immunity and driving current," *Materials Science in Semiconductor Processing*, vol. 140, p. 106337, 2022.
- [95] X. Cheng, Y. Li, F. Zhao, A. Chen, H. Liu, C. Li, Q. Zhang, H. Yin, J. Luo, and W. Wang, "4-levels vertically stacked sige channel nanowires gate-allaround transistor with novel channel releasing and source and drain silicide process," *Nanomaterials*, vol. 12, no. 5, p. 889, 2022.
- [96] Y. Taur, X. Liang, W. Wang, and H. Lu, "Continuous, Analytic Drain-Current Model for DG MOSFETs," *IEEE Electron Device Letters*, vol. 25, no. 2, pp. 107–107, 2004.
- [97] A. Ortiz-Conde, F. J. G. Sanchez, and J. Muci, "Rigorous Analytic Solution for the Drain current of Undoped Symmetric Dual-Gate MOSFETs," *Solid-State Electronics*, vol. 49, no. 4, pp. 640–647, 2005.

- [98] N. Fasarakis, A. Tsormpatzoglou, D. H. Tassis, I. Pappas, K. Papathanasiou, M. Bucher, G. Ghibaudo, and C. A. Dimitriadis, "Compact Model of Drain Current in Short-Channel Triple-Gate FinFETs," *IEEE transactions* on electron devices, vol. 59, no. 7, pp. 1891–1898, 2012.
- [99] N. Paydavosi, S. Venugopalan, Y. S. Chauhan, J. P. Duarte, S. Jandhyala, A. M. Niknejad, and C. C. Hu, "BSIM–SPICE Models Enable FinFET and UTB IC Designs," *IEEE Access*, vol. 1, pp. 201–215, 2013.
- [100] V. Kumari, K. Sharmetha, M. Saxena, and M. Gupta, "Underlapped Fin-FET on Insulator: Quasi 3D Analytical Model," *Solid-State Electronics*, vol. 129, pp. 138–149, 2017.
- [101] U. F. Ahmed and M. M. Ahmed, "A 3-d Potential Model to Assess DC Characteristics of Si FinFETs," *Journal of Computational Electronics*, vol. 18, no. 3, pp. 893–905, 2019.
- [102] M. Khan, U. Ahmed, M. Ahmed, and S. Rehman, "Simulation and Comparative Analysis of the DC Characteristics of Submicron GaN HEMTs for Use in CAD Software," *Journal of Computational Electronics*, vol. 18, no. 2, pp. 482–491, 2019.
- [103] B. Raj, A. Saxena, and S. Dasgupta, "A Compact Drain Current and Threshold Voltage Quantum Mechanical Analytical Modeling for FinFETs," J Nanoelectron Optoelectron, vol. 3, no. 2, p. 163, 2008.
- [104] A. Yesayan, F. Prégaldiny, N. Chevillon, C. Lallement, and J.-M. Sallese, "Physics-Based Compact Model for Ultra-Scaled FinFETs," *Solid-state electronics*, vol. 62, no. 1, pp. 165–173, 2011.
- [105] P. Ghosh, S. Haldar, R. Gupta, and M. Gupta, "An Analytical Drain Current Model for Dual Material Engineered Cylindrical/surrounded Gate MOS-FET," *Microelectronics Journal*, vol. 43, no. 1, pp. 17–24, 2012.
- [106] J. P. Duarte, S. J. Choi, D. I. Moon, J. H. Ahn, J. Y. Kim, S. Kim, and Y. K. Choi, "A Universal Core Model for Multiple-Gate Field-Effect Transistors."

Part II: Drain Current Model," *IEEE Transactions on Electron Devices*, vol. 60, no. 2, pp. 848–855, 2013.

- [107] M. Alam, A. Kranti, and G. Armstrong, "An Efficient Neural Network Approach for Nanoscale FinFET Modelling and Circuit Simulation," International Journal of Numerical Modelling: Electronic Networks, Devices and Fields, vol. 22, no. 5, pp. 379–393, 2009.
- [108] Z. Marinković, G. Crupi, D. M. P. Schreurs, A. Caddemi, and V. Marković, "Microwave FinFET Modeling Based on Artificial Neural Networks Including Lossy Silicon Substrate," *Microelectronic engineering*, vol. 88, no. 10, pp. 3158–3163, 2011.
- [109] W. C. Chen, C. P. Chang, M. K. Kang, T. U. Huang, K. B. Wu, and R. B. Wu, "Artificial Neural Network Modeling for Extrinsic Capacitance of FinFET," in 2014 IEEE 23rd Conference on Electrical Performance of Electronic Packaging and Systems. IEEE, 2014, pp. 95–98.
- [110] J. He, G. Hu, B. Xie, G. Ma, P. He, L. Song, C. Li, D. Lin, J. Liu, Y. Yu et al.,
 "Modeling Nanoscale FinFET Performance by a Neural Network Method," Sensors & Transducers, vol. 214, no. 7, p. 53, 2017.
- [111] H. C. Choi, H. Yun, J. S. Yoon, and R. H. Baek, "Neural Approach for Modeling and Optimizing Si-MOSFET Manufacturing," *IEEE Access*, vol. 8, pp. 159351–159370, 2020.
- [112] K. Mehta and H. Y. Wong, "Prediction of FinFET Current-Voltage and Capacitance-Voltage Curves using Machine Learning with Autoencoder," *IEEE Electron Device Letters*, vol. 42, no. 2, pp. 136–139, 2020.
- [113] R. Ghoshhajra, K. Biswas, and A. Sarkar, "Device Performance Prediction of Nanoscale Junctionless FinFET Using MISO Artificial Neural Network," *Silicon*, pp. 1–10, 2022.

- [114] C. Su, H. Ke, and T. Hubing, "Overview of electromagnetic modeling software," in Proc. of the 25th International Review of Progress in Applied Computational Electromagnetics, 2009, pp. 736–741.
- [115] N. Dhurandhar and P. Dwivedi, "A Review Article on Fin Field Effect Transistors Technology," CSVTU Research Journal on Engineering and Technology, vol. 8, no. 1, pp. 31–38, 2019.
- [116] R. S. Pal, S. Sharma, and S. Dasgupta, "Recent Trend of FinFET Devices and Its Challenges: A Review," in 2017 Conference on Emerging Devices and Smart Systems (ICEDSS). IEEE, 2017, pp. 150–154.
- [117] W. Zhang, S. Yin, W. Hu, and Y. Wang, "Novel Physics-Based Small-Signal Modeling and Characterization for Advanced RF Bulk FinFETs," *IEEE Transactions on Electron Devices*, vol. 68, no. 5, pp. 2160–2166, 2021.
- [118] P. Wang, X. Ma, M. Mi, M. Zhang, J. Zhu, Y. Zhou, S. Wu, J. Liu, L. Yang, B. Hou *et al.*, "Influence of Fin-Like Configuration Parameters on the Linearity of AlGaN/GaN HEMTs," *IEEE Transactions on Electron Devices*, vol. 68, no. 4, pp. 1563–1569, 2021.
- [119] M. Xu, H. Zhu, L. Zhao, H. Yin, J. Zhong, J. Li, C. Zhao, D. Chen, and T. Ye, "Improved Short Channel Effect Control in Bulk FinFETs With Vertical Implantation to form Self-Aligned Halo and Punch-Through Stop Pocket," *IEEE Electron Device Letters*, vol. 36, no. 7, pp. 648–650, 2015.
- [120] H. Zhou, D. Kwon, A. B. Sachid, Y. Liao, K. Chatterjee, A. J. Tan, A. K. Yadav, C. Hu, and S. Salahuddin, "Negative Capacitance, N-Channel, Si Fin-FETs: Bi-Directional Sub-60 mV/dec, Negative DIBL, Negative Differential Resistance and Improved Short Channel Effect," in 2018 IEEE Symposium on VLSI Technology. IEEE, 2018, pp. 53–54.
- [121] S. Rajabi, S. Mandal, B. Ercan, H. Li, M. A. Laurent, S. Keller, and S. Chowdhury, "A Demonstration of Nitrogen Polar Gallium Nitride Current Aperture Vertical Electron Transistor," *IEEE Electron Device Letters*, vol. 40, no. 6, pp. 885–888, 2019.

- [122] M. Saremi, R. Hathwar, M. Dutta, F. A. Koeck, R. J. Nemanich, S. Chowdhury, and S. M. Goodnick, "Analysis of the Reverse IV Characteristics of Diamond-Based PIN Diodes," *Applied Physics Letters*, vol. 111, no. 4, p. 043507, 2017.
- [123] M. Saremi, "Modeling and Simulation of the Programmable Metallization Cells (PMCs) and Diamond-Based Power Devices," Ph.D. dissertation, Arizona State University, 2017.
- [124] H. A. Moghadam, S. Dimitrijev, J. Han, D. Haasmann, and A. Aminbeidokhti, "Transient-Current Method for Measurement of Active Near-Interface Oxide Traps in 4H-SiC MOS Capacitors and MOSFETs," *IEEE Transactions on Electron Devices*, vol. 62, no. 8, pp. 2670–2674, 2015.
- [125] S. Li, S. Lu, and C. C. Mi, "Revolution of Electric Vehicle Charging Technologies Accelerated by Wide Bandgap Devices," *Proceedings of the IEEE*, vol. 109, no. 6, pp. 985–1003, 2021.
- [126] E. A. Jones, F. F. Wang, and D. Costinett, "Review of Commercial GaN Power Devices and GaN-Based Converter Design Challenges," *IEEE Journal* of Emerging and Selected Topics in Power Electronics, vol. 4, no. 3, pp. 707– 719, 2016.
- [127] T. J. Flack, B. N. Pushpakaran, and S. B. Bayne, "GaN Technology for Power Electronic Applications: A Review," *Journal of Electronic Materials*, vol. 45, no. 6, pp. 2673–2682, 2016.
- [128] I. Daumiller, C. Kirchner, M. Kamp, K. J. Ebeling, and E. Kohn, "Evaluation of the Temperature Stability of AlGaN/GaN Heterostructure FETs," *IEEE Electron Device Letters*, vol. 20, no. 9, pp. 448–450, 1999.
- [129] K. Ren, Y. C. Liang, and C.-F. Huang, "Compact Physical Models for AlGaN/GaN MIS-FinFET on Threshold Voltage and Saturation Current," *IEEE Transactions on Electron Devices*, vol. 65, no. 4, pp. 1348–1354, 2018.

- [130] N. Karumuri, S. Turuvekere, N. DasGupta, and A. DasGupta, "A Continuous Analytical Model for 2-DEG Charge Density in AlGaN/GaN HEMTs Valid for All Bias Voltages," *IEEE Transactions on Electron Devices*, vol. 61, no. 7, pp. 2343–2349, 2014.
- [131] P. Sai, D. But, I. Yahniuk, M. Grabowski, M. Sakowicz, P. Kruszewski, P. Prystawko, A. Khachapuridze, K. Nowakowski-Szkudlarek, J. Przybytek et al., "AlGaN/GaN Field Effect Transistor With Two Lateral Schottky Barrier Gates Towards Resonant Detection in Sub-Mm Range," Semiconductor Science and Technology, vol. 34, no. 2, p. 024002, 2019.
- [132] A. Kumar, B. Tiwari, S. Singh, M. M. Tripathi, and R. Chaujar, "Radiation Analysis of N-Channel TGRC-MOSFET: An X-Ray Dosimeter," *IEEE Transactions on Electron Devices*, vol. 65, no. 11, pp. 5014–5020, 2018.
- [133] A. Kumar, M. Tripathi, and R. Chaujar, "Sub-30nm In₂O₅Sn Gate Electrode Recessed Channel MOSFET: A Biosensor for Early Stage Diagnostics," *Vacuum*, vol. 164, pp. 46–52, 2019.
- [134] A. Martinez Lopez, A. Cerdeira, J. C. Tinoco, J. Alvarado, W. Padron, C. Mendoza, and J. P. Raskin, "RF Modeling of 40-Nm SOI Triple-Gate FinFET," *International Journal of Numerical Modelling: Electronic Net*works, Devices and Fields, vol. 28, no. 4, pp. 465–478, 2015.
- [135] G. Crupi, D. M.-P. Schreurs, A. Caddemi, I. Angelov, M. Homayouni, A. Raffo, G. Vannini, and B. Parvais, "Purely Analytical Extraction of an Improved Nonlinear FinFET Model Including Non-Quasi-Static Effects," *Microelectronic engineering*, vol. 86, no. 11, pp. 2283–2289, 2009.
- [136] A. Kumar, N. Gupta, S. K. Tripathi, M. Tripathi, and R. Chaujar, "Performance Evaluation of Linearity and Intermodulation Distortion of Nanoscale GaN-SOI FinFET for RFIC Design," *AEU-International Journal of Electronics and Communications*, vol. 115, p. 153052, 2020.

- [137] G. Avolio, D. M.-P. Schreurs, A. Raffo, G. Crupi, A. Caddemi, G. Vannini, and B. Nauwelaers, "Straightforward Modeling of Dynamic I – V Characteristics for Microwave FETs," International Journal of RF and Microwave Computer-Aided Engineering, vol. 24, no. 1, pp. 109–116, 2014.
- [138] F. Felgenhauer, M. Begoin, and W. Mathis, "Analysis of Parasitic Quantum Effects in Classical CMOS Circuits," *International Journal of Numerical Modelling: Electronic Networks, Devices and Fields*, vol. 18, no. 4, pp. 313– 323, 2005.
- [139] Y. K. Lin, J. P. Duarte, P. Kushwaha, H. Agarwal, H. L. Chang, A. Sachid, S. Salahuddin, and C. Hu, "Compact Modeling Source-to-Drain Tunneling in Sub-10-nm GAA FinFET with Industry Standard Model," *IEEE Transactions on Electron Devices*, vol. 64, no. 9, pp. 3576–3581, 2017.
- [140] F. A. Berezin and M. Shubin, *The Schrödinger Equation*. Springer Science & Business Media, 2012, vol. 66.
- [141] H. S. Dinh, "Simulation of Current-Voltage Characteristics of Graphene Field Effect Transistor (GFET)," Science and Technology Development Journal, vol. 16, no. 3, pp. 5–12, 2013.
- [142] R. Venugopal, M. Paulsson, S. Goasguen, S. Datta, and M. S. Lundstrom,
 "A Simple Quantum Mechanical Treatment of Scattering in Nanoscale Transistors," *Journal of Applied Physics*, vol. 93, no. 9, pp. 5613–5625, 2003.
- [143] N. Sarkar, "Study of Dephasing Mechanisms on the Potential Profile of a Nanowire FET," *Engineering Research Express*, vol. 1, no. 2, p. 025029, 2019.
- [144] S. Jin, Y. J. Park, and H. S. Min, "A Three-Dimensional Simulation of Quantum Transport in Silicon Nanowire Transistor in the Presence of Electron-Phonon Interactions," *Journal of Applied Physics*, vol. 99, no. 12, p. 123719, 2006.

- [145] H. Takeda and N. Mori, "Mode-Coupling Effects in Non-Equilibrium Green's Function Device Simulation," *Japanese journal of applied physics*, vol. 44, no. 4S, p. 2664, 2005.
- [146] Z. Ren, R. Venugopal, S. Goasguen, S. Datta, and M. S. Lundstrom, "NanoMOS 2.5: A Two-Dimensional Simulator for Quantum Transport in Double-Gate MOSFETs," *IEEE Transactions on Electron Devices*, vol. 50, no. 9, pp. 1914–1925, 2003.
- [147] B. Heinrich, Finite Difference Methods on Irregular Networks: A Generalized Approach to Second Order Elliptic Problems. Springer, 1987.
- [148] A. Cracknell and K. C. Wong, "Fermi Surface. Its Concept, Determination, and Use in the Physics of Metals," 1973.
- [149] Y.-S. Kim, S. H. Shin, S. H. Han, S. C. Yang, J. Sung, D. J. Lee, J. W. Lee, and T. Y. Chung, "Fabrication and Electrical Properties of Local Damascene FinFET Cell Array in Sub-60nm Feature Sized DRAM," *Journal of semiconductor technology and science*, vol. 6, no. 2, pp. 61–66, 2006.
- [150] S. Verma, S. L. Tripathi, and M. Bassi, "Performance Analysis of FinFET Device Using Qualitative Approach for Low-Power Applications," in 2019 Devices for Integrated Circuit (DevIC). IEEE, 2019, pp. 84–88.
- [151] B. Y. Tsui and C. P. Lin, "A Novel 25 nm Modified Schottky-Barrier FinFET with High Performance," *IEEE Electron Device Letters*, vol. 25, no. 6, pp. 430–432, 2004.
- [152] Y. G. Liaw, W. S. Liao, M. C. Wang, C. L. Lin, B. Zhou, H. Gu, D. Li, and X. Zou, "A High Aspect Ratio Silicon-fin FinFET Fabricated upon SOI Wafer," *Solid-State Electronics*, vol. 126, pp. 46–50, 2016.
- [153] S. Khandelwal, J. P. Duarte, A. Medury, Y. Chauhan, and C. Hu, "New Industry Standard FinFET Compact Model for Future Technology Nodes," in 2015 Symposium on VLSI Technology (VLSI Technology). IEEE, 2015, pp. T62–T63.

- [154] G. Saini and A. K. Rana, "Physical Scaling limits of FinFET Structure: A Simulation Study," International Journal of VLSI design & communication Systems (VLSICS), vol. 2, no. 1, pp. 26–35, 2011.
- [155] H. Zhou, Y. Song, Q. Xu, Y. Li, and H. Yin, "Fabrication of Bulk-Si FinFET using CMOS Compatible Process," *Microelectronic engineering*, vol. 94, pp. 26–28, 2012.
- [156] K. Bailey, K. Gurumurthy et al., "3D Device Modeling and Assessment of Triple Gate SOI FinFET for LSTP applications," International Journal of Computer Theory and Engineering, vol. 4, no. 5, pp. 831–836, 2012.
- [157] K. S. Im, C. H. Won, Y. W. Jo, J. H. Lee, M. Bawedin, S. Cristoloveanu, and J. H. Lee, "High-Performance GaN-Based Nanochannel FinFETs with/without AlGaN/GaN Heterostructure," *IEEE Transactions on Electron Devices*, vol. 60, no. 10, pp. 3012–3018, 2013.
- [158] M. K. Rai, V. Narendar, and R. A. Mishra, "Significance of Variation in Various Parameters on Electrical Characteristics of FinFET Devices," 2014 Students Conference on Engineering and Systems, pp. 1–6, 2014.
- [159] N. Somra, K. Mishra, and R. S. Sawhney, "Optimizing Current Characteristics of 32 nm FinFET by Controlling Fin Width," *Communications on Applied Electronics*, vol. 2, pp. 1–5, 2015.
- [160] V. Jovanović, T. Suligoj, M. Poljak, Y. Civale, and L. K. Nanver, "Ultra-High Aspect-Ratio FinFET Technology," *Solid-state electronics*, vol. 54, no. 9, pp. 870–876, 2010.
- [161] J. G. Fossum, Z. Zhou, L. Mathew, and B.-Y. Nguyen, "SOI Versus Bulk-Silicon Nanoscale FinFETs," *Solid-State Electronics*, vol. 54, no. 2, pp. 86– 89, 2010.
- [162] N. Kaur, M. Rattan, and S. S. Gill, "Impact of Gate Oxide Thickness and Aspect Ratio of Fin Height and Fin Width on Nanoscale Tapered FinFETs,"

in IEEE International Conference on Recent Trends in Electronics, Information & Communication Technology (RTEICT). IEEE, 2016, pp. 1787–1791.

- [163] U. F. Ahmed and M. M. Ahmed, "An Analytical Model to Assess DC Characteristics of Independent Gate Si FinFETs," *Turkish Journal of Electrical Engineering & Computer Sciences*, vol. 27, no. 4, pp. 2456–2465, 2019.
- [164] X. Zhang, D. Connelly, P. Zheng, H. Takeuchi, M. Hytha, R. J. Mears, and T.-J. K. Liu, "Analysis of 7/8-nm Bulk-Si FinFET Technologies for 6T-SRAM Scaling," *IEEE Transactions on Electron Devices*, vol. 63, no. 4, pp. 1502–1507, 2016.
- [165] A. Gaurav, S. S. Gill, N. Kaur, and M. Rattan, "Density Gradient Quantum Corrections Based Performance Optimization of Triangular TG Bulk FinFETs using ANN and GA," in 2016 20th International Symposium on VLSI Design and Test (VDAT). IEEE, 2016, pp. 1–5.
- [166] S. Hatami, M. Y. Azizi, H.-R. Bahrami, D. Motavalizadeh, and A. Afzali-Kusha, "Accurate and Efficient Modeling of SOI MOSFET with Technology Independent Neural Networks," *IEEE Transactions on Computer-Aided De*sign of Integrated Circuits and Systems, vol. 23, no. 11, pp. 1580–1587, 2004.
- [167] F. L. Da Silva and A. H. R. Costa, "A survey on transfer learning for multiagent reinforcement learning systems," *Journal of Artificial Intelligence Research*, vol. 64, pp. 645–703, 2019.
- [168] L. Torrey and J. Shavlik, "Transfer learning," in Handbook of research on machine learning applications and trends: algorithms, methods, and techniques. IGI global, 2010, pp. 242–264.
- [169] C. Multiphysics, "Introduction to COMSOL multiphysics," COMSOL Multiphysics, Burlington, MA, accessed Feb, vol. 9, p. 2018, 1998.
- [170] H. F. Ahmed, F. K. Malik, I. Akhtar, and Z. Bangash, "Application of Data-Driven and Physics-Embedded Neural Networks in Wake Dominated

Flows," in 2021 International Bhurban Conference on Applied Sciences and Technologies (IBCAST), 2021, pp. 810–816.

- [171] H. F. Ahmed, H. Farooq, I. Akhtar, and Z. Bangash, "Machine Learning Based Reduced-Order Modeling of Hydrodynamic Forces Using Pressure Mode Decomposition," *Proceedings of the Institution of Mechanical Engineers, Part G: Journal of Aerospace Engineering*, vol. 235, no. 16, pp. 2517– 2528, 2021.
- [172] Chollet, Francois and Others. (2015) Keras. [Online]. Available: https://github.com/fchollet/keras
- [173] D. P. Kingma and J. Ba, "Adam: A Method for Stochastic Optimization," arXiv preprint arXiv:1412.6980, 2014.
- [174] Ian Goodfellow and Yoshua Bengio and Aaron Courville, Deep Learning. MIT Press, 2016, http://www.deeplearningbook.org.
- [175] S. Mittal, H. Gupta, and S. Srivastava, "A survey on hardware security of dnn models and accelerators," *Journal of Systems Architecture*, vol. 117, p. 102163, 2021.